COURT FILE NO.

1501-06320

COURT

COURT OF QUEEN'S BENCH OF ALBERT.

JUDICIAL CENTRE

CALGARY

PLAINTIFF

WELLS FARGO FOOTHILL CANADA ULC

DEFENDANTS

BIG EAGLE HYDRO-VAC INC., BIG EAGLE LIMITED PARTNERSHIP, BIG EAGLE HYDRO-VAC (NORTH) INC.,

and BIG EAGLE (NORTH) LIMITED PARTNERSHIP

DOCUMENT

AUCTION AND ASSET

SOLICITATION PROCEDURES ORDER

ADDRESS FOR SERVICE AND

Gowling Lafleur Henderson LLP Suite 1600, 421-7th Avenue SW

Calgary, AB T2P 4K9

CONTACT INFORMATION

Phone: (403) 298-1818

OF PARTY FILING THIS

Fax: (403) 695-3558

DOCUMENT

File No.: A150430

Attention: Jeffrey Oliver

DATE ON WHICH ORDER WAS PRONOUNCED:

June 29, 2015

the original

Dated this:

NAME OF JUDGE WHO MADE THIS ORDER:

The Honourable Madam Justice J.

I hereby certify this to be a true copy of

Strekaf

LOCATION OF HEARING:

Calgary, Alberta

UPON the application of Deloitte Restructuring Inc. in its capacity as the court-appointed receiver and manager (the 'Receiver') of the undertaking, property and assets of Big Eagle Hydro-Vac Inc., Big Eagle Limited Partnership, Big Eagle Hydro-Vac (North) Inc., and Big Eagle (North) Limited Partnership (the 'Debtors') for an order approving, inter alia, the sale process contemplated by the auction service agreement between the Receiver and Great American Global Partners, LLC (the 'Stalking Horse Bid') and the auction and asset solicitation procedures (the "AASP") developed by the Receiver, each as appended to the First Report of the Receiver dated June 26, 2015 (the 'First Report');

AND UPON having read the Receivership Order dated June 8, 2015 (the "Receivership Order") and the First Report;

AND UPON hearing the submissions of counsel to the Receiver;

IT IS HEREBY ORDERED AND DECLARED THAT:

- 1. The time for service of the notice of application for this order is hereby abridged and service thereof is deemed good and sufficient.
- 2. The actions of the Receiver to date as reported in the First Report are hereby ratified and approved.
- 3. The Stalking Horse Bid, attached as Schedule "A" to this order, is hereby approved.
- 4. The AASP, attached as Schedule "B" to this Order, is hereby approved.
- 5. The Receiver is hereby authorized to take any steps that are reasonably necessary or desirable to carry out the AASP.
- 6. Paragraph 2 of the Receivership Order is hereby amended to read:

Pursuant to section 243(1) of the Bankruptcy and Insolvency Act, R.S.C. 1985, c. B-3 ("BIA") and section 13(2) of the Judicature Act, R.S.A. 2000, c.J-2 Deloitte Restructuring Inc. is hereby appointed Receiver, without security, of all of the Debtors' current and future assets, undertakings and properties of every nature and kind whatsoever, and wherever situate, including all proceeds thereof other than the Statutory Payments Reserve (as defined below) (the "Property"), including but not limited to the real property of Big Eagle Hydro-Vac Inc., more particularly known and described as

PLAN 8821687
BLOCK 1
LOT 8
EXCEPTING THEREOUT ALL MINES AND MINERALS

PLAN 8323083 BLOCK 3 LOT 14 EXCEPTING THEREOUT ALL MINES AND MINERALS.

PLAN 8323083
BLOCK 3
LOT 15
EXCEPTING THEREOUT ALL MINES AND MINERALS

LOT 6
PLAN BCP12890
DISTRICT LOT 1535
PID # 026-025-124 (the "BC Lands")

(collectively, the "Lands").

- 7. Upon presentation for registration in the Prince George Land Title Office a certified copy of this Order and the Receivership Order, the British Columbia Register of Land Titles is hereby directed to register a copy of this Order and the Receivership Order against title to the BC Lands.
- 8. The sale of five (5) coiled tubing units and trailers to 2015 DHS Capital Co. for \$490,000, as described in the First Report, is hereby approved.
- 9. This Order shall have full force and effect in all provinces and territories in Canada.
- 10. This Court requests the aid and recognition of any Court, tribunal, regulatory or administrative bodies, having jurisdiction in Canada or in the United States of America, to give effect to this order and to assist the Receiver and its respective agents in carrying out the terms of this order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Receiver, as an officer of this Court, as may be necessary or desirable to give effect to this order, or to assist the Receiver and its respective agents in carrying out the terms of this order.

- 11. The Receiver shall serve, by courier, fax transmission, e-mail transmission, or ordinary post, a copy of this Order on all parties present at this Application and on all parties who received notice of this Application or who are presently on the service list established in these proceedings, and such service shall be deemed good and sufficient for all purposes.
- 12. The Receiver and any other interested party shall be at liberty to apply for further advice, assistance and directions as necessary in order to give full force and effect to the terms of this order.

Jystice of the Court of Queen's Bench of Alberta

SCHEDULE A

Auction Services Agreement with Great American Global Partners LLC

AUCTION SERVICES AGREEMENT

This Agreement is made as of June 25, 2015 between

DELOITTE RESTRUCTURING INC., in its capacity as Receiver and Manager of the property, assets and undertaking of Big Eagle Hydro-Vac Inc., Big Eagle Limited Partnership, Big Eagle Hydro-Vac (North) Inc., and Big Eagle (North) Limited Partnership and not in its personal capacity (the "Receiver")

and

GREAT AMERICAN GLOBAL PARTNERS, LLC (the "Agent")

RECITALS

- A. Pursuant to an order of the Honourable Madam Justice Strekaf sitting as a Judge of the Court of Queen's Bench of Alberta (the "Court") dated June 8, 2015 (the "Receivership Order"), the Receiver was appointed as receiver and receiver and manager of all of the assets, undertaking and property of Big Eagle Hydro-Vac Inc., Big Eagle Limited Partnership, Big Eagle Hydro-Vac (North) Inc., and Big Eagle (North) Limited Partnership (collectively, the "Debtor") pursuant to section 243(1) of the Bankruptcy and Insolvency Act and section 13(2) of the Judicature Act (Alberta) under Court File No. 1501-06320 (the "Receivership").
- B. Subject to the terms therein, the Receivership Order authorizes the Receiver to sell or otherwise dispose of the assets, property and undertaking of the Debtor, or any part thereof. Exercising its powers pursuant to the Receivership Order and subject to the approval of the Court, the Receiver has agreed to retain the Agent to sell the Assets (as hereinafter defined).
- C. The Authorization and Approval Order is expected to contemplate that upon execution of bills of sale issued by the Agent or the Receiver to the ultimate purchasers of the Assets, title to the Assets will vest in the ultimate purchasers thereof free and clear of all liens, security interests, assignments, actions, levies, taxes, mortgages, charges, encumbrances, trust claims and all other adverse claims whatsoever of any person.

FOR VALUE RECEIVED, the parties agree as follows:

SECTION 1- INTERPRETATION

1.1 Definitions

In this Agreement:

- (1) Account has the meaning ascribed to it in Section 2.6;
- Accounting Deadline has the meaning ascribed to it in Section 2.7(2);
- (3) Accounting Items has the meaning ascribed to it in Section 2.7(1);
- (4) Agent's Expense Reimbursement means the amount of \$500,000 payable to the Agent in accordance with this Agreement on account of the Agent's out of pocket expenses in relation to the execution and performance of this Agreement.
- (5) Appraisal has the meaning ascribed to it in Section 2.3;
- (6) Appraisal Costs means the costs in the aggregate amount of \$50,000 incurred by the Agent to conduct an inspection of the Assets and prepare the Appraisal;
- (7) Approval Date means the date on which the Authorization and Approval Order is granted by the Court;
- (8) Assets means the right, title and interest of the Debtor and the Receiver in and to (i) the vehicles, rolling stock, machinery and equipment located at the Debtor's Premises or used by the Debtor in carrying on its Business listed on Schedule 1.1(8), (ii) the office equipment and office furniture located at the Debtor's Premises, and (iii) all trademarks, patents, industrial designs, trade names, domain names and other intellectual property, but, for greater certainty, does not include the Excluded Assets;
- (9) Auction has the meaning ascribed to it in Section 2.1;
- (10) Auction Date has the meaning ascribed to it in Section 2.1;
- (11) Authorization and Approval Order has the meaning ascribed to it in Section 4.1(b);
- (12) Bid Solicitation Process means the process designed by the Receiver consistent with the principles set out in the seminal decision of the Ontario Court of Appeal in Royal Bank v. Sound Air Corp. regarding sale processes and sale approval;
- (13) Break-up Fee has the meaning ascribed to it in Section 2.20;
- (14) Business Day means a day on which banks are open for business in the City of Calgary but does not include a Saturday, Sunday or statutory holiday in the Province of Alberta;
- (15) Buyer's Premium means the 15% fee charged to purchasers by the Agent upon sale of the Assets;
- (16) Commencement Date has the meaning ascribed to it in Section 1.1(27);
- (17) Court has the meaning ascribed to it in Recital A;
- (18) **Debtor** has the meaning ascribed to it in Recital A;

- (19) Debtor's Premises means the following premises:
 - (a) 5916 36 Street East, Edmonton International Airport, Leduc, Alberta;
 - (b) 9319 81 Avenue, Clairmont, Alberta; and
 - (c) 2747 North Service Road West, Swift Current, Saskatchewan;
- (20) Deposit has the meaning ascribed to it in Section 2.4(1);
- (21) Deposit Recovery has the meaning ascribed to it in Section 2.6(1)(b);
- (22) Distributions has the meaning ascribed to it in Section 2.4(2);
- (23) Employees has the meaning ascribed to it in Section 2.12;
- (24) Excluded Assets means (i) all of the Debtor's real property; (ii) all fixtures including, all components of the building's electrical (other than, with the consent of the Receiver acting reasonably, any electrical components installed by the Debtor and exclusively dedicated to the Assets), mechanical, plumbing, heating and air-conditioning systems, fire extinguishers and other fire safety equipment, and (iii) all motor vehicles and office equipment leased by the Debtor;
- (25) Indemnified Parties has the meaning ascribed to it in Section 2.9(6);
- (26) Interim Inspection has the meaning ascribed to it in Section 2.9(4);
- (27) Liquidation Period means the period commencing on the date upon which the conditions in Section 4.1 are satisfied or waived in accordance with this Agreement (the "Commencement Date"), unless the Receiver and the Agent agree in writing to a later date and ending at 12 noon (Calgary time) on the Termination Date;
- (28) Make-Ready Repair Costs means the costs and expenses approved, in advance, by the Receiver and the Agent, acting reasonably, to prepare the Assets for auction, which cost may include, in the discretion of the Receiver and the Agent, the costs to conduct exterior and interior cleaning, painting, welding and metal work, light mechanical repair (excluding the labour costs of up to three (3) of the Debtor's mechanics during regular business hours), decal replacement, glass work and upholstery repair, provided that the Agent acknowledges that in no event will the Make Ready Repair Costs exceed \$200,000 without prior written consent of the Receiver, which consent may be withheld in the Receiver's sole and unfettered discretion.
- (29) Net Minimum Guaranteed Amount means the amount referred to in Section 2.3;
- (30) **Person** means any natural person, sole proprietorship, partnership, corporation, trust, joint venture, any governmental authority or any incorporated or unincorporated entity or association of any nature;
- (31) Purchaser Bill of Sale has the meaning ascribed to it in Section 4.1(b);

- (32) Receiver has the meaning ascribed to it in first page of this Agreement;
- (33) Receiver's Recovery Amount has the meaning ascribed to it in Section 2.6(1)(c);
- (34) Receivership has the meaning ascribed to it in Recital A;
- (35) Receivership Order has the meaning ascribed to it in Recital A;
- (36) Sales Taxes has the meaning ascribed thereto in Section 2.15;
- (37) Superior Bid has the meaning scribed to it in Section 2.20.
- (38) Termination Date means thirty (30) days after the Auction Date.
- (39) Transaction means the liquidation, sale and removal of the Assets contemplated by this Agreement; and
- (40) Weekly Sale Reconciliation has the meaning ascribed to it in Section 2.7(1);
- (41) Wells Fargo has the meaning ascribed to it in Section 2.4(2).

1.2 Interpretation Not Affected by Headings, etc.

The division of this Agreement into sections and the insertion of headings are for convenience of reference only and shall not affect the construction or interpretation of this Agreement. Unless otherwise indicated, all references to a "Section" followed by a number and/or a letter refer to the specified section of this Agreement. The terms "this Agreement", "hereof", "herein" and "hereunder" and similar expressions refer to this Agreement and not to any particular section hereof.

1.3 Extended Meanings

Words importing the singular include the plural and vice versa, words importing gender include all genders and words importing persons include individuals, partnerships, associations, trusts, unincorporated organizations, corporations and governmental authorities. The term "including" means "including, without limitation," and such terms as "includes" have similar meanings.

SECTION 2 – SALE OF ASSETS AND EXCLUDED EQUIPMENT

2.1 Sale of Assets

The Receiver shall use its reasonable efforts to seek all court orders contemplated herein in a timely manner. Effective upon the issuance of the Authorization and Approval Order, the Receiver hereby appoints the Agent, and the Agent hereby agrees to serve, as the Receiver's agent for the limited purpose of conducting the Auction pursuant to the terms of this Agreement. The Agent, in a commercially reasonable manner and otherwise in accordance with the terms and conditions of this Agreement, shall sell the Assets for and on behalf of the Receiver, by way

4

of public auction or auctions (the "Auction") to be held by no later than forty-five (45) days after the granting of the Authorization and Approval Order or such later date as may be agreed to in writing by the Receiver and the Agent (the "Auction Date") or, by private sale from the Debtor's Premises, but in any event by no later than the expiry of the Liquidation Period. The Agent further agrees that payment terms for all sales shall be by cash, debit, credit card, wire transfer, certified cheque or a draft drawn on a major Canadian bank. Without limiting any other rights of the Receiver pursuant to this Agreement but subject to the Agent's rights under Section 2.3(2) the Agent acknowledges that the Receiver may, in its sole and unfettered discretion, exclude any Assets made available for sale from the Transaction if such Assets are subject to an actual, pending or threatened claim by a third party or if the Receiver determines in its sole and unfettered discretion, such Assets are damaged, provided the Receiver provides written notice to the Agent describing such Assets prior to the sale of such Assets by the Agent.

2.2 As Is, Where Is

The Agent acknowledges that it has made such inspections of the Assets as it deems appropriate and it is understood that the Assets are to be sold by the Agent on an "as is, where is" basis, at the Agent's and purchaser's own risk, and that the Receiver has not and will not make any representation, warranty or condition, whether statutory (including, without limitation, under the Sale of Goods Act or similar legislation), express or implied, oral or written, legal, equitable, collateral or otherwise, as to title, encumbrances, fitness for purpose, marketability, condition, quantity or quality thereof or in respect of any other matter or thing whatsoever. Any descriptions provided to purchasers or prospective purchasers by the Agent's sales staff, which may include former employees of the Debtor hired by the Agent, shall be given solely on behalf of the Agent in its own capacity and not as agent for the Receiver, and the Receiver shall have no obligations with respect thereto. The Agent agrees that the auction terms will provide that all sales are final, and that signs will be prominently displayed at the Debtor's Premises advising purchasers that all sales are final.

2.3 Net Minimum Guarantee

- (1) The Agent guarantees that the proceeds (net of any Sales Taxes and Buyer's Premium collected) to be paid to the Receiver for the Receiver's account as a result of the sale of the Assets shall not be less than \$18,700,000.00, subject to adjustment as set out in Section 2.3(2) (the "Net Minimum Guaranteed Amount"), notwithstanding that the actual proceeds generated from the sale of the Assets may be less than the Net Minimum Guaranteed Amount.
- (2) If the Receiver includes any property of the Debtor that is not an "Asset" or excludes any Assets otherwise to be made available for sale from the Transaction pursuant to Section 2.1, the Net Minimum Guaranteed Amount shall be increased or reduced, as applicable, by the lesser of (i) 100% of the average of the "low" and "high" estimated value for such Asset set out in the appraisal of substantially all of the material Assets provided by the Agent on June 17, 2015 (the "Appraisal"), and (ii) the actual proceeds of the sale of such Asset. In the event that the Appraisal does not include an appraised value for the subject assets, the Agent and the Receiver shall negotiate in good faith in an attempt to reach agreement as to such adjustments. Except in the limited circumstances set out in this Section 2.3(2) determined by the Receiver in its sole

23990382.8

discretion, there shall be no downward adjustment to the Net Minimum Guaranteed Amount for any reasons whatsoever.

2.4 Payment of Net Minimum Guaranteed Amount

- (1) Upon or within two (2) Business Days of the execution of this Agreement and in any event prior to 3:30 p.m. Calgary time on June 29, 2015, the Agent shall pay the Receiver an amount equal to twenty percent (20%) of the Net Minimum Guaranteed Amount (the "Deposit"). The Deposit (and any interest accrued thereon) shall be credited as a partial payment of the Net Minimum Guaranteed Amount and will be immediately released to the Receiver upon issuance of the Authorization and Approval Order. The Deposit shall, at all times prior to its release or return in accordance with the terms of this Agreement, be held by the Receiver in escrow in a segregated interest bearing account and in accordance with the terms of this Agreement, and, except for interest collected on the Deposit, no other money or funds shall be commingled in such account.
- (2) The Agent acknowledges that the Receiver intends to, at the same time as making the application for the Authorization and Approval Order, make an application to the Court to distribute the Deposit and all other amounts payable to the Receiver under this Agreement (collectively, the "Distributions") to Wells Fargo Bank, National Association, as administrative agent for Debtor's senior secured lenders ("Wells Fargo") or such other persons entitled to receive any portion of the Distributions in priority to Wells Fargo. The Agent agrees that it shall have no claims against the Receiver or Wells Fargo and no right to trace all or any portion of the Distributions from the Receiver or Wells Fargo. The Agent further agrees that it shall not, at any hearing for the purpose of obtaining Court approval for a distribution by the Receiver of all or any portion of the Distributions, protest the distribution of all or any portion of the Distributions to Wells Fargo regardless of the nature of any claim or priority to the Distributions that the Agent may assert.
- (3) In the event that (i) this Agreement is not approved by the Court on terms satisfactory to the Agent, acting reasonably, at the time of the initial application to the Court as contemplated in Section 4.1(a); or (ii) a Superior Bid is approved by the Court in lieu of this Agreement, as contemplated herein; or (iii) the Authorization and Approval Order is not issued by the Court within ten (10) Business Days after the Court hears the Receiver's motion seeking the Authorization and Approval Order on the approval hearing date contemplated in the Bid Solicitation Process, then the Deposit (and any interest accrued thereon) shall be returned in full to the Agent in accordance with Section 2.20 and the Agreement shall be deemed to be terminated. If the Agent commits any material breach of its obligations under this Agreement that is not cured within five (5) Business Days of receiving written notice of the breach by the Receiver, then the Agent shall have forfeited the Deposit to the Receiver. Delivery of the Deposit to the Receiver in accordance with the foregoing shall not constitute full compensation of any and all losses and expenses incurred by the Receiver, shall not constitute liquidated damages, and Receiver reserves the right to pursue any and all other remedies available at law or equity.
- (4) The balance of the Net Minimum Guaranteed Amount and the Receiver's Recovery Amount, if any, shall be paid by the Agent to the Receiver as part of the Weekly Sale Reconciliation in accordance with Section 2.6. The Agent shall have paid the balance of the Net

6

Minimum Guaranteed Amount to the Receiver before the Agent shall be entitled to receive any payments or amounts from the proceeds of the Assets other than the Buyer's Premium actually collected. Notwithstanding any other term or condition or the failure of the Auction to generate any proceeds, the Agent shall pay the balance of the Net Minimum Guaranteed Amount and the Receiver's Recovery Amount, if any, to the Receiver no later than the earlier of (a) ten (10) Business Days after the completion of the auction, and (b) September 25, 2015.

(5) All monies payable to the Receiver shall be paid to the Receiver by certified cheque, wire transfer or draft drawn on a chartered Canadian bank.

2.5 Buyer's Premium

The Agent shall be entitled to charge the Buyer's Premium to purchasers upon the sale of the Assets. The amounts payable to the Agent on account of Buyer's Premium shall be retained and paid over by the Receiver and the Agent in accordance with this Agreement.

2.6 Trust Account

- (1) The Agent and the Receiver shall establish one or more joint accounts with a Canadian chartered bank used exclusively in connection with receipts and disbursements made in accordance with this Agreement (the "Account"). The Account shall be in the name of both the Agent and the Receiver. All proceeds (including Sales Taxes, Buyer's Premium, and any deposits paid to the Agent by purchasers) from the sale of Assets, whether by auction or private sale, shall be deposited on a daily basis to the Account. Disbursements from the Account, including for remittance of Sales Taxes, may only be made by cheque or wire transfer signed by the Receiver and the Agent. The Agent and the Receiver will distribute all funds on deposit in the Account representing proceeds from the sale of Assets as follows:
 - (a) to the Receiver, until the Receiver receives payment of the Net Minimum Guaranteed Amount (net of Sales Taxes but after crediting the Deposit and all interest earned thereon);
 - (b) thereafter, to the Agent until the Agent has received the Agent's Expense Reimbursement and the Buyer's Premium net of Sales Taxes;
 - (c) thereafter, the Receiver shall receive 100 % of such funds, net of Sales Taxes.
- (2) The Agent shall distribute the funds owed to the Receiver hereunder without setoff, deduction or withholding of any kind, on a regular basis, but not less than weekly.

2.7 Accounting for Sale Proceeds

(1) On each Wednesday during the Liquidation Period, commencing on the second Wednesday after the Commencement Date (the "Weekly Sale Reconciliation"), the Agent and the Receiver shall cooperate to reconcile all receipts, Sales Taxes, Buyer's Premium collected from the sale of Assets, the Agent's Expense Reimbursement, and such other sale-related items as either party shall reasonably request (collectively, "Accounting Items"), in each case for the prior week or partial week (i.e. Sunday through Saturday), all pursuant to procedures agreed

LEGS1-23930382.8

upon by the Receiver and the Agent. Proceeds shall be paid to the Receiver and the Agent weekly on the last Business Day of the week by wire transfer, commencing on the second Friday after the Commencement Date. Proceeds shall be disbursed to the Receiver and the Agent in accordance with Section 2.6.

- (2) The Agent shall provide the Receiver with a complete and detailed final accounting with respect to the sale of all Assets and other Accounting Items within twenty-one (21) days following the end of the Liquidation Period (the "Accounting Deadline").
- (3) The Agent acknowledges that the Receiver intends to share with Wells Fargo and potentially other secured creditors of the Debtor all weekly reconciliations and the final accounting provided for in Subsections (1) and (2) above and the Agent hereby consents to the Receiver's disclosure of such information to Wells Fargo and such other secured creditors.

2.8 Access to Debtor's Premises

Subject to Section 2.9, the Receiver shall ensure that the Agent has access to the Assets and Debtor's Premises, without charge during the Liquidation Period, for the purposes of carrying out the Transaction as contemplated under this Agreement during the Liquidation Period. For greater certainty and subject to Section 2.9, during the Liquidation Period, the Agent shall have rent-free use of the Debtor's Premises.

2.9 Obligations with Respect to Debtor's Premises and Assets

- (1) The Agent shall deliver vacant possession of each of the Debtor's Premises on or before the Termination Date.
- (2) The Agent agrees to leave the Debtor's Premises in a clean and broom-swept condition and shall be responsible for all costs of clean-up of the Debtor's Premises. Notwithstanding the foregoing, the Agent shall not be responsible for the costs of: (i) removal, clean-up or disposition of any of the Debtor's books and records, (ii) the costs of waste disposal bins and dumping charges, and (iii) any environmentally hazardous chemicals or substances found at the Debtor's Premises, save and except to the extent that any discharge or spill of any environmentally hazardous chemicals or substances is caused by the Agent or Persons for whom it is responsible.
- (3) The Agent agrees to act in a prudent manner while at the Debtor's Premises and the Debtor's Premises and Assets shall be maintained by the Agent in the same state of repair as existed as at the commencement of the Liquidation Period, reasonable wear and tear excepted. The Agent undertakes to restore the Debtor's Premises to their condition as at the commencement of the Liquidation Period and to repair any damages caused by the Agent, its invitees or anyone for whom it is in law responsible during the Liquidation Period at its sole expense forthwith but in any event, before the expiry of the Liquidation Period. For greater certainty, the Agent shall not be responsible for repairing any damages existing prior to the Commencement Date, except any such damages caused by the Agent, its invitees or anyone for whom it is in law responsible prior to the Commencement Date.
- (4) The Receiver and the Agent, acting reasonably, agree to conduct periodic inspections of each of the Debtor's Premises to identify any damage caused after the commencement of the

LEGAL 23990382.8 14330747.3 Liquidation Period and, without limiting the foregoing, in respect of each of the Debtor's Premises the Receiver and Agent shall conduct an interim inspection by not later than three (3) days prior to the date that the Agent is required to vacate such premises in accordance with Section 2.9(1) (the "Interim Inspection"). The Agent shall immediately conduct or cause to be conducted all repairs, restoration and clean-up of the Debtor's Premises identified as being required pursuant to the Interim Inspection and complete such work prior to the expiry of the Liquidation Period. After the Interim Inspection and on the final day of the Agent's intended occupancy of the applicable Debtor's Premises in accordance with Section 2.9(1), the Receiver and the Agent shall conduct a final inspection of the condition of each of the Debtor's Premises noting and identifying any deficiencies or damage to the Debtor's Premises caused during the Liquidation Period that has not been repaired, restored or completed by the Agent to the standard required pursuant to this Section 2.9.

- (5) The Agent agrees to maintain the Assets in a prudent manner and the Agent undertakes to repair any damages caused to the Assets (reasonable wear and tear exempted) by the Agent, its invitees or anyone for whom it is in law responsible during the Liquidation Period at its sole expense forthwith but in any event, before the expiry of the Liquidation Period. For greater certainty, the Agent shall not be responsible for repairing any damage to the Assets existing prior to the Commencement Date, except any such damages caused by the Agent, its invitees or anyone for whom it is in law responsible prior to the Commencement Date.
- (6) The Agent further agrees to indemnify, defend and hold the Receiver and the Debtor and their employees, agents and representatives (collectively, the "Indemnified Parties") harmless from and against all claims for damages, losses, injury or costs resulting from a breach of its obligations under this section and for damages, losses or injury caused to property or persons through the actions or negligence of the Agent, its invitees or anyone for whom it is in law responsible.

2.10 Payment of Occupancy Costs

The Receiver shall be responsible to pay when due all of the following costs and expenses to the extent referable to the Liquidation Period:

- (a) security, rent and other occupancy costs relating to the Debtor's Premises;
- (b) all utilities and other reasonable overhead expenses, including trash and dumpster services, relating to the Debtor's Premises;
- costs and expenses to move all of the Assets to the three Debtor's Premises, for purposes of the Auction;
- (d) pro-rated property taxes in respect of the Debtor's Premises; and
- (e) all Make-Ready Repair Costs.

2.11 Expenses

Other than those expenses which are the responsibility of the Receiver as set out in Section 2.10 and elsewhere in this Agreement, the Agent shall be responsible for all expenses incurred in connection with the Transaction, including without limitation, costs associated with long distance telephone costs, postage, courier services, supplies needed by the Agent, bank and credit or debit card charges, cheque verification services and personnel hired or provided by the Agent. The Agent shall reimburse the Receiver to the extent that such expenses are incurred by the Agent and paid by the Receiver and such amounts paid to the Receiver are in addition to and shall not reduce or otherwise affect the amount of the Net Minimum Guaranteed Amount or the Receiver's entitlement pursuant to Section 2.6. Notwithstanding the foregoing, the Agent shall be entitled to receive payment of the Agent's Expense Reimbursement from the proceeds of the Asset sales in accordance with Section 2.6.

2.12 Personnel

The Agent shall be responsible at its own cost for providing competent personnel to prepare for and perform all tasks relating to the Transaction. It is understood that the Receiver is not and will not in any event be an employer of any such personnel or liable to pay any amounts to or with respect to such personnel. The Receiver does not assume any responsibilities whatsoever with respect to the continuation of the employment of any existing employees of the Debtor (the "Employees") pursuant to this Section. In particular, the Receiver does not warrant or guarantee that the employment of any Employees can be continued for all or any part of the Liquidation Period.

2.13 Trade Names

The Agent shall be permitted and is hereby granted a non-exclusive, temporary licence to use the name Big Eagle and make reference to "Deloitte Restructuring Inc. as Receiver of "Big Eagle" and to the receivership of the Debtor solely for the purposes of this Agreement and any advertising pursuant to Section 2.14 during the Liquidation Period.

2.14 Advertising

The Agent, at its own expense, shall advertise and otherwise promote the liquidation and auction of the Assets by all appropriate means, as approved by the Receiver, in order to give adequate exposure to the Assets to the maximum number of potential purchasers. All advertising copy is to be submitted initially to the Receiver for approval (which approval shall not be unreasonably withheld) not less than five (5) Business Days prior to its first publication and use by the Agent. If the Receiver has not objected to the Agent's proposed advertising within two (2) Business Days after receipt of the Agent's proposal, the Receiver will be deemed to have approved such advertising. The Agent shall be responsible for paying all costs of advertising.

2.15 Authorizations and Remittance of Taxes

The Agent shall be responsible for ensuring that all necessary governmental or other approvals, permits or authorizations are obtained in order to conduct its liquidation program in compliance with all applicable laws. In addition, the Agent shall ensure that all applicable taxes

LEGAL 23990382.8 14330747.3 and duties including, without limitation, Goods and Services Tax and provincial retail sales tax (collectively, the "Sales Taxes") are collected and remitted to the proper authorities when due. The Agent agrees to indemnify and save the Receiver harmless from and against all claims for payment of the above-mentioned taxes including penalties and interest thereon and any liability or costs incurred as a result of any failure to pay such taxes when due and all claims arising out of any failure to obtain all necessary government or other approvals, permits or authorizations.

2.16 Insurance

- (1) The Agent will be responsible for arranging third-party liability insurance with respect to the Agent's access to and use of the Debtor's Premises during the Liquidation Period and shall be responsible for the costs of such insurance. The third party liability insurance shall provide for not less than \$2,000,000.00 coverage per occurrence. The Agent shall provide proof of such insurance to the Receiver.
- (2) The Receiver shall maintain the existing insurance coverage against loss of or damage to the Assets and shall be responsible for the costs of such insurance until the earlier of (i) the date the Agent has vacated the Debtor's Premises, and (ii) the expiry of the Liquidation Period.
- (3) In the event that all or any material part of the Assets are destroyed or damaged by fire or other casualty or stolen at any time prior to the sale of such Assets, the insurance proceeds attributable to such damaged Assets shall be deemed to be the proceeds of the sale of such Assets for the purposes of this Agreement and, for greater certainty, shall be included for the purposes of distributing funds in accordance with Section 2.6. The Receiver and the Agent agree not to settle any insurance claim without the prior written consent of the other, such consent not to be unreasonably withheld.

2.17 Extension of Credit

In the event that the Agent extends credit (other than by way of credit cards) to any purchaser, the Agent will be responsible for all related credit risks and costs thereof.

2.18 Extended Sales

In the event there are any unsold Assets remaining at the end of the Liquidation Period, the Agent shall, at its own cost, move such Assets to a location or locations to be agreed upon with the Receiver for subsequent sale and the proceeds (net of any Sales Taxes collected) shall be treated as sale proceeds in accordance with Section 2.6 hereof. Once the Agent has paid to the Receiver the Net Minimum Guaranteed Amount, the Receiver and the Agent may agree that the Receiver shall transfer title in and to any unsold Asset to the Agent. The form of any Bill of Sale to be delivered to the Agent in the event that the Receiver exercises this option shall be in the form attached hereto as Schedulc 2.18. The Authorization and Approval Order shall provide that title to any such remaining Assets sold to the Agent shall vest in the Agent free and clear.

2.19 Movement of Assets

The Agent shall not remove any Assets from the Debtor's Premises prior to sale without the prior written consent of the Receiver, which consent shall not be unreasonably withheld. The

11

Agent agrees to indemnify, defend and hold the Indemnified Parties harmless from and against any losses, damages, costs or claims caused by or resulting from the removal or transportation of the Assets. The Agent shall be responsible for the costs of any such move, including without limitation, any insurance, permit or licence costs.

2.20 Break-Up Fee; Superior Bid; Appraisal Costs

This Agreement is subject to other offers presented to the Receiver in accordance with the Bid Solicitation Process. In the event that the Court determines that an offer submitted pursuant to the Bid Solicitation Process is a Superior Bid (as defined below) and approves such offer in lieu of this Agreement, then Receiver shall within one (1) Business Day return the Deposit (and any interest accrued thereon) to the Agent and pay to the Agent, by wire transfer of immediately available funds, a break-up fee in the amount of one percent (1%) of the Net Minimum Guaranteed Amount under this Agreement (the "Break-Up Fee") and the Appraisal Costs. The Break-Up Fee, the return of the Deposit and a reimbursement of the Appraisal Costs by the Receiver shall be the Agent's sole remedy as a result of the Receiver proceeding with a Superior Bid in lieu of the Transactions contemplated by this Agreement. For purposes of this Agreement, a "Superior Bid" shall mean (1) in the case of a superior net minimum guarantee auction proposal, a credible offer that (i) is a higher or better bid than set out in this Agreement provided that the net minimum guaranteed amount in such bid exceeds the Net Minimum Guaranteed Amount in this Agreement by not less than the amount of the Break-Up Fee plus \$200,000 with no materially unfavourable changes having been made to the other economic terms in this Agreement (ii) made by a recognized, reputable and credit worthy North American liquidator, (iii) otherwise incorporates all or substantially all of the terms and conditions of this Agreement, and (iv) otherwise complies with the Bid Solicitation Process, and (2) in the case of a superior Asset purchase bid, a credible offer to acquire all or substantially all of the Assets (i) that is an all cash offer the Receiver reasonably believes should realize net cash proceeds greater than the net cash proceeds reasonably expected to be realized by the Transaction, (ii) made by a creditworthy bidder determined by the Receiver to be qualified to participate in the process by the Receiver, (iii) is on a "as is, where is" basis with no more onerous representations or warranties by the Receiver than contained in this Agreement and contains no material holdbacks or downward purchase price adjustments that could lead to a lower realization of net proceeds than contemplated by this Agreement, (iv) is reasonably expected by the Receiver to be completed prior to the Termination Date set out in this Agreement.

2.21 Auction

In the event that the Receiver receives one or more Superior Bid(s) prior to the bid deadline set out in the Bid Solicitation Process, the Receiver shall invite the bidder(s) that submitted such Superior Bid(s) and the Agent to an auction and such auction shall be conducted by the Receiver at the Receiver's offices on July 17, 2015 in accordance with auction procedures to be established by Receiver.

12.

SECTION 3 - REPRESENTATIONS AND WARRANTIES

3.1 Representation and Warranty of the Receiver

The Receiver represents and warrants to the Agent that it has been duly appointed as the and receiver and manager of the property, assets and undertaking of the Debtor pursuant to the Receivership Order and, subject to the granting of the Authorization and Approval Order, will have the right to enter into and carry out its obligations under this Agreement,

3.2 Representation and Warranty of the Agent

The Agent represents and warrants to the Receiver that it has full right, power and authority to enter into and carry out its obligations under this Agreement and this Agreement has been duly and validly authorized, executed and delivered by the Agent.

SECTION 4 – CONDITIONS

4.1 Mutual Conditions

The obligations of the Receiver and the Agent are subject to the conditions that:

- (a) on or before July 6, 2015 an order in form and substance acceptable to the Receiver and the Agent, each acting reasonably, shall have been made by the Court approving (i) and authorizing the Receiver entering into this Agreement and approving this Agreement as the stalking-horse agreement for the purposes of the Bid Solicitation Process, (ii) the Bid Solicitation Process, the Break-up Fee and Appraisal Cost reimbursement and (iii) a court ordered charge to secure the payment of the Break-up Fee and the Appraisal Cost reimbursement by the Receiver ranking in priority to all other liens, encumbrances charges other than the Receiver's Charge and the Receiver's Borrowing Charge as defined in the Receivership Order;
- (b) after the conclusion of the Bid Solicitation Process and on or before the commencement of the Liquidation Period, an order in form and substance acceptable to the Receiver and the Agent, each acting reasonably, shall have been made by the Court (i) declaring this Agreement to be the successful bid (ii) authorizing and approving this Agreement and the Transaction, (ii) providing that upon the Agent completing the sale of any of the Assets to a purchaser, in accordance with the terms of the this Agreement, and delivering a bill of sale to the purchaser (the "Purchaser Bill of Sale"), all right, title and interest of the Debtor in and to the asset(s) described in the Purchaser Bill of Sale shall be vested in such purchaser, free and clear of all claims and encumbrances, and (iii) approving and authorizing, at the Receiver's election, the vesting of any remaining unsold Assets in and to the Agent pursuant to Section 2.18 (Extended Sales) (the "Authorization and Approval Order") and such order shall have been obtained by no later than July 29, 2015; and

13

(c) as at the commencement of the Liquidation Period, the Authorization and Approval Order shall not have been stayed, varied or vacated, and no order shall have been issued and no action or proceeding shall be pending to restrain or prohibit the completion of the Transaction.

If the conditions set out above are not satisfied by the time provided therefore (or have not been waived by the Receiver or the Agent as applicable), then this Agreement shall be terminated without any penalty or liability whatsoever to the Receiver or the Agent, other than the return of the Deposit and Appraisal Costs reimbursement by the Receiver to the Agent, but without cost or other compensation.

SECTION 5 - GENERAL

5.1 Notices

Any demand, notice or other communication to be given in connection with this Agreement shall be given in writing and shall be given by personal delivery (in which case it shall be left with a responsible officer of the recipient) or by electronic communication addressed to the recipient as follows:

in the case of the Receiver:

Deloitte Restructuring Inc.
700, 850 - 2 Street SW
Calgary, Alberta T2P OR8
Attention: Jeff Keeble, CA, CIRP, CBV
Email: jkeeble@deloitte.ca

Tel No.: 403-503-1458

with a copy to:

McMillan LLP

in the case of the Agent;

Attention: Adam Maerov Telephone No: (403) 215-2752 Facsimile No.: (403) 531-4720

Great American Global Partners, LLC 21860 Burbank Blvd. #300 Woodland Hills, California 91367 Attention: Adam Alexander Telephone No: (818) 884 3737

with a copy to: 590 Madison Avenue, 29th Floor New York, New York 10022 Attention: Alan N. Forman
Telephone No: (212) 457-9947
and with a further copy to:
Miller Thomson LLP
Suite 5800 – 40 King Street West
Toronto, Ontario
M5H 3S1
Attention: Jeffrey Carhart
Telephone No: (416)595-8615

Facsimile No: (416) 595-8695

or to such other address, individual or electronic communication number as may be designated by notice given by either party to the other. Any demand, notice or other communication shall be conclusively deemed to have been given, if given by personal delivery, on the day of actual delivery thereof and, if given by electronic communication, on the day of transmittal thereof if transmitted during normal business hours of the recipient on a Business Day and on the Business

Day following the transmittal thereof if not so transmitted.

5.2 Disputes

In the event any dispute arises between the parties concerning the subject matter of this Agreement, the parties agree that the Court shall have exclusive jurisdiction to resolve any and all such disputes. Prior to filing any motion, application with the Court or commencing any proceeding the moving party shall issue a written notice (with all relevant particulars of the dispute) of its intention to do so five (5) Business Days prior to taking any steps to file or commence any such motion, application or proceeding, as applicable. Such written notice will not replace the required notice period under the rules of civil procedure of Alberta.

5.3 Time of Essence

Time shall, in all respects, be of the essence hereof, provided that the time for doing or completing any matter provided for herein may be extended or abridged by an agreement in writing signed by the Receiver and the Agent or by their respective solicitors.

5.4 Currency

All references herein to money amounts are in Canadian currency.

5.5 Agreement Costs

The parties agree to bear their own respective legal and other expenses for preparing, negotiating and executing this Agreement and any related documents.

5.6 Further Assurances

Each party shall, at the other party's expense, from time to time execute and deliver, or cause to be executed and delivered, all such documents and instruments and do, or cause to be done, all such acts and things as the other party may, either before or after the Transaction,

LEGAL 23990382.8 14330747.3 reasonably require to effectively carry out or better evidence or perfect the full intent and meaning of this Agreement. Without limitation, the Receiver will use reasonable efforts to provide all documentation necessary to evidence conveyance of title to all motor vehicles which are included in the Assets.

5.7 Obligations to Survive

The obligations, representations and warranties of the parties hereto shall survive the completion of the Transaction.

5.8 Entire Agreement

This Agreement constitutes the only agreement between the parties with respect to the subject matter hereof and supersedes any and all prior negotiations and understandings. No amendment of this Agreement shall be binding unless in writing and signed by the parties. No waiver by a party of any breach of this Agreement shall take effect or be binding upon the party unless it is in writing and signed by the party and, unless otherwise expressly stated therein, any such waiver shall be limited to the specific breach waived.

5.9 Governing Law

This Agreement shall be governed by and construed in accordance with the laws of the Province of Alberta and the laws of Canada applicable therein and each of the parties hereby irrevocably attorns to the exclusive jurisdiction of the Courts of the Province of Alberta.

5.10 Benefit of Agreement

This Agreement shall be binding upon and enure to the benefit of the parties hereto and their respective successors and permitted assigns; provided that the Agent shall not assign the benefit of this Agreement (other than to an affiliate of the Agent) without the prior written consent of the Receiver. In the event that Agent assigns this Agreement to an affiliate, the Agent shall remain jointly and severally liable hereunder with such affiliate assignee.

5.11 Severability

If any provision of this Agreement or any document delivered in connection with this Agreement is partially or completely invalid or unenforceable, the invalidity or unenforceability of that provision shall not affect the validity or enforceability of any other provision of this Agreement, all of which shall be construed and enforced as if that invalid or unenforceable provision were omitted. The invalidity or unenforceability of any provision in one jurisdiction shall not affect such provisions validity or enforceability in any other jurisdiction.

5.12 Capacity

Deloitte Restructuring Inc. is acting solely in its capacity as Receiver of Big Eagle Hydro-Vac Inc., Big Eagle Limited Partnership, Big Eagle Hydro-Vac (North) Inc., and Big Eagle (North) Limited Partnership and shall have no corporate or personal liability under this Agreement or for any other matter whatsoever relating to the Transaction.

L,23990382.8 16

5.13 Agency Relationship

The Agent acknowledges that it will not hold itself out as agent of the Receiver except as specifically provided for in this Agreement and that the Agent's authority as agent for the Receiver is limited to the powers specifically provided for in this Agreement.

5.14 Counterparts Electronic Execution and Transmission

This Agreement may be executed in counterparts and delivered via facsimile or pdf via email or other electronic transmission, and all counterparts, when taken together, shall constitute one Agreement and electronic transmission of an executed copy of this Agreement will be deemed to be receipt of an original.

[SIGNATURE PAGE FOLLOWS]

17

The parties have executed this Agreement.

DELOITTE RESTRUCTURING INC., in its capacity as Receiver of the property, assets and undertaking of the Debtor and not in its personal capacity

By: Name: SEFF Kresie
Title: SEFFE VILE (REFLOE-F

GREAT AMERICAN GLOBAL PARTNERS, LLC

By:
Name:
Title:

The parties have executed this Agreement.

DELOITTE RESTRUCTURING INC., in its capacity as Receiver of the property, assets and undertaking of the Debtor and not in its personal capacity

By:
Name:
Title:

GREAT AMERICAN GLOBAL PARTNERS, LEG

Bv:

Title

Lanaging Member

Schedule 1.1(8) - Assets

Please see attached.

14283795;4

Equip #	Year	Make	Description	. Moreoviero
NAT001	1 9 74	RUSSELL	T/A ALUM NITROGEN TANK TRAILER, CGA-341, 61350-LB CAPACITY, TANK S/N 976, INSULATED, 4	ED
			CYLINDER KUBOTA DIESEL ENGINE, TURBO PUMP, 11R22.5 TIRES-ALUM BUDS, (LIC V510-93)	
NAT002	1977	RUSSELL	ALUM TANDEM-AXLE NITROGEN TRAILER, TIRES 295/75R22.5, ALUM BUDS, S/N NAT00267643 (LIC V510-94) CGA-341, 68,209-LB TANK	
G040	1981	CHAMPION	MOTOR GRADER	ED
CP001	1981	MACK	PICKER TRUCK, 12.5-TON	· ED
SS002	1988	BOBCAT	SKID STEER LOADER, TIRES 12-16.5 NHS	MEDICINE
TWP051	1990	WABASH	T/A WATER PUP, TIRES 11R24.5-STEEL BUDS, NON-SPEC STEEL TANK, (LIC F77605), 12,000-LITRE	HAT MEDICINE
Q\$013	1990	WEST TANK	CAPACITY WEST TANK QUAD WAGON TANK PUP TRAILER, TIRES 11R22.5, STEEL BUDS, SPRING SUSPENSION,	HAT SW{FT
		ļ	TC306, 32M3 CAP, S/N 2WWTA28DXLR000076	CURRENT
Q\$014	1990	WEST TANK	WEST TANK QUAD WAGON TANK PUP TRAILER, TC306 CRUDE, 32M3 CAPACITY, TIRES 11R24.5,	MH
			DAYTONS, SPRING SUSPENSION, FULL TURN TABLE, (LIC V547-89) [NOT IN SERVICE AT TIME OF INSPECTION]	
QS009	1993	BEALL	BEALL TANK QUAD WAGON PUP TRAILER, MC406, 34M3 CAPACITY, TIRES 11R24.5, ALUM BUDS, SPRING SUSPENSION, (LIC V547-84)	MH
RN066	1994	ROUGHNECK	TANDEM-AXLE RV TRAILER, PROPANE PIG	ED
WH028	1994	FREIGHTLINER	WATER TRUCK, TANDEM AXLE, FRONT TIRES 12R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS. (LIC G10818)	ED
TWPS001	1994	WABASH	TRI/AXLE TANK PUP TRAILER, 160-BARREL, TIRES 11R24.5-ALUM BUDS, AIR RIDE SUSPENSION	MEDICINE HAT
TWPS018	1994	WABASH	T/A STIFF POLE ALUM WATER TANK PUP TRAILER, TIRES 11R24.5, STEEL DAYTON, SPRING SUSPENSION, C/W ALUM TANK, TC30616M3	SWIFT
WK018	1995	MESTERN STAR	T/A WATER TRUCK, CAT 3406E 435-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	CURRENT
TVRDIE .	1220	TWESTERN STAN	TIRES 11R24.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, CHALMERS AIR RIDE SUSPENSION,	ZAMA CITY
			C/W HAMMS 100-BARREL STEEL TANK S/N 030205278, TC WATER, 3" FLUID PUMP, CAT 3406E 435-	
			HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 11R24.5-ALUM BUDS, REAR	
			TIRES 11R24.5-ALUM BUDS, CHALMERS AIR RIDE SUSPENSION, C/W HAMMS 100 BARREL STEEL	
			TANK S/N 030705278, TC WATER 3" FILIID PILMP	
QS002	1995	ADVANCE	QUAD PUP, TC306 CRUDE, TIRES 11R24.5-STEEL BUDS, SPRING SUSPENSION, 2 COMPARTMENT,	SWIFT
опоот	400=		15,000-LITRE, 10,000-LITRE CAPACITY, FULL TURN TABLE, (LIC V547-77)	CURRENT
CC095	1995	N/A	DA MAN TANDEM-AXLE CATCH TANK, 10M3, 7.5R16 TIRES	
UT004	1996	N/A	CERTIFIED UTILITY T/A TRAILER, 18' WOOD DECK, 245/75R16 TIRES, CUSTOM MANUFACTURED	SC
CSC002	1997	N/A	SIDE SPOOLER CUSTOM BUILT, SKID MOUNTED	ED
TWPS008	1998	POLAR	TRIDEM ALUM TANK TRAILER, 34M3 CAP, 11R24.5, ALUM BUDS, SPRING SUSPENSION	SWIFT
				CURRENT
TWPS016	1998	ADVANCE	TRIDEM-TC306 ALUM TANK TRAILER, TIRES 11R24.5-STEEL BUDS, 37,000-LITRE CAPACITY, AIR RIDE SUSPENSION, (LIC V547-91)	SWIFT CURRENT
W043	1998	KENWORTH	T/A WATER TRUCK, CAT 3406 ENGINE, EATON FULLER 18-SPEED, (LIC F16255), C/W CUSTOM VAC	МН
			S/N 6444402, TCRN 25.437, TC 350/312, 15,609 L, 578,980-KILOMETERS INDICATED, 22,501-HOURS	
			INDICATED	
СТ099	1998	VOLVO/AUTOC	AUTOCAR TANDEM-AXLE STEER/TRI-AXLE DRIVE COILED TUBING UNIT, CAT 500-HP, 18-SPEED	ED
		AR	FULLER, 16000-LB X 2 FRONT AXLE RATING, 425/65R22.5 - FRONT, ALUM BUDS, 69000-LB REAR AXLE	
			RATING, 11R24.5 - REAR, AIR RIDE, SPICER 792, /W COILED TUBING UNIT, 3100M OF 1 1/4" TUBING,	
			RT20 INJECTOR - 3/4" TO 2", CAT 4517 TRIPLEX PUMP, 50L/MIN @ 21 MPA, W/HURRICANE MODEL	
		1	903-85 COMPRESSOR, 18M3 @ 14 MPA, ENGINE 903 CUMMINS, W/TEREX ATLAS 145CS-A4 PICKER	
			14-TON, 6.7-M REACH, SIDE OUTRIGGERS, S/N 100103	
TWPS019	1998	ADVANCE	TRIDEM TC306 ALUM TANK TRAILER 34M3 CAP, 11R24.5, STEEL DAYTON, SPRING SUSPENSION, TRI-	
		1	AXLE, 30,000-LiTRE CAPACITY	

Fouir #	Year	Make	Description	Location
CB902	1999	A STATE OF THE PARTY OF THE PAR	T/A COMBO UNIT, M11-370 ESP ENGINE, EATON FULLER 8-SPEED TRANSMISSION, 445/65R22.5	ED
C0502,	برييا	1	FRONT TIRES, ALUM BUDS, 11R24.5 REAR TIRES, ALUM BUDS, (LIC F96607), C/W CUSTOM VAC	LD
		1	SERVICES COMBO UNIT S/N 0498026, 10,862-LITRE CAPACITY, TC350, HOTSY STEAM, WATER BLAST,	
		1		
		·	BLOWER, TRIPLEX PUMP, 271,389-KILOMETERS INDICATED, 5,541-HOURS INDICATED	
TWPK004	1999	JASPER	T/A STIFF POLE TANK TRAILER PUP, TIRES 11R24.5-STEEL DAYTONS, C/W JASPER SERVICE CARGO	ED
			TANK S/N 2J9JT1827YS031104, TC307, 14,956-LITRE CAPACITY	m
FKLC003	1999	LINDE	LIFT TRUCK, 6000-LB CAPACITY, LPG, DOUBLE MAST, 192" MAXIMUM LIFTING HEIGHT, SIDE SHIFT, 5'	ED
			FORKS, PNEUMATIC TIRES, 9,996-HOURS INDICATED	
FH002	2000	INTERNATIONA	FUEL TANKER TRUCK, SINGLE AXLE, DT 466E ENGINE, 6-SPEED, FRONT TIRES 11R22.5, STEEL BUDS,	CNRL
		L	REAR TIRES 11R22.5, STEEL BUDS, C/W 1989 HUTCHINSON IND. TANK, 4 COMPARTMENTS - 4,000 L,	
			3,000 L, 2,500 E, 2,000 L, S/N 89-6386	
\$\$\$003	2000	CATERPILLAR	SKID STEER LOADER, W/ GENERAL PURPOSE BUCKET, SNOW PLOW, 1914-HOURS INDICATED	SWIFT
				CURRENT
TR\$112	2000	PETERBILT	T/A HIGHWAY TRACTOR W/ 63" STAND UP SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED	SWIFT
			TRANSMISSION, FRONT TIRES 11R24.5-ALUM BUDS, 12,000-LB AXLE, REAR TIRES 11R24.5-ALUM	CURRENT
			BUDS, 46,000-LB AXLES,238" WB, (LIC G10811), [NOT IN SERVICE AT TIME OF INSPECTION, ROLL	
			OVER DAMAGE]	
CT201	2000	VOLVO/AUTOC	AUTOCAR TANDEM FRONT/REAR-AXLE DRIVE COILED TUBING UNIT, CUMMINS N14 ENGINE, EATON	ED
		AR	FULLER 18-SPEED, 37,400-LB FRONT AXLES, FRONT TIRES 445/65R22.5, ALUM BUDS, 46,000-LB REAR	
			AXLES, REAR TIRES 11R24.5, ALUM BUDS, (LIC G20898), C/W COILED TUBING UNIT, 3100M OF 1 1/4"	
			TUBING, HURRICANE COMPRESSOR 903-84 S/N 80394, HYDRARIG RT25 INJECTOR W/ARCH	
			GOOSENECK, ATLAS CRANE KNUCKLE PICKER AK1401-A4 S/N 104700, HYDRAULIC SPOOL, CONTROL	
			CAR_CAT_4517 TRIPI FX PLIMP	
PD118	2000	VOLVO	PICKER PILE DRIVER	ED
ATK007	2000	JASPER	TRIDEM TANK TRAILER, TIRES 11R24.5-ALUM BUDS, SPRING SUSPENSION, C/W 2 COMPARTMENT	ZAMA CITY
			STEEL TANK - 32,750-LITRE CAPACITY, DOT 412	
W033	2001	FREIGHTLINER	WATER TRUCK, T/A, CAT C12 355-HP ENGINE, EATON FULLER 13-SPEED TRANSMISSION, FRONT	ED
			TIRES 425/6SR22.5-ALUM BUDS, 14,000-LB FRONT AXLE, REAR TIRES 11R24.5-STEEL BUDS, 40,000-	
			LB REAR AXLES, AIR RIDE SUSPENSION, C/W ADVANCE WATER TANK S/N 2AESTHB042E000150,	
			TC306-MS CRUDE, 15,900-LITRE CAPACITY, BOWIE 4" PRODUCTION PUMP, 1,975-KILOMETERS	
			SHOWING (NEW GAUGE), 9 829-HOURS REPORTED	
W039	2001	WESTERN STAR	4964FX TANDEM-AXLE WATER TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED, 16000-LB FRONT	ED
			/ 23000-LB X 2 REAR AXLES, FRONT 425/65R22.5-ALUM BUDS, REAR 11R24.5-STEEL BUDS, AIR RIDE	
			SUSPENSION, C/W ADVANCE CRUDE TANK, 15900-LITRE, S/N 2AESTHB032E000169 [COLLISION	
			DAMAGED : SALVAGE ONLY)	
SSK004	2001	MACK	T/A SOUR SEAL TRUCK W/ 46" SLEEPER, E7 460 XT ENGINE, EATON FULLER 13-SPEED	GP
			TRANSMISSION, FRONT TIRES 315/80R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, AIR RIDE	
			SUSPENSION, C/W JASPER STEEL TANK S/N 2J9JT1837Y031017, TC307, 14,956-LITRE CAPACITY, 3"	
			FLUID PUMP	
ST200	2001	KENWORTH	HD SERVICE TRUCK W/ SLEEPER, EATON FULLER TRANSMISSION, PALFINGER 22' BOOM W/ 4,450-LB.	ED
			CAPACITY, WELDER/GENERATOR, AIR COMPRESSOR	
NP016	2001	ASPEN	NITROGEN PUMP TRAILER, INCLUDES PUMP, DRIVEN BY DIESEL ENGINE, NITROGEN TANK, AND	ED
			OPERATORS CAB, UNITIZED ON TRI-DEM TRAILER, PURCHASED USED IN 2014, APPEARS REBUILT	
			·	
ATK046	2001	JASPER	TRIDEM TANK TRAILER, TIRES 11R24.5-ALUM BUDS, SPRING SUSPENSION, C/W 2 COMPARTMENT	ZAMA CITY
			32,750-LITRE, TC407/412	
SSK045	2001	WESTERN STAR	T/A SOUR SEAL TRUCK W/ 48" SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ZAMA CITY
			FRONT TIRES 385/65R22.5-ALUM BUDS, 14,600-LB FRONT AXLE, REAR TIRES 40X31.00-20 NHS,	
			46,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W JASPER STEEL TANK, TC412, 14,900-LITRE	
			CAPACITY. 3" T&F FLUID PUMP	
V029	2001	WESTERN STAR	T/A VACUUM TRUCK, 4964SX, CAT C12, EATON FULLER 18-SPEED, 425/65R22.5 FRONT, ALUM BUDS,	MEDICINE
	l	1	48X31.00-20 FLOATATION TIRES, STEEL RIMS, TC350 CRUDE TANK, ESTIMATED 12,000-LITRE	HAT
			IGONDITURES DE L'ORIATION TRES, STEEL MINS, I COOU CRODE TANNE ESTIMATEU IZIMIATURE	

Big Eagle Limited Partnership Asset List* GA Global Partners Report

Equip. #	Year	Make	Description	horation
WS108	2001	1	T/A TANK TRUCK W/ SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 385/65R22.5-ALUM BUDS, 14,600-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB	SWIFT CURRENT
			AXLES, AIR RIDE SUSPENSION, 236" WB, PINTLE HITCH, SPRAY BAR, (LIC G10812), C/W WABASH	
			TC306 ALUM 16M3 TANK, S/N 63022121D16001275, HEATED SIDE CABINET W/HYDRAULICS, 4" T&E HYD PUMP, 517,739-KILOMETERS INDICATED, 18,918-HOURS INDICATED	
WS118	2001	KENWORTH	T/A TANK TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, 12,000-LB FRONT AXLE, FRONT	SWIFT
			TIRES 11R24.5, ALUM BUDS, 46,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, 236" WB, (LIC G10808), C/W TC406 CRUDE TANK, SPRAY BAR, PINTLE HITCH, 16M3	CURRENT
			CAP, HEATED SIDE CABINET'S W/HYDRAULICS, 4" T&E HYD PUMP, 726,868-KILOMETERS INDICATED,	
			17 721-HOURS INDICATED	
UTC332	2001	ROUGHNECK	32' TANDEM-AXLE GOOSENECK FLATDECK TRAILER, TIRES 235/85R16-STEEL BUDS, 40,000-LB AXLE	ED
V020	2001	WESTERN STAR	TANDEM-AXLE VACUUM TRUCK, CAT C12 ENGINE, EATON FULLER 18-SPEED, 150,939-KILOMETERS	GP
			INDICATED, 20,685-HOURS INDICATED, FRONT AXLE 425/65R22.5 ALUM BUDS 16,000-LB, REAR	
			AXLES 48X31.00-20 STEEL BUDS 46,000-LB, C/W VACUUM TECH UNIT TC 350, TCRN 25-383, 13,229 L	
STD209	2001	FORD	CAPACITY S/N 1000029. HIBON BLOWER SERVICE TRUCK, 4,X4, DUALLY, DIESEL ENGINE, 6-SPEED TRANSMISSION, TIRES 245/70R19.5, C/W	ED
0,0200	2002	1 3.12	GENERAL BODY & EQUIPMENT LTD. 11'X8' SERVICE BODY W/ LIFTMORE 3200 10,000-LB TRUCK	
			CRANE, UMAC UNDERHOOD ROTARY AIR COMPRESSOR	
SSK005	2001	WESTERN STAR	T/A SOUR SEAL TRUCK W/ 36" SLEEPER, CUMMINS N14 ENGINE, EATON FULLER 18-SPEED	
			TRANSMISSION, FRONT TIRES 385/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, C/W JASPER STEEL CARGO TANK S/N	
			219/T18371S031073, TC307, 14.956-LITRE CAPACITY	·
CC201	2001	C-TECH	CATCH TANK, TANDEM-AXLE, 7.5R16 TIRES, 10000-LB X 2 AXLES	***
TWPK003	2002	JASPER	TRIDEM TANK TRAILER, TIRES 11R24.5-ALUM BUDS, SPRING SUSPENSION, S/N 2J9JT00371S031001,	ZAMA CITY
BT001	2002		C/W 2 COMPARTMENT STEEL TANK, 32,750-LITRE CAPACITY, DOT 412 TRAILER, FLATBED, 24' WOOD DECK, TANDEM-AXLE	MED(CINE
PIOOT	2002	MANUFACTUR	TRAILER, FLATBED, 24 WOOD DECK, TAINDENT-AALE	HAT
		ED		
VR101	2002	KENWORTH	T/A VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	MEDICINE
			TIRES 385/65R22.5-ALUM BUDS,19,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-	HAT
		:	LB REAR AXLES, CHALMERS SUSPENSION, C/W PROXY WELD INDUSTRIES S/N 03-0611, TC407/412, 13.033-LITRE CAPACITY, HIBON VTB 800 BLOWER	
RN034	2002	ROUGHNECK	T/A 28' RV TRAILER, TIRES 235/85R16, COMPLETE	ED
RV039	2002 ′	CITATION	TRAVEL TRAILER, 26', CREW SHACK	ED
RN027	2002	ROUGHNECK	28' T/A RV TRAILER, TIRES £T235/85R16-50%, FURNACE, WATER HEATER, RANGE, REFRIGERATOR,	
B1 (0=B	0007	OUTATION	A/C, TOILET, SHOWER [REPORTED AS PERMANENTLY OUT OF SERVICE]	
RV050	2002	CITATION	26' TRAVEL TRAILER, TIRES LT235/85R16, STEEL BUDS	МН
V082	2003	WESTERN STAR	TRI-AXLE VACUUM TRUCK, CAT C15, EATON FULLER 18-SPEED, 445/65R22.5 FRONT, 11R24.5 REAR,	ED
			ALUM BUDS, AIR RIDE, (LIC G-10823), C/W CUSTOM: VAC UNIT, TC412, 19193 L CAP, HIBON VTB820,	
141072	2002	MICCIEDNI CTAD	S/N 12471203, 234,152-KILOMETERS INDICATED, 22,384-HOURS INDICATED	
W072	2003	WESTERN STAR	WATER TRUCK, TANDEM	ED
PTI105	2003	INTERNATIONA	S/A PRESSURE TEST TRUCK, DT 466 ENGINE, EATON FULLER 10-SPEED ENGINE, FRONT TIRES 11R22.5-	GP
		L	ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 11R22.5-ALUM BUDS, 20,000-LB REAR AXLE, AIR	
		1	RIDE SUSPENSION, (LIC G03344), C/W INTERCONTINENTAL 20' TRUCK BODY S/N 4685,	
			BUTTERWORTH TRIPLEX PUMP, 4 ALUM TANKS, OPERATOR CONTROL ROOM [REPORTED AS OUT OF SERVICE]	
S001	2003	WESTERN STAR	T/A STEAMER UNIT TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, FRONT AXLE	CNRL
			425/65R22.5 ALUM BUDS 16,000-LB, REAR AXLES 11R22.5 ALUM BUDS 46,000-LB, C/W THERMO	
			KING VAN BODY S/N VB3190, HOTSY 9480 STEAMER, ESTIMATED 8,000-LITRE CAPACITY TANK	

				III.
Equip #	Year	Make		toration -
V055	2003	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 375-HP ENGINE, EATON FULLER 18-SPEED	ED
	· '		TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
			SERVICES S/N 19179803, TC412, 15,934-LITRE CAPACITY, HIBON BLOWER, TIRE PRESSURE	
	<u> </u>		MONITORS	
V056	2003	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB AXLES, AIR	
			RIDE SUSPENSION, (LIC F16261) C/W CUSTOM VAC SERVICES, TC SPEC 412, 19,200-LITRE CAPACITY -	
			S/N 191799-03. HIBON BLOWER- S/N 15571551	
V057	2003	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	ED
			445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK, S/N 349461-	
		i	03, 18,280 L, 271209-KILOMETERS INDICATED	
SR130	2003	INTERNATIONA	SINGLE AXLE STEAM TRUCK, EATON FULLER 10-SPEED TRANSMISSION, 11R22.5 TIRES, ALUM BUDS,	ZC
		l L	C/W INTERCONTINENTAL TRUCK BODY S/N ITB0211836, HOTSY STEAMER, 122,421-KILOMETERS	
		_	INDICATED, 20,828-HOURS INDICATED	
PF302	2003	GMC	PICKUP TRUCK, EXTENDED CAB, 4X4	ED
		-	Total mosty extended only the	
SSK003	2003	WESTERN STAR	TRI-DRIVE SOUR SEAL TANK TRUCK W/ 36' SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-	ZAMA CITY
	1		SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLES, REAR TIRES	
		ļ	11R24.5-ALUM BUDS, 42,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W JASPER TANK, 3"	
			FLUID PUMP	
Q002	2003	CUSTOM VAC		SC
		SERVICES	SUSPENSION, C/W CUSTOM VAC TANK S/N 10110703, 21,746-LITRE, TC412	
V059	2003		TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	MEDICINE
•000	2000	· ·	TIRES 65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, SLEEPER, C/W CUSTOM VAC	
			SERVICES TANK, 18,200-LITRE, S/N 349473-03	HAT
CD099	2003	DODGE	SERVICES TANK, 18,200-LITRE, 5/N 349473-03 SERVICE TRUCK [CONVERTED PICKUP TRUCK] FULL UTILITY BODY, QUAD CAB, 4WD [ENGINE	SWIFT
CD033	2003	DODGE	REPLACED, 361,324-KILOMETERS INDICATED	
TWPS010	2003	ALLWELD	TRIDEM TC406 ALUM TANK TRAILER, 74,000-LB AXLES, TIRES 11R24.5, ALUM BUDS, AIR RIDE	CURRENT
10010	2005	ALEVVELD		SWIFT
UT001	7003	SOUTHLAND	SUSPENSION, SIDE FLOATS, (LIC V547-85)	CURRENT
01001	2003	SOUTHLAND	T/A DUMP TRAILER, LT235/85R15 TIRES, STEEL BUDS, (LIC R300-39)	ED
HR120	2003	KENWORTH	TRI-DRIVE HYDRO-VACUUM TRUCK W/36" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-	ED
1111223	2005	I KENTONINI	SPEED, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM	LD
			BUDS, 58,000-LB REAR AXLES, NEW WAY AIR RIDE SUSPENSION, C/W WESTECH VAC SYSTEM HYDRO-	
			VAC SYSTEM, 11,166-LITRE CAPACITY, DOT 407, S/N 1202801, HIBON SIAV 822 BLOWER, S/N	
			046346732, HOTSY WATER BLAST WB 9460 HEATER, S/N 51880903, 16,465-HOURS INDICATED	
			[SPEEDOMETER REPLACED; KILOMETERS ESTIMATED TO BE 360,000]	
MV005	2003	MOUNTAIN	TRIDEM WELLSITE TRAILER, TIRES LT265/75R16	ED ED
		VIEW	THE CHARLETT THE CELEGY FOR TO	LD
ST205	2003	FORD	SERVICE BODY TRUCK, SUPER CREW, 4X4, POWER STROKE V8 DIESEL ENGINE, AUTOMATIC	
			TRANSMISSION, FRONT TIRES 225/70R19.5 ALUM BUDS, REAR TIRES 225/70R19.5 STEEL BUDS, C/W	
5002	2003	MESTERN STAR	WESTERN TRUCK BODY SERVICE BODY, GRILL GUARD T/A STEAMER UNIT, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, 16,000-LB FRONT AXLE,	NALL :
5002	2003	WESTERN STAR		MH '
			FRONT TIRES 425/65R22.5, ALUM BUDS, 46,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS,	
			AIR RIDE SUSPENSION, C/W THERMO KING 4-13248 VAN BODY S/N VB3186, HOTSY 9480 STEAMER,	
			WATER BLAST, HOSE REELS, 3 POLY TANKS, ESTIMATED 5,000-LITRE CAPACITY TANK	
WS110	2003	MACK	T/A TANK TRUCK, MACK 460, MAXITORQUE 18-SPEED, 11R24.5 FRONT, 12000LB FRONT, ALUM	
***************************************	2003	1	BUDS, 11R24.5 REAR, 46000-LB REAR, AIR RIDE, 236" WB, C/W TC406 CRUDE TANK W/SPRAY BAR,	
		1	PINTLE HITCH, PTO BOWEE PUMP, 16M3 CAP, TANK S/N 2AESTHB044S000184 [REPORTED AS	
V004	2004	MECTERNICTAR	PERMANENTLY OUT OF SERVICE	MEDIANIE
V084	2004	ANESTERIA STAK	TRI-AXLE VACUUM TRUCK, CAT ENGINE, EATON FULLER 3-SPEED TRANSMISSION, FRONT TIRES	MEDICINE
		İ	445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 54,000-LB AXLES, AIR	HAT
			RIDE SUSPENSION, (LIC F32691) C/W CUSTOM VAC, TC SPEC 412, 19,193-LITRE CAPACITY -	
			SN17841204, HIBON BLOWER, TIREBOSS TIRE PRESSURE CONTROL	

Eguip #	(49) 2004	Make DODGE	Description PICKUP TRUCK, QUAD CAB 4WD, 1-TON DUALLY, 5.9L CUMMINS TURBO DIESEL, AUTO, PW/PDL,	Rosestrone CNRL
CD201	2004	DODGE	235/80R17 TIRES, C/W 10' SERVICE DECK, CABINETS, DIESEL/METHANOL TANKS	CINIC
MV017	2004	MOUNTAIN	WELLSITE TRAILER, TRI/AXLE, TIRES 75R16	ED
		VIEW		
MV019	2004	MOUNTAIN	TRI-AXLE WELLSITE TRAILER, TIRES 265/75R16-STEEL DAYTONS, (LIC T17913)	ED
		VIEW		
V061	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			445/65R22.5-ALUM BUDS, 16,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 58,000-LB AXLES, AIR	
			RIDE SUSPENSION, (LIC G20935) C/W CUSTOM VAC, TC SPEC 412, 18,280-LITRE CAPACITY - S/N 349475-03. HIBON BLOWER	
V064	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, FRONT AXLE	ED
			65R22.5 ALUM BUDS 20,000-LB, REAR AXLES 11R24.5 ALUM BUDS 78,000-LB, AIR RIDE SUSPENSION,	
			C/W CUSTOM VAC SERVICES TANK, TC 412, 19,193-LITRE S/N 10781003, HIBON 820 BLOWER	
W071	2004	WESTERN STAR	STRAIGHT T/A WATER TRUCK, CAT C12 ENGINE, 18-SPEED TRANSMISSION, FRONT TIRES	ED
			425/65R22.5-ALUM BUDS, 16,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB AXLES,	
			CHALMERS SUSPENSION, S/N 5KKHALLASS4M16154, C/W CUSTOM VAC SERVICES 16,528-LITRE	
			TANK, TC SPEC 412, S/N 10240703, BOWIE PUMP, 221,085-KILOMETERS INDICATED, 23,587-HOURS	
			INDICATED	
W074	2004	WESTERN STAR	T/A WATER TRUCK, CAT C12 430-HP ENGINE, EATON FULLER 18-SPEED, 16,000-LB FRONT AXLE,	ED
			FRONT TIRES 425/65R22.5, ALUM BUDS, 46,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS,	
MV004	2004	MOUNTAIN	C/W CUSTOM VAC STEEL TANK. S/N 10260903, 16,528 L, TC412, BOWIE PUMP TRIDEM WELLSITE TRAILER, 10'X30', TIRES 265/75R16, C/W LPG PIG, FRIDGE, MICROWAVE, STOVE,	EĐ
1010004	2004	VIEW	WASHROOM, SHOWER, FOLD UP STEP, RAILING	LD
PS118	2004	KENWORTH	S/A PRESSURE TRUCK W/ 40" SLEEPER, CUMMINS ISC 315-HP ENGINE, EATON FULLER 10-SPEED	CNRL
			TRANSMISSION, FRONT TIRES 11R22.5 TIRES-ALUM BUDS, REAR TIRES 11R22.5-ALUM BUDS, AIR	
			RIDE SUSPENSION, (LIC G50667), C/W HAMM'S TANK S/N 46ANM062M31029424, TC406, 2	
			COMPARTMENT - 4,045-LITRE/4,055-LITRE CAPACITY, GARDNER DENVER TRIPLEX PUMP	
PTI108	2004	INTERNATIONA	S/A PRESSURE TEST TRUCK, DT 466 ENGINE, EATON FULLER 10-SPEED ENGINE, 11R22.5 TIRES, ALUM	GP
		L	BUDS, (LIC G03323), C/W TRUCK BODY S/N ITB312253, OPERATOR CONTROL CABIN, TRIPLEX PUMP	
			AND TANKS, 10,000-PSI, 40-L/M, -KILOMETERS REPORTED	
SSP404	2004	THOMAS	SKID STEER LOADER, TIRES 10-16.5, C/W FORKS, 3206-HOURS INDICATED	ED
TRO01	2004	WESTERN STAR	T/A HIGHWAY TRACTOR W/ 34" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	GP
			TRANSMISSION, FRONT TIRES 425/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, (LIC G10843), C/W	
			WET KIT, SLIDING 5TH WHEEL, 424,021-KILOMETERS INDICATED, 13,093-HOURS INDICATED	
UT003	2004	SOUTHLAND	DUMP TRAILER, LIC. #T17915	GP
W075	2004	WESTERN STAR	T/A TANK TRUCK, CAT C12 430-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	CNRL
			425/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB	HORIZON
			REAR AXLES, C/W CUSTOM VAC SERVICES S/N 10270903, TC412, 16,528-LITRE CAPACITY, BOWIE 3"	PROJECT
			PRODUCTION PUMP	
W309	2004	WESTERN STAR	T/A WATER TRUCK, EATON FULLER, 425/65R22.5 FRONT, 11R24.5 REAR, ALUM BUDS, 23000-LB X 2	ED
VI0.53	2004	MECTERALICTIC	AXLES, (LIC G-19583), C/W CUSTOM TANK 16418L CAP, S/N 15331004	
V062	2004	VVESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 58,000-LB AXLES, AIR RIDE SUSPENSION, (LIC G19595) C/W CUSTOM VAC SERVICES, TC SPEC 412, 18,355-LITRE CAPACITY -	
			INDE SUSPENSION, (IIC G19595) C/W CUSTOM VAC SERVICES, TC SPEC 412, 18,355-LITRE CAPACITY - IS/N 10390803. H/BON BLOWER	
V067	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB AXLES, AIR	
			RIDE SUSPENSION, (LIC G19593) C/W CUSTOM VAC, TC SPEC 412, 19,193-LITRE CAPACITY - S/N	
			11511003, HIBON BLOWER.	

Foure, 8	Year	Make	Description	Location
V069	2004		TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, 475-HP ENGINE, EATON FULLER 18-SPEED	ED
1003	2004	VV ESTERNI STYLL	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 60,000-LB REAR AXLES, (LIC F32688), C/W CUSTOM VAC SERVICES VAC UNIT	
		1		
			S/N 11551103, 19,193-LITRE CAPACITY, TC412 SPEC, HIBON BLOWER, HYDRAULIC HOIST, REAR	
V085	2004	VALECTE DAL CTAD	DOOR VIRRATOR DUMP GATE	- FD
VU85	2004	MESTERN STAR	TRI-AXLE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL BUDS, 56,000-LB AXLES, AIR	
			RIDE SUSPENSION, (LIC F32692), C/W CUSTOM VAC TANK, 19,193-LITRE CAPACITY - S/N 13510204	
V089	2004	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT SERIES 60 ENGINE, 515-HP, EATON FULLER 18-SPEED	ED
		1	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W	
		1 :	CUSTOM VAC SERVICES TANK, 19,193-LITRE CAPACITY, S/N 13540304, 14.9-PSI RATING, HYDRAULIC	
			BOOM, 165,450-KILOMETERS INDICATED, 17,507-HOURS INDICATED	
MV003	2004 -	MOUNTAIN	TRI-AXLE WELLSITE TRAILER, TIRES 265/75R16-STEEL DAYTONS, PROPANE TANK, (LIC SS61-54)	ED
1111000	2004	VIEW	THE AREE WELLSHE HARLER, THES 2007 THE OFFICE DATFORD, THOUARE TANK, (EC 3302-34)	LU
MV042	2004		TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 2-BEDROOMS], TIRES 265/75R16-STEEL	ED
1010042	2004	VIEW	DAYTONS, PROPANE TANK, (LIC T260-43)	
W308	2004		T/A WATER TRUCK, DETROIT 60 SERIES ENGINE, EATON FULLER 18-SPEED, 18000-LB FRONT AXLE,	ED
W306	2004	MESTERN STAN		ED
			445/65R22.5 FRONT TIRES, ALUM BUDS, 46000-LB REAR AXLES, 11R22.5 REAR TIRES, STEEL BUDS,	
		1	AIR RIDE SUSPENSION, ESTIMATED 16,000-LITRE CAPACITY (LIC. G10829), C/W STEEL TANK,	
TD000	2004	LUCETERU CTAR	PRODUCT PUMP	
TR002	2004	MESTERN STAR	T/A TRACTOR W/ SLEEPER, DETROIT SERIES 60 500-HP ENGINE, EATON FULLER 18-SPEED	ZAMA CITY
			TRANSMISSION, FRONT TIRES 425/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	
		İ	11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W WET KIT, SLIDING 5TH	
		ļ	WHEEL, ROPER 4" PRODUCTION PUMP	
MV018	2004		TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 1-BEDROOM], 265/75R16 TIRES PROPANE PIG,	MEDICINE
			KITCHEN, BATH, BEDROOM, (LIC S891-96)	HAT
RV047	2004	HAULMARK	TRI-AXLE GOOSE NECK ENCLOSED 38' SITE TRAILER, LT 235/85R16 TIRES, ALUM BUDS, W/ REAR	MEDICINE
			DOOR, FRONT LIVING QUARTERS, SIDE DOOR, GENSET, AWNINGS, A/C, (LIC 1070-85)	HAT
V066	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB AXLES, AIR	-
			RIDE SUSPENSION, (LIC F32656), C/W CUSTOM VAC S/N 11491003, TCRN Z.05.437.03.04, TC 412,	
·		1	19,193-LITRE CAPACITY, HIBON BLOWER, TIRE BOSS PRESSURE GAUGE	
11000	2004	W.ECZEDN CEAD	TO DON'T VICE WATCHER OF THE PROPERTY OF THE P	
V083	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	MEDICINE
			TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W CUSTOM VAC SERVICES	HAT
1			TANK, 19,193-LITRE, S/N 2530104 [IN ACCIDENT ON 8/27/10; WHOLE FRONT END TO BE REPLACED	
			ALONG W/ PTO, FUEL TANKS, HOSES. EXTENSIVE BODY WORK TO BE PERFORMED; CONSIDERED	
			REPAIRED]	
AT001	2004	HEIL	TRIDEM ALUM WATER TANKER, 60,000-LB AXLES, TIRES 11R24.5, STEEL BUDS, (LIC T055-68), C/W	SWIFT
			ALUM TANK, 37,854 L, TC407	CURRENT
P136	2004	CHEVROLET	PICKUP TRUCK, CREW CAB, 3/4-TON, DURAMAX DIESEL, 4WD, LEATHER SEATS (NOT IN SERVICE AT	SWIFT
			TIME OF INSPECTION]	CURRENT
TR003	2004	WESTERN STAR	PRODUCTION WATER TRUCK, BULKER, 36" SLEEPER CAB, 279,837-KILOMETERS INDICATED	MH
V065	. 2004	MIECTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, 475-HP, EATON FULLER 18-SPEED TRANSMISSION,	ED
VUOS	- 2004			ED
1			FRONT TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W CUSTOM VAC	
			SERVICES VAC UNIT, 19,193-LITRE CAPACITY, TC412 SPEC, S/N 11481003, C/W HIBON BLOWER,	
			HYDRAULIC HOIST, REAR DOOR, VIBRATOR, DUMP GATE & KNIFE VALVE	
V068	2004		TRI-AXLE VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, 20,000-LB FRONT	ED
			AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 60,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM	
			BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC STEEL TANK, 19,193-LITRE, TC412, S/N 11531003,	
			HIBON BLOWER, AIR BOSS TIRE INFLATION SYSTEM	
V086	2004	WESTERN STAR	T/A VACUUM TRUCK, CAT C12 ENGINE, EATON FULLER 18-SPEED, FRONT AXLE 425/65R22.5 ALUM	ED
			BUDS 16,000-LB, REAR AXLES 11R24.5 ALUM BUDS 46,000-LB, C/W CUSTOM VAC TANK, TC350, S/N	
				1
			13200204, 19,193-LITRE CAPACITY, HIBON BLOWER, 181,321-KILOMETERS INDICATED, 14,505-	



Big Eagle Limited Partnership Asset List* GA Global Partners Report

Equip. #	Year	Make	Qescription .	Attraction.
S003	2004		STEAM TRUCK, 24" SLEEPER CAB, EATON FULLER 13-SPEED TRANSMISSION [NEW STEAMER ADDED	MH
5003	2001	11.63121111317111	IN 2015]	
V063	2004	WESTERN STAR	TRI-AXLE VACUUM TRUCK, CAT ENGINE, EATON FULLER 3-SPEED TRANSMISSION, FRONT TIRES	ED
1003	2001	11123121111311111	445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 54,000-LB AXLES, AIR	
			RIDE SUSPENSION, C/W CUSTOM VAC, TC SPEC 412, 19,193-LITRE CAPACITY, S/N 10771003, HIBON	
•				
V088	2004	WESTSON STAD	BLOWER. 146.650-KILOMETERS INDICATED. 18.573-HOURS INDICATED TRI-DRIVE VACUUM TRUCK, CAT 3637 ENGINE, EATON FULLER 13-SPEED TRANSMISSION, FRONT	MH
VU88	2004			IVIE
			TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 57,000-LB	
			AXLES, AIR RIDE SUSPENSION, (LIC F32681) C/W CUSTOM VAC, TC SPEC 412, 18,262-LITRE CAPACITY	
			S/N 13210204. HIBON BLOWER	
MV007	2004		WELLSITE TRAILER, TRI/AXLE, TIRES 75R16	G₽
		VIEW		
UTC314	2004		V-NOSE 14' TANDEM-AXLE OFFICE TRAILER, 225/75R15, ALUM MAGS, SIDE DOOR, REAR BARN	
		JUNCTION	DOORS, HEATER, CARRIER AIRV AC, C/W SUNMAR EXCEL TOILET, SYLVANIA MW, BUILT-IN	
			DESK/COUNTER, SOFA, OFFICE CHAIR, (LIC W060-64)	
UTC316	2004	SL TRAILERS	TANDEM-AXLE 16' FLATDECK TRAILER, TIRES 235/80R16-STEEL BUDS, (LIC W060-65)	
MV041	2004	MOUNTAIN	TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 1-BEDROOM], TIRES 75R16	ED
		VłEW		
AT002	2004	HEIL	TRIDEM WATER TANKER, TIRES 11R24.5, STEEL BUDS, AIR RIDE SUSPENSION, S/N	ĘD
			5HTDL423245J74948, C/W 1 COMPARTMENT ALUM TANK, 37,854 L, TC407	
MV012	2004	MOUNTAIN	WELLSITE TRAILER, TRI/AXLE	MEDICINI
		VIEW		HAT
MV014	2004	MOUNTAIN	TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 1-BEDROOM], TIRES 75R16	MEDICINI
101002-7	2001	VIEW	THIS ENTITY THE BEET LETTING QUARTERS WY 2 SEPTIONING, THEO TOTALS	HAT
UTK010	2004	JASPER	UTILITY TRAILER	ПАІ
OTROLO	2004	JASER	OTILITY TRAJLER	
CCCCC	2005	N/A	SIDE SPOOLER CUSTOM BUILT TANDEM-AXLE TRAILER MOUNTED, 215/75R17.5 TIRES, 20000-LB	ED
CSC001	2005	I N/A		ED
boose	2005	FORD	AXLES, C/W 1978 INTERNATIONAL DT466 200-HP DIESEL ENGINE, 2018-HOURS INDICATED	
PC014	2005	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE, 3/4-TON, 288,892-KILOMETERS INDICATED	ED
PF303	2005	FORD	PICKUP TRUCK, LARIAT KING RANCH QUAD CAB, 4X4, TURBO DIESEL ENGINE, TIRES 285/65R18,	ED
FFQUQ	2003	FORD	FULLY LOADED, SHORT BED, (LIC G-19621), C/W ALUM TOOL BOX	LD
CTOO	2005	INTERNATIONA	SINGLE AXLE SERVICE TRUCK, SLEEPER, EATON FULLER 10-SPEED TRANSMISSION, 10,433-LB. FRONT	CNRL
ST202	2005	1		CINIC
		1	AXLE, 23,000-LB. REAR AXLES, SERVICE BODY W/ STORAGE BOXES, WELDER/GENERATOR, AIR	
			COMPRESSOR, AUTO CRANE 22' BOOM CRANE W/ 3,200-LB CAPACITY, 181,613-KILOMETERS	
		<u> </u>	INDICATED, 8.910-HOURS INDICATED	
V502	2005	WESTERN STAR	VACUUM TRUCK, TRI-AXLE, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ED
			CUSTOM VAC 19,193-LITRE TANK S/N 15230704, HYDRAULIC BOOM, HIBON BLOWER, -KILOMETERS	
			INDICATED, 22,310-HOURS INDICATED	
V503	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT SERIES 60 575-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
		-	11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES	
			S/N 15240704, TC407/412, 19,193-LITRE CAPACITY, HIBON BLOWER, 138,076-KILOMTERES	
			REPORTED, 15,493-HOLIRS REPORTED	
V504	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT DIESEL SERIES 60 ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL	
			BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION, (LIC F24231), C/W CUSTOM VAC-S/N 15250704.	
			19.193-HOURS INDICATED	
V505	2005	WESTERN STAR	VACUUM TRUCK, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	ED
A 203	2003	TATESTERNASTAR	TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB	
	1			
	1		AXLES, S/N 5KKPALCK25PW69021, C/W CUSTOM VAC SERVICES TANK, 19,193-LITRE, S/N 15260704	
MEGG	200-	11/557531.57:-	AND CHAINATE HOLD DETERMINED TO A CONTRACT OF THE CONTRACT OF	
V506	2005	WESTERN STAR	VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 15,777-KILOMETERS	ED
			INDICATED, 11,576-HOURS INDICATED, FRONT TIRES 445/65R22.5 ALUM BUDS, REAR TIRES 11R24.5	
			ALUM BUDS-10, W/ SLEEPER, TIRE PRESSURE MONITOR, ALUM CABINETS, C/W CUSTOM VAC UNIT	
			TC407/412, TCRN Z-05-437-03-04, 19.193 L CAPACITY S/N 15270804	

Equip.#	Year	Wake	Description	La colli
V508	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
			SERVICES S/N 15751204, TC407/412, 19,193-LITRE CAPACITY, HIBON BLOWER, 24,375-KILOMETERS	
			INDICATED, 4.018-HOURS INDICATED	
V511	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT DIESEL SERIES 60 ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT 20,000-LB AXLE, REAR 61,000-LB AXLES, (LIC F49363) S/N	
			5KKPALCK65PV23785, C/W CUSTOM VAC TCRN Z.05.437.03.04 TC SPEC 407/412, 19,193-LITRE	
		1	CAPACITY, 144,120-KILOMETERS INDICATED, 18,922-HOURS INDICATED	
V517	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT DIESEL, EATON FULLER 18-SPEED, 445/65R22.5 FRONT, 11R24.5	ED
			REAR, ALUM BUDS, AIR RIDE, 20400-LB X 3 AXLES, TIRE AIR PRESSURE CONTROL, (LIC F-49375), C/W	
			CUSTOM VAC UNIT, HIBON VTB820, TC407/412, 19193 L CAP, S/N 15731104	
V518	2005	WESTERN STAR	TRI-DRIVE VAC TRUCK, DETROIT SERIES 60 DIESEL ENGINE, 515-HP, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W	
			CUSTOM VAC SERVICES VAC UNIT, 18,262-LITRE CAPACITY, S/N 17831204, C/W HIBON BLOWER,	
			HYDRAULIC HOIST, REAR DOOR, VIBRATOR, DUMP GATE, 167,730-KILOMETERS INDICATED, 12,140-	
			HOURS INDICATED	
W302	2005	WESTERN STAR	TRI-AXLE TANK TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	GP
•			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
		1	11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
		1	SERVICES S/N 10290404, TC407/412, 20,123-LITRE CAPACITY, BOWIE 4" PRODUCTION PUMP	
			SENTICES SAN 20230 10 17 124 203225 ETHE GIVEN THE STAN TO THE SAN THE SAN TO THE SAN TO THE SAN TO THE SAN TO THE SAN TO THE SAN THE SAN TO THE SAN TO THE SAN TO THE SAN TO THE SAN THE SA	
·W304	2005	WESTERN STAR	PRODUCTION WATER TRUCK, BULKER	ED
W305	2005	WESTERN STAR	TRI-DRIVE WATER TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	EĐ
	2.003		FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,200-	
			LB AXLES, AIR RIDE SUSPENSION, (LIC F24219), C/W CUSTOM VAC S/N 10320504, TC 407/412, 20,123-	
			LITRE CAPACITY	
W307	2005	WESTERN STAR	TRI-AXLE WATER TRUCK, DETROIT 60 SERIES ENGINE, EATON FULLER 18-SPEED, 20000-LB FRONT /	ED
11307	2000		20400-LB X 3 REAR AXLES, FRONT 445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE	
			SUSPENSION, C/W CUSTOM VAC SERVICES STEEL TANK, TC 406, 21123 L CAPACITY S/N 10340604	
			303FENSION, G VV CO3: OV; VAC SERVICES STEEL (ANK, TC 400, Z1125 L CAPACITI 3/N 10340004	
W310	2005	KENWORTH	WATER TRUCK, TANDEM-AXLE, EATON FULLER 18-SPEED TRANSMISSION, CUSTOM VAC TANK S/N	ED
			15341004, 16,418-LITRE CAPACITY, 16,967-KILOMETERS INDICATED, 1,900-HOURS INDICATED	
			·	
W312 .	2005	WESTERN STAR	TANDEM-AXLE WATER TRUCK, CAT, EATON FULLER 18-SPEED, 385/65R22.5 FRONT, 11R24.5 REAR,	ED
			ALUM BUDS, AIR RIDE, 23000-LB X 2 AXLES, (LIC F-65461), C/W ADVANCE WATER TANK, 16000-LITRE	
			CAPACITY, 134,597-KILOMETERS INDICATED, 18,931-HOURS INDICATED	
W313	2005	WESTERN STAR	TRI-DRIVE WATER TRUCK, DETROIT DIESEL, 425/65R22.5 FRONT, 11R24.5 REAR, ALUM BUDS, AIR	ED
:			RIDE, 19400-LB X 3 AXLES (LIC F-65474), C/W CUSTOM WATER TANK 20123L CAP, TC407, S/N	
			19990505	
W314	2005	STERLING	T/A WATER TRUCK, DETROIT 60 SERIES DIESEL ENGINE, EATON FULLER 18-SPEED, 13000-LB FRONT /	ED
			20000-LB X 2 REAR AXLES, FRONT 11R24.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, ESTIMATED	
			15,000-Litre, FLOAT TIRES	
AT005	2005	HEIL	TRIDEM WATER TANKER, TIRES 11R24.5-STEEL BUDS, AIR RIDE SUSPENSION, C/W FIBERGLASS	GP
			COATED TANK	
HOK047	2005	WESTERN STAR	HOT OIL TRUCK, 186,994-KILOMETERS INDICATED, 5,249-HOURS INDICATED	GP
HU001	2005	I	T/A FLUID HEATER UNIT W/ SLEEPER, CUMMINS ISM ENGINE, EATON FULLER 18-SPEED	GP
		L	TRANSMISSION, FRONT TIRES 12R24.5-ALUM BUDS, 14,000-LB AXLE, REAR TIRES 11R24.5-STEEL	
			BUDS, 40,000-LB AXLES, (LIC H-26755), C/W CUSTOM WELDING TANK- S/N TN00038-2144 MDIN	
			D000190, 8,840-LITRE CAPACITY TANK, 8,000,000- BTU HEATER, 3 COMPARTMENT TANK 1,400-	
		ho	LITRE, 1,400-LITRE, AND 6,040-LITRE CAPACITY	
PR206	2005	DODGE	PICKUP TRUCK, QUAD CAB, 4WD, TURBO DIESEL ENGINE,179,590-KILOMETERS INDICATED	ED
	2005	LAUFETE DAL CTAD	VACUUM TRUCK, TRI-DRIVE	GP
V521				

Year Description Location WESTERN STAR TRI-DRIVE COMBO TRUCK, DETROIT 60 SERIES 14L 515-HP ENGINE, FATON FULLER 18-SPEED, FRONT ED CB904 2005 TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC UNIT, S/N 16210904, 12,682-LITRE WESTERN STAR TRI-DRIVE WATER TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-CNRL 2005 W306 ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,000-LB AXLES, AIR RIDE HORIZON SUSPENSION, (LIC F24220), C/W CUSTOM VAC SERVICES LTD TANK, TO SPEC 406, 21,123-LITRE PROJECT CAPACITY - S/N 10330604 WESTERN STAR TRI-DRIVE VACUUM TRUCK, EATON FULLER 18-SPEED TRANSMISSION, AIR RIDE SUSPENSION, C/W ED V513 2005 CUSTOM VAC TANK WESTERN STAR TRI-AXLE VACUUM TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-FD V522 2005 ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION, (LIC F24220) C/W CUSTOM VAC, TC SPEC 407/412, 19,193-LITRE CAPACITY -SN17841204, HIBON BLOWER TRI-AXLE VACUUM TRUCK, DDC 60 SERIES ENGINE, EATON FULLER 18-SPEED, 20,000-LB FRONT ED V523 2005 WESTERN STAR AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC S/N 17871204, 19,193-LITRE CAPACITY, TC 407/412, ROBUSCHI BLOWER ROUGHNECK T/A 28' RV TRAILER, TIRES 235/80R16, COMPLETE ΕĐ RN063 2005 2005 FORD PICKUP TRUCK, CREW CAB, 3/4-TON, SERVICE TRUCK ED PR203 INTERNATIONA S/A PRESSURE TEST TRUCK W/ SLEEPER, DT466 ENGINE, EATON FULLER 6-SPEED TRANSMISSION, PTF104 2005 FD FRONT TIRES 1R22.5 TIRES-ALUM BUDS, REAR TIRES 11R22.5-ALUM BUDS, AIR RIDE SUSPENSION, C/W INTERCONTINENTAL 161 TRUCK BODY S/N 6575, TRIPLEX PUMP, OPERATOR CONTROL ROOM INOT IN SERVICE ON DATE OF INSPECTION! мн 2005 ROUGHNECK RN053 TRAILER, CREW SHACK RN056 2005 ROUGHNECK T/A RV TRAILER, (LIC T537-40), C/W PROPANE PIG ED ROUGHNECK TANDEM-AXLE RV TRAILER, W/ PINTLE HITCH EXTENSION, PIG, 100-LB ED 2005 RN065 TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 2-BEDROOMS AND CENTRAL AIR], TIRES ĔΒ MV045 2005 MOUNTAIN VIEW 265/75R16, STEEL DAYTONS, PROPANE TANK ROUGHNECK T/A RV TRAILER, 235/80R16 TIRES, KITCHEN, BED, 100-LB PROPANE PIG FD RN062 2005 WESTERN STAR TRI-AXLE NITROGEN PUMP UNIT TRUCK, SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-ED NP001 2005 SPEED, FRONT TIRES 425/65R22.5, ALUM BUDS, REAR TIRES11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, C/W HYDRA RIG NITROGEN PUMP UNIT, TC 338, TCRN Z-05-507-06-05 S/N HRC180K10KHR110003-1, WESSINGTON TANK TC338 S/N 460, CAIN HEAT RECOVERY SYSTEM, TRIPLEX PUMP, CONTROL CAB, 364,745-KILOMETERS INDICATED, 14,874-HOURS INDICATED WESTERN STAR TRI-AXLE NITROGEN PUMP UNIT, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED FD NP002 2005 TRANSMISSION, 20000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61200-LB REAR AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, SLEEPER, (LIC G19596), C/W HYDRA RIG NITROGEN PUMP UNIT, CAIN HEAT RECOVERY SYSTEM S/N 4183, WESSINGTON TANK ID# 464, TC338 TANK, CONTROL CAB, BACKEND S/N HRC180K10KHR110004-2 WESTERN STAR TRI-AXLE NITROGEN PUMP UNIT TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, FRONT ΕD NP004 2005 TIRES 445/65R22.5 ALUM BUDS, REAR TIRES 11R24.5 ALUM BUDS, AIR RIDE SUSPENSION, C/W HYDRA RIG NZ PUMPER UNIT TC 338, TCRN Z-05-507-06-05 S/N HRC180K10KHR110004-3, WESSINGTON CRYOGENICS TANK S/N 448, SLEEPER WESTERN STAR TRI-AXLE NITROGEN TRANSPORTER TRUCK W/ SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-FD 2005 NT001 SPEED, 20,000-LB FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC F65500), C/W HYDRA RIG NITROGEN

352 760-KII OMETERS INDICATED: 8.180-HOURS INDICATED

TANK SETUP, WESSINGTON CRYOGENIC TANK, 15,897 L, TC338 TANK, MODEL RTV 4200 S/N 465,

			Make	Description	Location
(CB901	2005		TRI-DRIVE COMBO UNIT, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, 19705-LB FRONT AXLE,	RL
				445/65R22.5 FRONT TIRES, ALUM BUDS, 58000-LB REAR AXLES, 11R24.5 REAR TIRES, ALUM BUDS,	
				NEWAY AIR RIDE SUSPENSION, SLEEPER, (LIC F96606), C/W CUSTOM VAC SERVICES COMBO UNIT	
				S/N 16421004, 13,147-LITRE CAPACITY, TC407/412 DUAL COMPARTMENT TANK, ROBUSCH!	
				BLOWER, HOTSY STEAMER, WATER BLAST, HYDRAULIC BOOM, DUMP GATE VALVES	
	PS120	2005	FORD -	PRESSURE TRUCK, TESTER	RL
	7TI102	2005	FORD	SUPER DUTY 4X4 PRESSURE TEST TRUCK, POWER STROKE DIESEL ENGINE, 6-SPEED TRANSMISSION,	RL
r	11102	2003	l I	FRONT TIRES 225/70R19.5-ALUM BUDS, REAR TIRES 225/70R19.5-STEEL BUDS, SPRING SUSPENSION,	N.L
			1	(LIC G03331), C/W INTERCONTINENTAL 16' TRUCK BODY S/N ITB0111279,2 STAINLESS STEEL TANKS,	
			l I	3 ALUM TANKS, NATIONAL J30 3000-PSI TRIPLEX PUMP S/N 12616-D, OPERATOR CONTROL ROOM	
				5 ALON TAINS, NATIONAL 130 3000-1 31 TRIL LEAT OWN 3/N 12010-0, OF ENATOR CONTROL MODIN	
(CB905	2005	WESTERN STAR	TRIDRIVE COMBO VACUUM TRUCK, DDC 60 490-HP ENGINE, AUTOMATIC TRANSMISSION, FRONT	GP
				TIRES 445/65R22.5-ALUM BUDS, 18,000-LB FRONT AXLE, REAR TIRES 11R24.5-AUMINUM BUDS,	
	٠.			57,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES TANK S/N 16230105,	
				TC406/412, 2 COMPARTMENT - 10,146-LITRE CAPACITY, HIBON BLOWER, WATERBLAST WASHER	
-	CB906	2005	WESTERN STAR	TRIDRIVE COMBO VACUUM TRUCK, DDC 60 515-HP ENGINE, AUTOMATIC TRANSMISSION, FRONT	GP
		2000	77.23(2,111.3), 11	TIRES 445/65R22.5-ALUM BUDS, 18,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-	٥.
				LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES STEEL TANK S/N 16240205,	
				TC407/412, 2 COMPARTMENT - 12,682-LITRE CAPACITY, HIBON BLOWER, WATERBLAST HEATER	
Ŋ	ЛV044	2005	1	TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 1-BEDROOM],TIRES 265/75R16-STEEL DAYTONS,	MEDICINE
	NOCZ	2005	VIEW	PROPANE TANK, (LIC T260-45) T/A RV TRAILER, TIRES 235/80R16-STEEL BUDS, PROPANE PIG, (LIC T537-37)	HAT
ſ	RN057	2005	ROUGHNECK		MEDICINE HAT
F	RN058	2005	ROUGHNECK	T/A RV TRAILER, CREW SHACK	MEDICINE
					HAT
	ST203	2005	STERLING	S/A SERVICE BODY TRUCK, ACTERRA MERCEDES BENZ ENGINE, STANDARD TRANSMISSION, FRONT	MEDICINE
				TIRES 11R22,5-STEEL BUDS, 12,000-LB AXLE, REAR TIRES 11R22.5-STEEL BUDS, 21,000-LB AXLE, (LIC	HAT
				G19600), C/W BRUTUS SERVICE DECK, STORAGE COMPARTMENT, TIGER MODEL 6541E 4-TON	
	TR004	2005	MECTERN CTAR	CRANE. AIR COMPRESSOR T/A TRACTOR, 24" SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	MEDICINE
	111004	2003	M COSTINA DI MI	TIRES 65R22.5-ALUM BUDS, 16,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB AXLES,	HAT
				372,058-KILOMETERS INDICATED	11/01
	TR006	2005	WESTERN STAR	T/A TRACTOR, SLEEPER, DETROIT SERIES 60 ENGINE, 515-HP, EATON FULLER 18-SPEED	MEDICINE
				TRANSMISSION, 367,108-KILOMETERS INDICATED	HAT
Т	WP006	2005	ADVANCE	TRIDEM VAC STIFF POLE PUP TRAILER, 24,000-LITRE, TIRES 11R24.5-ALUM BUDS, AIR RIDE	MEDICINE
				SUSPENSION	HAT
	V514	2005	1	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED, FRONT TIRES 445/65R22.5,	MEDICINE
				ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W 20005 CUSTOM VAC TANK, S/N 17821204,	HAT
				17,333 L, TC407/412	
	V524	2005	1	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 DIESEL ENGINE, 515-HP, EATON FULLER 18-SPEED	MEDICINE
			ſ	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W	HAT
				CUSTOM VAC SERVICES VACUUM TRUCK, 18,262-LITRE CAPACITY, S/N 17890105, C/W HIBON	
1	HR123	2005	KENWORTH	BLOWER, REAR DOOR, VIBRATOR, DUMP GATE TRI-DRIVE HYDRO-VACUUM TRUCK, INTEGRAL SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED	SWIFT
ľ			E .	TRANSMISSION, 20,000-LB FRONT AXLE, 425/65R22.5 FRONT TIRES, ALUM BUDS, 58,000-LB REAR	CURRENT
			1	AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G03328), C/W WESTECH VAC	
			E .	SYSTEMS S/N WT0604934, 13,756-LITRE CAPACITY, HIBON BLOWER, HOTSY, DUMP GATE VALVES,	
				30,009-KILOMETERS INDICATED, 2,191-HOURS INDICATED	
	Q003	2005	CHETOMANA	QUAD WAGON VAC TANK TRAILER, TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK S/N	CIANCE
	1 1/1 18 154	70KP	TELLISTUM VAC	TOUAD WAGON VAC TANK TKAILEK. TIKES 3.1KZ4.S. ALDIVI BUDS, C/W CUSTOW VAC TANK S/N	SWIFT
	2003	2005		1	CHRDENT
	VT001	2005	SERVICES	18850405, 21,820-LITRE QUAD WAGON VAC TANK TRAILER, TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK S/N	CURRENT SWIFT

FOURD 7	Year	Wake	Description	- Acres (Vien
TWP015	2005		TRAILER, VACUUM TRAILER, TRI-AXLE, W/ PROGRESSIVE TANK 4,230-GALLON VACUUM TANK S/N V-	SWIFT
IAALOTO	2003	1975	21911C (2006), DOT 407/412, CUSTOMVAC BLOWER	CURRENT
TWPS021	2005	ADVANCE	ADVANCE TRIDEM OIL PUP, TIRES 285/75R24.5, ALUM BUDS, 15,900-LITRE CAPACITY	SWIFT
IWFSUZI	2003	ADVANCE	ADVANCE INIDENTOR, TIRES 2007/31(24.0), ALONI BODO, 10,000-11(1), CAPACITY	CURRENT
V515	2005	MECTEDNI STAD	VACUUM TRUCK, TRI-DRIVE, CUSTOM VAC UNIT, 18,000L [NOT IN SERVICE AT TIME OF INSPECTION]	MH
ASTS	2003	AAESIEVIA SIWK	VACOUM TROCK, TRI-DRIVE, COSTOM VACOURI, 18,000E [NOT IN SERVICE AT TIME OF INSECTION]	19111
VELC	2005	VA/ECTCDAL CTAD	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 DIESEL ENGINE, 515-HP, EATON FULLER 18-SPEED	SWIFT
V516	2005		· · · · · · · · · · · · · · · · · · ·	
VEAO	2005		TRANSMISSION, C/W CUSTOM VAC SERVICES VACUUM TRUCK TRI-DRIVE VACUUM TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-	CURRENT SWIFT
V519	2005			
			ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL BUDS, 61,000-LB AXLES, AIR RIDE	CURRENT
			SUSPENSION, (LIC F49396), C/W CUSTOM VAC TANK, TCRN Z.05.437.03.04, TC SPEC 407/412, 18,262-	
1000	2005	KENWORTH	LITRE CAPACITY T/A VACUUM TRUCK, CAT C15 ACERT 475-HP ENGINE, EATON FULLER 18-SPEED, 18740-LB FRONT /	MH
VR103	2005			IAIL
			23000-LB X 2 REAR AXLES, FRONT 385/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, NEWAY AIR	
ļ	1		RIDE SUSP, C/W PROXY WELD INDUSTRIES 13033-LITRE TANK, TC SPEC 407, S/N 04-0757, HIBON VTB-	
14.770.044.0	8005		820 BLOWER, INOT IN SERVICE AT TIME OF INSPECTION!	CHRET
WTRS119	2005	MACK	T/A TANK TRUCK, MACK ENGINE, 18-SPEED, FRONT TIRES 11R24.5, ALUM BUDS, REAR TIRES	SWIFT
1	İ		11R24.5, ALUM BUDS, C/W ADVANCE TANK S/N 2AESTHBOX4S000206, 15,900-LITRE, 349,305-	CURRENT
	.		KILOMETERS INDICATED, 15,379-HOURS INDICATED	
PC010	2005	FORD	SPORT UTILITY VEHICLE, 4WD, TURBO DIESEL ENGINE V10	ED
DAAC	2005	OUT POUT	COSTU CAR RICK LIP DUDAMAN ENGINE AUTO TRECTAE (TERAC	SC
P146	2005	CHEVROLET	CREW CAB PICK UP, DURAMAX ENGINE, AUTO, TIRES 245/75R16	30
CT205	2005	KENWORTH	TANDEM-AXLE STEER/TRI-AXLE DRIVE COILED TUBING UNIT, CAT C15 550-HP ENGINE, EATON	ED
0.200		,,	FULLER 18-SPEED, 32,000-LB FRONT AXLES, FRONT TIRES 425/65R22.5, ALUM BUDS, 69,000-LB REAR	
			AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, SPICER 792, C/W COILED TUBING	
			UNIT, 3100M OF 1 1/4" TUBING, HYDRA RIG RT25 INJECTOR - 3/4" TO 2", CAT 4517 TRIPLEX PUMP,	1
	1		50L/MIN @ 21 MPA, W/NORTHWEST MODEL 5450 COMPRESSOR, 18M3 @ 14 MPA, W/TEREX	
			ATLAS 145CS-A4 14-TON PICKER, 6.7M REACH, S/N 100104 (LIC G-20899)	
AT004	2005	HEIL	TRIDEM WATER TANKER, 10,000-GALLON TANK, TIRES 11R24.5-STEEL BUDS	MEDICINE
				HAT
AT006	2005	HEIL	TRIDEM ALUM WATER TRAILER, 37,854-LITRE CAPACITY, TIRES 11R24.5-STEEL BUDS, (LIC T894-75)	MEDICINE
				HAT
V507	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED, FRONT	ED
	1		AXLE 445/65R22.5 ALUM BUDS, 20,000-LB, REAR AXLES 11R24.5 ALUM BUDS, 69,000-LB, AIRLINER	
			AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES 19,193 L CAPACITY, TC SPEC 407/412, S/N	
			15741204, HIBON VTB-820 BLOWER, S/N N/A, TIRE PRESSURE CONTROL SYSTEM, HYDRAULIC	
			BOOM: 144.056-KILOMETERS INDICATED: 20.542-HOURS INDICATED	
CB903	2005	WESTERN STAR	TRI-AXLE COMBO TRUCK, DAY CAB, DETROIT 60 SERIES, 14L 515-HP ENGINE, AUTOMATIC	MH
		•	TRANSMISSION, 18,000-LB. FRONT AXLE, 18,850-LB. REAR AXLES, CUSTOM VAC TANK S/N 16220105	
			W/ 13,147-LITRE CAPACITY	ļ
FKL002	2005	DAEWOO	LIFT TRUCK, LPG, 4450-LB CAPACITY, 3-STAGE, 186" MAST, SIDE-SHIFT, SOLID TIRES, 956-HOURS	MEDICINE
			INDICATED	HAT
P151	2005	CHEVROLET	CREW CAB 4X4 S/B PICKUP, DURAMAX DIESEL, AUTO, TIRES 265/75R16, FULLY LOADED, C/W TIDY	
	<u>ļ</u>		TANK, ALUM BOX, 349,302-KILOMETERS REPORTED AS OF AUGUST 2011	
RN060	2005	ROUGHNECK	T/A RV TRAILER, (LIC T537-66), C/W PROPANE PIG	
ST201	2005	GMC	S/A 4X4 SERVICE TRUCK, DURAMAX DIESEL ENGINE, AUTOMATIC TRANSMISSION, FRONT TIRES	
] 3,201	2005	1 31110	245/70R19.5, REAR TIRES 245/70R19.5, C/W SERVICE BODY, SERVICE TRUCK INTERNATIONAL TIGER	1
		1	PICKER, 10,000-LB CAPACITY, S/N 1688 35 FH, 269, 065-KILOMETERS INDICATED (NOT IN SERVICE	
1		1	ON DATE OF INSPECTION	1
UTC310	2005	TITAN	TANDEM-AXLE 12' UTILITY TRAILER, TIRES 285/85R16-STEEL BUDS, (LIC W060-63)	
0,0010		''''	The second secon	

Equip. #	Y	Make		
RN099	2006	The second secon	Description XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG	Location
INNOSS	2000	ROOGHINECK	ATREINE DOTT 20 TANDENI-AALE TRAILER, TIRES 235/80R16, PROPAINE PIG	ED
RN101	2006	ROUGHNECK	T/A RV TRAILER, CREW SHACK	ED
	1		The state of the s	LU
RN109	2006	N/A	RV TRAILER, TIRES 235/80R16 STEEL BUDS, 30' LENGTH	ED

RN111	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, 20', 235/85R16 TIRES, PROPANE PIG, KITCHEN BATHROOM, BEDROOM	ED
V529	2006	WESTERN STAR	VACUUM TRUCK, TRI-DRIVE	ED
		<u> </u>		
V531	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, 18-SPEED, 71217-KILOMETERS	ED
			INDICATED, 6929-HOURS INDICATED, 20000-LB FRONT AXLE, 23000-LB X 3 REAR AXLES, FRONT	
			445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIRLINER AIR RIDE SUSP, C/W CUSTOM VAC	
			SERVICES 19193-LITRE TANK, TC407/412, S/N 20581005, HIBON VTB-820 BLOWER, S/N 1550103,	
V533	2006	WESTERN STAR	147 089-KILOMETERS INDICATED. 13.855-HOURS INDICATED. TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED.	ED
*888		1	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	LD
		1	11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES	
			S/N 19711105, TC407/412, 19.193-LITRE CAPACITY, HIBON BLOWER	
V535	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED, 20000-LB	ED
			FRONT AXLE, 20400-LB X 3 REAR AXLES, FRONT 445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM	
			BUDS, AIR RIDE SUSP, C/W CUSTOM VAC UNIT, TC407/412 TANK 19193 L CAPACITY, HIBON	
			BLOWER, S/N 20621005	
V537	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED TRANSMISSION, AIR	ED
			RIDE SUSPENSION, C/W CUSTOM VAC, 19,193-LITRE CAPACITY, HIBON BLOWER	
V538	2006	MECTERNICTAR	TEN DENGE MACHINA TRUCK DOC CONTRICT AS HE ENGINE SATON SHILED AS COSED SO OCCUP.	
V 200	2006	MESSEUN STAU	TRI-DRIVE VACUUM TRUCK, DDC 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR AXLES, 11R24.5 REAR TIRES,	ED
			ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC S/N 20651205, 19,193-LITRE CAPACITY,	
ï			HIBON BLOWER	
V539	2006	WESTERN STAR	TRI-AXLE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, 62,931-	ED
		4	KILOMETERS INDICATED, 5,703-HOURS INDICATED, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-	
			LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB AXLES, AIR RIDE SUSPENSION, W/TIRE PRES	
			CTRL, C/W CUSTOM VAC SERVICE TANK, 19,193-LITRE TANK, S/N 119731105	
V542	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DDC 60-14-515-HP ENGINE, EATON FULLER 18-SPEED, 20000-LB FRONT	ED
			/ 20333-LB X 3 REAR AXLES, FRONT 445/65R22.5 ALUM BUDŞ, REAR 11R24.5 ALUM BUDS, AIR RIDE	
			SUSP, C/W CUSTOM VAC UNIT, 19193L TANK CAPACITY, S/N 20680905, HIBON VTB-820 BLOWER,	
V547	2006	MIECTEDALCTAD	167.263-KILOMETERS INDICATED. 17.284-HOURS INDICATED	
V547	2006	WESTERIVSTAR	VACUUM TRUCK, TRI-DRIVE, 104,227-KILOMETERS INDICATED, 13,371-HOURS INDICATED	ED
V548	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, EATON FULLER 18-SPEED, 16600-LB FRONT / 18100-LB X 3 REAR AXLES,	ED
13.0	2000	11/20121111011111	FRONT 445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE SUSP, C/W CUSTOM VAC	LD
			UNIT, TC407/412 TANK 19193 L CAPACITY S/N 20661205, HIBON BLOWER	
V549	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, 18-SPEED TRANSMISSION, FRONT	ED
		1	TIRES 445/65R22.5-ALUMINUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUMINUM BUDS.	
			69,000-LB AXLES, 19,000-LITRE CAPACITY	
V552	2006	WESTERN STAR	TRI-AXLE VACUUM TRUCK, W/TIRE PRES CTRL[BAD, VIN]	ED
1M240	2005	MESTERN STAR	TRI AVI E MATER TRI OV. DETROUT CERVES CO CANONES. EL TOUR	
W318	2006	1 1	TRI-AXLE WATER TRUCK, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ED
		1	FRONT TIRES 415/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS,	
		1	60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, S/N 5KKPALCK36PY20332, C/W ADVANCE	
			TANK S/N 2AESTKB005E000227, TC406, 22,000-LITRE CAPACITY, 4" BOWIE FLUID PUMP	
W320	2006	STERLING	TANDEM-AXLE WATER TRUCK, EATON FULLER 18-SPEED TRANSMISSION, 20,000-LB FRONT AXLE,	ED
			23,000-LB. REAR AXLES, ADVANCE BODY SPEC 400 TANK S/N 2AESTGA086E000153 W/ 20,000-LITRE	
			CAPACITY, 105,290-KILOMETERS INDICATED, 14,500-HOURS INDICATED	

Едиір. Н	Year		Description	Logatio
W324	2006	WESTERN STAR	WATER TRUCK, TRI-DRIVE	€D
W327	2006	WESTERN STAR	TRI-DRIVE TANK TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
		I	11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W ADVANCE TANK	
			S/N 2AESTKB046E000250, TC406, 22,000-LITRE CAPACITY, BOWIE 4" PRODUCTION PUMP, TOP	
AT013	2006	STEPHENS	SAFFTY RAII TRIDEM WATER TANK TRAILER, TIRES 11R24.5, STEEL BUDS, 60,000-LB AXLES, ALUM 38M3 TANK,	GP
ATULS	2005	2 LENDEIA2		GP
AT014	2006	STEPHENS	(LIC W132-88) TRIDEM ALUM WATER TANK TRAILER, 38M3 CAPACITY, TIRES 11R24.5-ALUM BUDS	GP
AIU14	2000	STEFFICIVS	TRIBLEW ALDIN WATER TANK TRAILER, SON'S CAPACITY, TIRLS IIRZ4.5-ALDIN BODS	Gr
CB911	2006	WESTERN STAR	TRI-AXLE COMBO TRUCK W/ 36" SLEEPER, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED	GP
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	
			BUDS, 60,000-LB AXLES, AIR RIDE SUSPENSION, (LIC G20867), C/W PROGRESS TANK S/N V-22142C,	
			APPROXIMATELY 12,000-LITRE CAPACITY, HIBON BLOWER, 198,251-KILOMETERS INDICATED, 9,119-	
			HOURS INDICATED	
CB912	2006	WESTERN STAR	TRI-DRIVE COMBO TRUCK, DETROIT DIESEL SERIES 60 ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, 20000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61200-LB REAR	
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, INTEGRAL SLEEPER, C/W	
			PROGRESS TANK S/N 22171C. HIBON BLOWER, BECKETT STEAMER	
NTR003	2006	WESTERN STAR	T/A TRACTOR W/ 36" SLEEPER, DETROIT 60 SERIES 14L 515-HP ENGINE, EATON FULLER 18-SPEED	ΕD
			TRANSMISSION, FRONT TIRES 385/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W WET KIT, SLIDING 5TH	
			WHEEL, 338,490-KILOMETERS INDICATED, 9,765-HOURS INDICATED	
NTR004	2006	WESTERN STAR	T/A HIGHWAY TRACTOR, SLEEPER, CAT, 385/65R22.5 FRONT, 11R24.5 REAR, ALUM BUD, AIR RIDE,	GP
			(LIC F-96643)	
PS115	2006	INTERNATIONA	S/A PRESSURE TRUCK, DT466 285-HP ENGINE, EATON FULLER 12-SPEED, FRONT TIRES 12,000-LB	GP
		L	FRONT AXLE, FRONT TIRES 11R22.5, ALUM BUDS, 21,000-LB REAR AXLE, REAR TIRES 11R22.5, ALUM	
			BUDS, AIR RIDE SUSPENSION, C/W JASPER 4250H STEEL TANK S/N 2J9JT10275S031041, 2	
			COMPARTMENTS - 2,100-LITRE/4,200 L,TC406, CAT TRIPLEX PUMP MODEL 3560, S/N 60406528,	
			339 502-KÜ OMETERS INDICATED 10 896-HOURS INDICATED	
RN084	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, (LIC V222-16), C/W PROPANE PIG	ED
ST204	2006	EREIGHTLINER	SERVICE BODY TRUCK W/INTEGRAL SLEEPER, 4X4, S/A, MERCEDES OM 906 LA ENGINE, EATON	GP
31204	2000	THE CONTENTENT	FULLER 9-SPEED TRANSMISSION, FRONT TIRES 11R22.5 TIRES-ALUM BUDS, 10,000-LB FRONT AXLE,	01
			REAR TIRES 11R22.5-ALUMIUNUM BUDS, 21,000 REAR AXLE, AIR RIDE SUSPENSION, (LIC F96616),	
			C/W BRUTUS SERVICE DECK, TIGER CRANE 3-TON MODEL 6000EH, MILLER WELDER, UNDER HOOD	
			AIR COMPRESSOR 188 882-KII OMETERS INDICATED 8 167-HOURS INDICATED	
TR007	2006	WESTERN STAR	T/A TRACTOR W/ 36" SLEEPER, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	GP
			TRANSMISSION, FRONT TIRES 385/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	
	1		11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIR RIDE SUSPENSION, S/N 5KJJALCK76PY17783, C/W	
			WET KIT. SLIDING 5TH WHEEL. BOWIE 4" PRODUCTION PUMP	
TR008	2006	WESTERN STAR	T/A TRACTOR, SLEEPER, DETROIT SERIES 60 ENGINE, 515-HP, EATON FULLER 18-SPEED	GP
			TRANSMISSION	
TR009	2006	WESTERN STAR	T/A HIGHWAY TRACTOR, INTEGRAL SLEEPER, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	GP
			16,000-LB FRONT AXLE, 425/65R22.5 FRONT TIRES, ALUM BUDS, 46,000-LB REAR AXLES, 11R24.5	
			REAR TIRES, ALUM BUDS, C/W WET KIT, 199,029-KILOMETERS INDICATED, 8,275-HOURS INDICATED	
TR011	2006	WESTERN STAR	T/A TRACTOR, SLEEPER, DETROIT SERIES 60 ENGINE, 18-SPEED TRANSMISSION	GP
V540	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	GP _.
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIR RIDE SUSPENSION, (LIC G-10766), C/W CUSTOM	
		1	VAC SERVICES.S/N 20671205, TC407/412, 19,193-LITRE CAPACITY, HIBON BLOWER	

TT 17 77 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7				
Equip.#	Year	Make	Description	Location
V545	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED, 20000-LB FRONT AXLE,	€D
			20000-LB X 3 REAR AXLES, FRONT 445/65R22.5, ALUM BUDS, 11R24.5 REAR ALUM BUDS, AIR RIDE	
			SUSP, C/W CUSTOM VAC UNIT, TC407/412 TANK 19193 L CAPACITY S/N 19721105, HiBON BLOWER,	
			HOIST, AGITATOR, VIBRATOR [NOT IN SERVICE ON DATE OF INSPECTION]	
W322	2006	INTERNATIONA	6X4 WATER TRUCK, ENGINE, AUTOMATIC TRANSMISSION, FRONT TIRES 1.1R22.5-STEEL BUDS, REAR	ZC
		L	TIRES 11R22.5-STEEL BUDS, AIR RIDE SUSPENSION, C/W TREMCAR TANK, 16,000-LITRE, S/N 4962,	
			116,180-KILOMETERS INDICATED, 11,182-HOURS INDICATED	
W323	2006	INTERNATIONA	T/A WATER TRUCK, DT 570 9.3 L ENGINE, AUTO, FRONT TIRES 11R22.5, STEEL BUDS, REAR TIRES	ED
		L	11R22.5, STEEL, C/W TREMCAR ALUM TANK, S/N 4961, 16,000-LITRE, BOWIE PUMP, 187,950-	
			KILOMETERS INDICATED	
V530	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ED
		}	FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS,	
		İ	61,200-LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES S/N 20560905,	
			TC407/412. 19.193-LITRE CAPACITY. HIBON BLOWER	
W325	2006	WESTERN STAR	TRI-AXLE TANK TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 441/65R22.5-ALUM	CNRL
			BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION,	HORIZON
		ĺ	(LIC G10752) S/N 5KKPALCK86PY66724, C/W ADVANCE TANK, TC 406 SPEC, 22,000-LITRE CAPACITY -	PROJECT
H709	2006	MECTEDNICTAD	S/N 2AFSTKB045F000229 TRI-DRIVE HYDRO-VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	ED.
11709	2006	ME21 EKIA 21 MK	TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,000-LB REAR	ED
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES	
			HYDRO-VAC UNIT S/N 19640605, TC407/412 TANK, 10,822-LITRE CAPACITY, BOC EDWARDS BLOWER HOTSY HEATER HOTSY STEAMER WATER BLAST SYSTEM	
H711	2006	WESTERN STAR	TRI-DRIVE HYDRO VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	ED
117.11	2000	WESTERN STAR	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	
			BUDS, 61,200-LB AXLES, AIR RIDE SUSPENSION, (LIC F77613), C/W CUSTOM VAC S/N 19660805,	
			TCRN Z.05.438.05.04, TC 402/412, 10,822-LITRE CAPACITY, ROBUSCHI BLOWER S/N 05-06887	
			TORRY 2.03.438.03.04, TO 402/412, 10,022 ETRE CALACITY, ROBOGOTH DEGWER S/R 03-00007	
H712	2006	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, FRONT AXLE	ED
			445/65R22.5 ALUM BUDS 20,000-LB, REAR AXLES 11R24.5 ALUM BUDS 61,200-LB, AIR RIDE	
			SUSPENSION, C/W CUSTOM VAC SERVICES HYDRO-VAC, 10,822 L CAPACITY, TC407/412 SPEC S/N	
			19670805, C/W HYDRAULIC HOIST, REAR DOOR, VIBRATOR, DUMP GATE & KNIFE VALVE, HOTSY,	
			HYDRAULIC BOOM	
H724	2006	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED,	CNRL
			20,000-LB FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES, REAR TIRES	
			11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES HYDRO-VAC UNIT S/N	
			24590306, 10,822 L TANK, TC 407/412, EDWARDS BLOWER, HOTSY STEAMER & WATER BLAST	
			SYSTEM, HYDRAULIC BOOM, 140,019-KILOMETERS INDICATED, 11,187-HOURS INDICATED	
P168	2006	CHEVROLET	PICKUP TRUCK, 3/4-TON, CREW CAB, DURAMAX DIESEL	ED
RN077	2006	ROUGHNECK	T/A RV TRAILER, PROPANE PIG, KITCHEN, BATHROOM, BEDROOM, TIRES 235/80R16	ED
ST206	2006	FREIGHTLINER	SERVICE TRUCK W/ INTEGRAL SLEEPER, S/A, MERCEDES BENZ ENGINE, EATON FULLER 10-SPEED	ED
			TRANSMISSION, FRONT TIRES 11R22.5-ALUM BUDS, 10,000-LB AXLE, REAR TIRES 11R22.5-ALUM	
			BUDS, 21,000-LB AXLE, (LIC G10836), C/W TIGER CRANE MODEL 6031C, 3-TON BLOCK, BRUTUS	
1000	2000	WEGTERN GT: 5	SERVICE DECK	
V544	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED, 141,368-KILOMETERS	ED
			INDICATED, 14,690-HOURS INDICATED, 16600-LB FRONT AXLE, 17750-LB X 3 REAR AXLES, FRONT	
			445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE SUSP, C/W CUSTOM VAC UNIT,	
RN068	2006	ROUGHNECK	TC407/412 TANK 19193 L CAPACITY. S/N 23030106. HIBON VTB-820 BLOWER T/A RV TRAILER, (LIC V068-26), C/W PROPANE PIG	ED .
NNADO	2006	ROOGHNECK	· · · · · · · · · · · · · · · · · · ·	ED
RN073	2006	ROUGHNECK	XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG	ED

Equip. 4	Year	Make		Location
RN075	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER	ED
DA1070	2006	POLICUNICAL	TDANED CDEMICIACY	ED
RN078	2006	ROUGHNECK	TRAILER, CREW SHACK	ED
RN088	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, (LIC V381-00), C/W PROPANE PIG	ED
1,11000				
RN103	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, 235/85R16 TIRES, PROPANE PIG, KITCHEN BATHROOM, BEDROOM	ED
	<u>.</u>			
RN104	2006	ROUGHNECK	XTREME DUTY TANDEM-AXLE TRAILER, TIRES 235/80R16-STEEL BUDS, PROPANE PIG, (LIC P842-99)	ED
			WALL AND SHARWAY TO A CTOP WAS INCOME.	
NTRO02	2006	ME21 EKM 21 AK	TANDEM-AXLE HIGHWAY TRACTOR W/SLEEPER, DETROIT DIESEL SERIES 60 14L 515-HP ENGINE,	ED
			EATON FULLER 18-SPEED, FRONT TIRES 385/65R22.5, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE	
			SUSPENSION, (LIC F-77618), 253,514-KILOMETERS INDICATED, 5,941-HOURS INDICATED	
RN087	2006	ROUGHNECK	T/A RV TRAILER, KITCHEN, BED, 100-LB PROPANE RIG	ED
1111007	2000	THOUGHT THE GIVE	The Handley Stay 200 to Harring	
RN090	2006	ROUGHNECK	T/A RV TRAILER, (LIC V381-02), C/W PROPANE PIG	ED
RN091	2006	ROUGHNECK	XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG, S/N	ED
			2TTNT30046R600431	
V528	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, 165,136-KILOMETERS INDICATED	
V534	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 445/65R22.5 FRONT	ED
	1		TIRES, 11R24.5 REAR TIRES, AIR RIDE SUSPENSION, TIRE PRESSURE MONITOR, (LIC. F96632), C/W	
			CUSTOM VAC UNIT, TC 407/412, 19,193-LFTRE CAPACITY, S/N 20511005, HIBON BLOWER	
V543	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	ED
¥343	2000	TVESTERIT STATE	TRANSMISSION	Lo
W317	2006	WESTERN STAR	TRI-DRIVE TANK TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM	ED
			BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION,	
		•	(LIC F96634), C/W ADVANCE TANK, TC SPEC 406, 22,000-LITRE CAPACITY - S/N 2AESTKB055E00024	
			[SUCKED-IN DENTED BACK PANEL DUE TO DISCHARGING WITHOUT VENTING; STILL USABLE]	
H 721	2006	WESTERN STAR	HYDRO VACUUM TRUCK, CATERPILLAR C15 475-HP ENGINE, EATON FULLER 18-SPEED	CNRL
			TRANSMISSION, CUSTOM VAC SERVICES 10,822-LITRE TANK	ON D
H726	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	CNRL
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	
			BUDS, 60,000-LB AXLES, AIR RIDE SUSPENSION, 36" SLEEPER, C/W CUSTOM VAC SERVICES, 10,822-	
H731	2006	WESTERN STAR	LITRE TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED,	ED
117.51	2000	WESTERNSTAN	20,000-LB FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES, REAR TIRES	2.5
			11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G20885), C/W CUSTOM VAC SERVICES HYDRO-	
			VAC UNIT S/N 25370606, 10,822 L TANK, TC 407/412, ROBUSHI BLOWER, HOTSY STEAMER & WATER	
			BLAST SYSTEM: HYDRAULIC BOOM. NEW DROP BOX	
H734	2006	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	. ED
			445/65R22.5, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G-19638), C/W CUSTOM	
			VAC UNIT, S/N 25320406, 10.822 L, TC407/412	
NPO09	2006	WESTERN STAR	TRI-AXLE NITROGEN PUMP UNIT TRUCK, CAT C15 ACERT 475-HP ENGINE, EATON FULLER 18-SPEED	GP
			TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 69,000-LB REAR	
			AXLES,11R24.5 REAR TIRES, ALUM BUDS, AIRLINER AIR RIDE SUSPENSION, SLEEPER/CONTROL CAB	
			COMBO, (LIC G20901), C/W CRYOGENIC VESSEL ALTERNATIVES NITROGEN VESSEL S/N 30135, MC-	
			338, 26,370-LB WATER CAPACITY, BLACKMORE MODEL TX03E PUMP, BUTECH PRESSURE SYSTEM,	
			XCHANGER INC. MODEL C-250 COIL HEATER, 10 1"X10' JOINTS, 5 5' JOINTS, 195,421-KILOMETERS	
			INDICATED 10.521-HOURS INDICATED	
NST003	2006	AMKO	TRIDEM NITROGEN TANK TRAILER, 69,000-LB AXLES, TIRES 11R225, ALUM BUDS, AIR RIDE	ED
			SUSPENSION, (LIC N65174), C/W TC341 TANK S/N V4472, 36,877 L TANK	



Equip. #	- Year	Make	Description	koestiisa
NST004	2006		TRI-AXLE NITROGEN TANK TRAILER, 69,000-LB AXLES, TIRES 11R22.5, AIR RIDE SUSPENSION, C/W	ED
		I .	ALLOY TANK, TC341, 82361-LB CAPACITY, TCRN Z053590305, TANK S/N V4473, 9,742-GALLON	
		E .	CAPACITY	
RN079	2006		TRAILER, CREW SHACK	ED
RN102	2006	ROUGHNECK	XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG	ED
RN105	2006	ROUGHNECK	XTREME DUTY 20' TANDEM-AXLE TRAILER, THRES 235/80R16, PROPANÉ PIG	ED
V551	2006	WESTERN STAR	TRIDRIVE VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED	ĊNRL
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
•			SERVICES STEEL TANKS/N 23060106, TC SPEC 407/412, 19,193-LITRE CAPACITY, HIBON VTB-820	
V557	2006	WESTERN STAR	BLOWER S/N 37933510. TIRE PRESSURE CONTROL SYSTEM. TRI-DRIVE VACUUM TRUCK W/ SLEEPER, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-	CNRL
V357	2008		l	CINIL
			SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
•	1		11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CARGO TANK,	
			TC407/412. 15.959-LITRE CAPACITY, HIBON BLOWER	·
AT007	2006	HEIL	TRIDEM ALUM 407 WATER TANKER, 37,854-LITRE CAPACITY, TIRES 11R24.5-STEEL BUDS, 60,000-LB	GP
			AXLES, AIR RIDE SUSPENSION, (LIC V158-52)	
AT011	2006	N/A	TRIDEM WATER TANKER, 10,000-GALLON TANK	ZC
V532	2006	IMECTEDNI CTAD	TRI-DRIVE VACUUM TRUCK, DDC 60 SERIES ENGINE, EATON FULLER 18-SPEED, 20,000-LB FRONT	GP
V332	2006			Gr
			AXLE, 445/65R22.4 FRONT TIRES, ALUM BUDS, 61,200 REAR AXLES, 11R24.5 REAR TIRES, ALUM	
			BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC, TC407/412 TANK 19193L CAPACITY S/N 20591005,	
		1	HIBON BLOWER	
H73 6	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED	ED
•			TRANSMISSION, ROBUSHI BLOWER, HOTSY HOT WATER SYSTEM, ONAN HYDRAULIC BOOM,	
			CUSTOM VAC TANK S/N 24700306 W/ 10,822-LITRE CAPACITY	
RN115	2006	ROUGHNECK	XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG	ED
RN116	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, PROPANE PIG, KITCHEN, BATHROOM, SEDROOM, 235/80RTIRES	ED
H754	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, 18-SPEED TRANSMISSION, C/W CUSTOM VAC SERVICES STEEL	, ED ·
			TANK [DAMAGED IN ROLLOVER; NOW OPERATIONAL; UNIT APPRAISED AS IN REPAIRED CONDITION]	
W328	2006	PETERBILT	T/A WATER TRUCK, CAT C7 ENGINE, EATON FULLER 8 TRANSMISSION, FRONT TIRES 11R22.5-ALUM	ED
			BUDS, 12,000-LB AXLE, REAR TIRES 11R22.5-ALUM BUDS, 40,000-LB AXLES, AIR RIDE SUSPENSION,	
			(LIC G57786), C/W ADVANCE TANK S/N 2AESTGBOX5S000136, 14,320-LITRE CAPACITY	
				•
AT010	2006	HEIL	TRIDEM ALUM WATER TANKER, TIRES 11R24.5-STEEL BUDS, HENDRICKSON AIR RIDE SUSPENSION,	ZAMA CITY
			C/W ALUM TANK, DOT 407, 37,850-LITRE CAPACITY, EXTERIOR INSULATION	
NTR001	2006	WESTERN STAR	T/A HIGHWAY TRACTOR, SLEEPER, DETROIT DIESEL 14L 515-HP, EATON FULLER, 385/65R22.5	ZAMA CITY
			FRONT, 11R24.5 REAR, ALUM BUDS, 23000-LB X 2 AXLES, (LIC F-77617)	
SU012	2006	WESTERN STAR	TRI-DRIVE SOUR SEAL TRUCK W/ 32" SLEEPER, DDC 60 515-HP ENGINE, EATON FULLER 18-SPEED	ZAMA CITY
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			48X31.00-20 NHS ND, 60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
		i i	SERVICES LTD STEEL TANK S/N 25600706, TC406, 20,123-LITRE CAPACITY, 3" T&E FLUID PUMP	
			The state of the s	
V550	2006		TRI-DRIVE VACUUM TRUCK, DDC 60-14 515-HP ENGINE, EATON FULLER 18-SPEED 20000-LB FRONT /	ZAMA CITY
			20333-LB X 3 REAR AXLES, FRONT 445/65R22.5, REAR 11R24.5, AIR RIDE SUSP, C/W CUSTOM VAC	
			UNIT, 19193 L CAPACITY TANK S/N 23040105, HIBON VTB-820 BLOWER	
W321	2006	INTERNATIONA	T/A WATER TRUCK, ALLISON AUTO TRANS, FRONT 11R22.5 STEEL BUDS, REAR 11R22.5 STEEL BUDS,	GP
		L	C/W TREMCAR ALUM TANK, 16000-LITRE CAPACITY S/N4963	
P161	2006	FORD	PICKUP TRUCK, CREW CAB, 3/4-TON, 75-GALLON WESTEEL DIESEL TANK W/ 12V DC PUMP, 287,237-	MEDICINE
			KILOMETERS INDICATED	HAT
			INCOME LEG MATCHED	1 (1)

Equip #	Year	Make	Description	Location
P178	2006	CHEVROLET	PICKUP TRUCK, 4X4 CREW CAB S/B, DURAMAX ENGINE, AUTOMATIC TRANSMISSION, TIRES	MEDICINE
			LT265/75R16, MOTOROLA PHONE, CROSS OVER TOOL BOX, HITCH, 253,404-KILOMETERS INDICATED	HAT
P188	2006	DODGE	PICKUP TRUCK, MEGA CAB, 4WD, TURBO DIESEL ENGINE [NOT IN SERVICE ON DATE OF INSPECTION]	MEDICINE
1 100	2000	DODGE	FICKOF TROCK, MEDA CAD, 4WD, TORBO DIESEL ENGINE [NOT IN SERVICE ON DATE OF INSPECTION]	HAT
P191	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, TURBO DIESEL ENGINE	MEDICINE
				HAT
P196	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, ALLISON DURAMAX, 471-LITRE TANK W/ ELECTRIC PUMP	MEDICINE
D407	2005	OUE VE OU ET	DIGINIS TRUEY SPENGAR AND SURPLINES AND SURP	HAT
P197	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, DURAMAX DIESEL, 4X4, (LIC G10772)	MEDICINE
P257	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, DURAMAX DIESEL	HAT SC
1237	1 2000	0,,2,,,3221	Traces (Traces) and traces and property and	. 50
RN076	2006	ROUGHNECK	T/A RV TRAILER, C/W PROPANE PIG	MEDICINE
				HAT
RN086	2006	ROUGHNECK	T/A RV TRAILER (LIC V222-18), C/W PROPANE PIG	MEDICINE
5004	2000	DETERMIT	STEAM TRUCK, ESTIMATED 8,000-LITRE CAPACITY TANK, 24' BOX VAN, THERMO KING MODEL 4-	HAT
S004	2006	PETERBILT	15822, S/N VB3601 BOX	SC
UT105	2006	CUSTOM	TRAILER, WELDING, 10' FLATDECK, W/ GENERATOR	MEDICINE
		MANUFACTUR		HAT
		ED		
AT009	2006	HEIL	TRIDEM WATER TANKER, 10,000-GALLON TANK	SWIFT
				CURRENT
H729	2006	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, FULLER 18-SPEED, 20,000-LB REARS, CUSTOM VAC UNIT S/N	SWIFT
H732	2006	WESTERNISTAR	25350506, 10,822L CAP, DUMP BED, WATER BLAST BOILER TRI-DRIVE HYDRO VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED	CURRENT SWIFT
"",32	2000	WESTERNSTAR	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	CURRENT
			BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION, (LIC G20877), C/W CUSTOM VAC S/N 25290606,	COMMENT
			TCRN Z.05.437.05.04, TC 407/412, 10,822-LITRE CAPACITY, ROBUSCHI BLOWER S/N 05-27026	
	ļ			
H740	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 20,000-LB	SWIFT
			FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES, REAR TIRES 11R24.5,	CURRENT
			ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC UNIT, TC407/412, 10,822 L, ROUSCHI BLOWER, HOTSY, WATER BLAST WB9460 HEATER, HYDRAULIC BOOM, S/N 27270806	
			BLOWER, HOTST, WATER BLAST WB9460 HEATER, HTDRAOLIC BOOW, S/N 272/0806	
H742	2006	WESTERN STAR	TRI-DRIVE HYDRO VAC W/ SLEEPER, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED	SWIFT
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	CURRENT
			BUDS, 61,200-LB AXLES, AIR RIDE SUSPENSION, (LIC G20946), C/W CUSTOM VAC S/N 27910906,	
9250	2000	CUEVBOLET	TCRN Z.OS.437.0S.04. TC 407/412, 10822-LITRE CAPACITY	CHOLL
P256	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, SHORT BED, DURAMAX DIESEL ENGINE, ALLISON AUTOMATIC, TIRES 265/75R16, FULLY LOADED, C/W SLIP TANK AND PUMP INOT IN SERVICE AT TIME OF INSPECTION!	SWIFT CURRENT
			2007/3R.to, POLLT LOADED, C/W SEP TANK AND POMP [NOT IN SERVICE AT TIME OF INSPECTION]	CORNEIVE
P262	2006	GMC	PICKUP TRUCK, CREW CAB 4X4 S/B, DURAMAX DIESEL ENGINE, ALLISON TRANSMISSION, TIRES	SWIFT
			265/75R16, C/W ALUM CABINET, 175,018-KILOMETERS INDICATED	CURRENT
RN114	2006	ROUGHNECK	T/A RV TRAILER, CREW SHACK	MH
FTOO	2000	(METERNATION:	NACING E AVE CENTICE TRUCK PARE FLOIR FATOUR FATOUR FOR A CREEK PRODUCTION	C)4//
ST208	2006	1	4X4 SINGLE-AXLE SERVICE TRUCK, D255 ENGINE, EATON FULLER 18-SPEED, FRONT TIRES 11R22.5, STEEL BUDS, (LIC G19632), C/W CRANE & RIG BODY S/N	SWIFT
		L L	CRT060801, CRANE RKI3200 ERX15 S/N 32003, COMPRESSOR S/N 174730606, GETEC 250-5.5	CURRENT
			WELDER INOT IN SERVICE AT TIME OF INSPECTION	
UT357	2006	SWS	T/A FLATDECK GOOSENECK TRAILER, TIRES ST235/80R16-STEEL BUDS, 14,000-LB AXLES, (LIC V510-	SWIFT
			82)	CURRENT
V554	2006	KENWORTH	PRODUCTION VACUUM TRUCK, TANDEM, CW CUSTOM VAC TANK, S/N ZO543704TC407/412, 13,612-	SWIFT
	<u> </u>	1	LITRE, 160,827-KILOMETERS, 8134-HOURS INDICATED	CURRENT



Big Eagle Limited Partnership Asset List* GA Global Partners Report

Equip 4	7	Make	Description	. Morealitin
V555	2006		T/A VACUUM TRUCK, CUMMINS ISX-500 ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	SWIFT
V 255	2000	KENWORTH	445/65R22.5 ALUM BUDS, REAR TIRES 11R24.5 ALUM BUDS, NEWAY AIR RIDE SUSPENSION, C/W	CURRENT
			CUSTOM VAC UNIT, HIBON BLOWER, TC407/412, 13612 L CAP, S/N 23381205, 295,975-KILOMETERS,	COMILITY
	, i		15.474-HOURS REPORTED	
V556	2006	WESTERN STAR	VACUUM TRUCK, TANDEM, CUSTOM VAC 13,612L VAC UNIT, 227,484-KILOMETERS, 15,488-HOURS	SWIFT
•350	2000	WEST ENTRY OF	REPORTED	CURRENT
VR127	2006	KENWORTH	TRI-DRIVE VACUUM TRUCK W/ 36" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	SWIFT
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	CURRENT
			11R24.5-ALUM BUDS, 52,000-LB REAR AXLES, NEWAY AIR RIDE SUSPENSION, C/W WESTECH VAC	
			SYSTEM S/N WT0705045, TC407, 14,600-LITRE CAPACITY, HIBON BLOWER, 88,713-KILOMETERS	
			INDICATED, 667-HOURS INDICATED	
VTR001	2006	WESTERN STAR	SEMIVAC HIGHWAY TRACTOR, TRI-AXLE	GP
		·		
P250	2006	DODGE	PICKUP TRUCK, MEGA CAB, TIRES 265/70R17, C/W BUCKET SEATS, CUMMINS TURBO DIESEL	SC
			ENGINE, 4WD	
CT2061	2006	KENWORTH	TANDEM-AXLE STEER TRI-AXLE COILED TUBING UNIT, CAT ENGINE, EATON FULLER TRANSMISSION,	ED
			FRONT TIRES 425/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION,	
			C/W FASSI F150AC.24 7-TON PICKER S/N 4994, SIDE OUTRIGGERS, 60" GOOSENECK, HYDRA RIG RT25	
			INJECTOR, OPERATORS CABIN	
AT012	2006	STEPHENS	TRIDEM WATER TANK TRAILER, 60,000-LB AXLES, TIRES 11R24.5, ALUM BUDS, ALUM 38M3	MEDICINE
			CAPACITY, (LIC V510-83)	HAT
V553	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	ED
		·	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUMINUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-	
			ALUMINUM BUDS	
H717	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	ZC
			445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC UNIT, S/N	
		<u> </u>	20491205, 10,822-LITRE	
RN098	2006 ⁻	ROUGHNECK	T/A RV TRAILER, (LIC V381-30), C/W PROPANE PIG	МН
RN107	2006	ROUGHNECK	T/A RV TRAILER (LIC V455-19), C/W PROPANE PIG	мн
TWP013	2006	CARGO	TRIDEM STIFF POLE TANK PUP TRAILER, TIRES 11R24.5-ALUM BUDS, AIR RIDE SUSPENSION, (LIC	ED
		1	G50681), C/W ALUM TANK S/N V-219106, TC407/412, 15,985-LITRE CAPACITY	
TRP802	2006	GMC	PICKER TRUCK, QUAD CAB, FLAT DECK, FRONT TIRES 245/70R19.5, ALUM BUDS, REAR TIRES	MH
			245/70R19.5, ALUM BUDS, C/W PALFINGER PK7501 KNUCKLE PICKER S/N 100041114, FLAT DECK,	
			CABINETS, 5TH WHEEL, SIDE OUTRIGGERS, 196,659-KILOMETERS AS OF JULY 2011	
UT202	2006	MUSTANG	TRAILER, GOOSENECK, STEP DECK, 10' TOP DECK, 30' WORKING DECK	MH
			·	
RN080	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER	
V527	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	
			FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,000-	
			LB AXLE, AIR RIDE SUSPENSION, (LIC F77620), C/W CUSTOM VAC, TC SPEC 407/412, 19,193-LITRE	
			CAPACITY - S/N 20570905, HIBON BLOWER	
P251	2006	CHEVROLET	PICKUP TRUCK, CREW CAB 4X4 SHORT BED, DURAMAX DIESEL ENGINE, LT 245/75R16 TIRES, ALUM	•
, mae	2000	WESTERN STAR	BUDS, C/W TIDY TANK, ALUM TOOL BOX, LEATHER, FIBERGLASS CANOPY	
V536	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, W/TIRE PRESSURE CONTROL, DETROIT 60 SERIES ENGINE, EATON	
			FULLER 18-SPEED, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR	
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC UNIT,	
SAIDA C	3000	IMPEREDAL CEAR	TC407/412, 19193 L CAP, HIBON VTB820, S/N 20631005	
W316	2006	ANESTERN STAR	TRI-DRIVE TANK TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W ADVANCE TANK, TC403,	
		1	22,000-LITRE CAPACITY, BOWIE 4" PRODUCTION PUMP, TOP SAFETY RAIL, 138,175-KILOMETERS	
	I	<u> </u>	INDICATED, 20.128-HOURS INDICATED	l

Source #	Year	Make	Description	Lorealtion
W319	2006		TRI-AXLE WATER TRUCK, DETROIT 60 SERIES ENGINE, EATON FULLER TRANSMISSION, FRONT	T237711 E. ANT (AUST)
.,,,,,			445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE SUSPENSION, C/W ADVANCE TANK	
			TC 406. 22.000-LITRE CAPACITY S/N 2AESTKB075E000225	
V541-A	2006	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, PURCHASED USED IN 2014, DETAILS NOT AVAILABLE, ASSUMED	
		1	SIMILAR TO LIKE ASSETS AND REBUILT BASED ON COST INFO PROVIDED	
AT015	2006	STEPHENS	TRIDEM ALUM WATER TRAILER, TIRES 11R24.5-STEEL BUDS, 60,000-LB AXLES, (LIC W132-94)	
HUTO02	2006	NORBERTS	TRIDEM ENCLOSED GLYCOL HEATING UNIT, TIRES LT235/85R16-STEEL BUDS, 21,000-LB AXLES, (LIC	МН
			V510-90) S/N 2N9E7S7416G017303, C/W HEATER WATER BLAST MODEL WB9460-SN 61781206,	
			WATER PRESSURE WAND, 2-832-LITRE CAPACITY TANKS, 500-LITRE CAPACITY TANK, 4,542-LITRE	
			CAPACITY TANK, MITSUBISHI DIESEL GENERATOR	
RN072	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, PROPANE PIG	ED
V546	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	
			TRANSMISSION	
ST207	2006	FREIGHTLINER	SERVICE TRUCK, S/A, 4X4, EXTENDED CAB, MERCEDES BENZ ENGINE, 8-SPEED TRANSMISSION,	ZAMA CITY
			FRONT TIRES 11R24.5-ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS,	
			21,000-LB REAR AXLE, C/W BRUTUS 8'X14' SERVICE BODY W/ TIGER 6541E TRUCK CRANE, 40,000-LB	
			S/N 102441. FRONT BUMPER WINCH	
TR010	2006	WESTERN STAR	T/A TRACTOR, SLEEPER, DETROIT SERIES 60 DIESEL ENGINE, 515-HP, EATON FULLER 18-SPEED	ZAMA CITY
			TRANSMISSION	
TWP014	2006	CUSTOM VAC	TRIDEM STIFF POLE TANK TRAILER PUP, TIRES 11R24.5-ALUM BUDS, C/W CUSTOM SERVICES VAC	
		SERVICES	STEEL TANK S/N 28401106, TC407, 20,123-LITRE CAPACITY, TITAN LOGIX LEVEL METER	
CD2061	2007		SINGLE-AXLE BOOM TRUCK 4X4, MERCEDES BENZ ENGINE, ALLISON AUTOMATIC, FRONT TIRES	ED
			315/80R22.5, ALUM BUDS, C/W FASSI F110A-24 KNUCKLE BOOM CRANE, S/N 6779, SIDE	
			OUTRIGGERS, DECK	
CD2062	2007	FREIGHTLINER	BOOM TRUCK, SINGLE AXLE, 4X4, MERCEDES BENZ ENGINE, AUTOMATIC TRANSMISSION, 1LB FRONT	ED
			AXLE 11R24.5 FRONT TIRES, ALUM BUDS, 21000-LB REAR AXLE, 11R24.5 REAR TIRES, ALUM BUDS,	
			AIR RIDE SUSPENSION, INTEGRATED SLEEPER, C/W FASSI F110A-24 KNUCKLE BOOM CRANE, S/N	
			7076, SIDE OUTRIGGERS, DIESEL, METHANOL TANK, TOOL CABINET, 12' DECK, 150,485-KILOMETERS	
			INDICATED 5 490-HOURS INDICATED	
CD2072	2007	FREIGHTUNER	BOOM TRUCK, SINGLE-AXLE, 4X4, MERCEDES BENZ ENGINE, FRONT TIRES 11R22.5-STEEL BUDS,	ED
			12,000-LB AXLE, REAR TIRES 11R22.5-STEEL BUDS, 21,000-LB AXLE, (LIC G-57779) S/N	
			1FYDCXCS07HY24000, C/W FASSI F110A.24 KNUCKLE BOOM CRANE- S/N 6788, 12' DECK 110,904-	
CD2072	2007	FOCICITUMES	KILOMETERS INDICATED. 3.618-HOURS INDICATED BOOM TRUCK, SINGLE-AXLE, 4X4, W/ INTEGRAL SLEEPER, MERCEDES BENZ ENGINE, ALLISON	ED
CD2073	2007	PREIGHTLINER	1	ED
•			TRANSMISSION, 12,000-LB FRONT AXLE, FRONT TIRES 11R24.5, STEEL BUDS, 21,000-LB REAR AXLE,	
			REAR TIRES 11R24.5, STEEL BUDS, (LIC G-64535) S/N 1FYDCXCSX7HX29699, C/W FASSI F100 A.24	
		1	KNUCKLE BOOM CRANE- S/N 7044, 12' DECK 89,600-KILOMETERS INDICATED, 3,837-HOURS	
CT2071	2007	KENWORTH	INDICATED: TANDEM-AXLE STEER/TRI-AXLE DRIVE COILED TUBING UNIT, CAT C15 550-HP ENGINE, EATON	ED
C120/1	2007	KENWOKIH	FULLER 18-SPEED, 32,000-LB FRONT AXLES, FRONT TIRES 425/65R22.5, ALUM BUDS, 69,000-LB REAR	LD
			AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, SPICER 792, C/W COLLED TUBING	
		İ		
			UNIT, 3100M OF 1 1/4" TUBING, HYDRA RIG RT25 INJECTOR - 3/4" TO 2", CAT 4517 TRIPLEX PUMP,	
			50L/MIN @ 21 MPA, W/NORTHWEST MODEL 5450 COMPRESSOR, 18M3 @ 14 MPA, W/ AMCO	
			VEBA V815F-45 PICKER, 6.7M REACH, S/N 20159, 162,147-KILOMETERS INDICATED, 9,538-HOURS	
NP012	2007	WESTERNI STAD	INDICATED TRI-AXLE NITROGEN PUMP UNIT W/ SLEEPER/CONTROL CAB COMBO, CAT C15 ENGINE, EATON	ED
INFOIZ	2007	MEDIEVINDIAL	FULLER 18-SPEED, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES, REAR TIRES	ED
			11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G-57750), C/W WELLQUIP NITROGEN PUMP UNIT	
			S/N 0506-89906, XCHANGER C-250 ACD RECIPROCATING TRIPLEX PUMP, CVA TANK 11,234 LITRE,	
			S/N 30202, TCRN 25670, TC341 TANK, C/W TOOLING & DOWNHOLE TOOLS, 218456-KILOMTERES	
P270	2007	GMC	INDICATED 8064-HOLIRS INDICATED PICKUP TRUCK, 3/4-TON, CREW CAB, 4WD, FUEL PUMP, TURBO DIESEL ENGINE, 214,989-	ED
1270	2007	GIVIC	KILOMETERS INDICATED	LD
		 		
P276	2007	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, SHORT BED, DURAMAX ENGINE, AUTOMATIC TRANSMISSION,	ED

Equipli	Year	Wake	Description	20(0)(0)1
CB908	2007	WESTERN STAR	TRI-DRIVE COMBO TRUCK, 32" SLEEPER, DETROIT SERIES 60 515-HP ENGINE, 18-SPEED, FRONT TIRES	GP
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 23000-LB X 3 AXLES,	
			AIRLINER AIR RIDE SUSPENSION, C/W CARGO TANK MFG 2-COMPARTMENT TANK, WATER FRONT	
		i	1,216-GALLON, 2,000-PS! WATER PUMP, REAR 2,978-GALLON VACUUM, 11,299-LITRE VACUUM	
			CAPACITY, TC SPEC 407/412, S/N V22326C, HIBON VTB-820 BLOWER, HOTSY WB 9460 HEATER	
			IREPORTED AS NO LONGER IN SERVICE!	
CB909	2007	WESTERN STAR	TRI-DRIVE COMBO VACUUM TRUCK W/ 36" SLEEPER, DETROIT SERIES 60 515-HP ENGINE, EATON	GP
			FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR	
			TIRES 11R24.5-ALUM BUDS, 69,000-LB AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CARCO 2	
			COMPARTMENT TANK S/N V22222C, DOT407/412, 4,675-LITRE CAPACITY , 11,299-LITRE CAPACITY ,	
			HIBON VTB 820 BLOWER, WATER BLAST HEATER, 2,000-LITRE WATER PUMP CAPACITY [REPORTED	
			AS NO LONGER IN SERVICE!	
H 738	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, 18-SPEED, 59,314-	GP
			KILOMETERS INDICATED, 5,484-HOURS INDICATED, FRONT AXLE 445/65R22.5 ALUM BUDS 20,000-	
			LB, REAR AXLES 11R24.5 ALUM BUDS 69,000-LB, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
			SERVICES LTD. 10,822-LITRE VACUUM CAPACITY, TC SPEC 407/412, S/N 27260806, ROBUSCHI RB DV	
			145 SP UN BLOWER, S/N 06-28427, HOTSY 9460 WATER BLAST HEATER, S/N 60160706; 2,000-PSI	
			WATER PUMP, ESTIMATED 5,000-LITRE CAPACITY WATER TANK	
H746	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT	GP
,-			TIRES 445/65R22.5 ALUM BUDS, REAR TIRES 11R24.5 ALUM BUDS-60, AIR RIDE SUSPENSION, C/W	
			CUSTOM VAC UNIT, TC407/412, 10822 L CAP, ROBUSCHI BLOWER, HOTSY, WATER BLAST WB9460	
			HEATER, HYDRAULIC BOOM, S/N 28841006	
HU003	2007	EREIGHTI INER	FLUID HEATER TRUCK W/ 48" SLEEPER, TRI AXLE, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	GP
.,		, , , = , , , , = , , = , ,	TRANSMISSION, FRONT TIRES 425/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	_,
			11R24.5-ALUM BUDS, 52,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W HAMMS TANK	
			S/N 46ANM063M61321828, TC406-MS-03-EL, 3 COMPARTMENT - 1,350-LITRE/1,350-LITRE/ 7,655-	
	1		LITRE CAPACITY, CHANDLER MFG DIESEL FIRE HEATER S/N 4810, 12 MILLION BTU, FAN EQUIPMENT	
			· · · · · · · · · · · · · · · · · · ·	
			S/N 06-256B, SIZE 1-35, 3430 RPM (16,440-KILOMETERS INDICATED, 8750-HOURS INDICATED)	
NTRO06	2007	WESTERN STAR	T/A TRACTOR Ŵ/ 44" SLEEPER, DETROIT 60 SERIES 14L 515-HP ENGINE, EATON FULLER 18-SPEED	GP
		1112721111	TRANSMISSION, FRONT TIRES 385/65R22.5-ALUM BUDS, 14,600-LB FRONT AXLE, REAR TIRES	•
			11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, EXTENDED CHASSIS,	
			C/W WET KIT, SLIDING 5TH WHEEL, PTO, HYDRAULIC PUMP.	
PS100	2007	FREIGHTLINER	PRESSURE TRUCK, S/A, MERCEDES BENZ ENGINE, ALLISON AUTOMATIC TRANSMISSION, 12,000-LB	ED
10100	2007	THE CONTENTENT	FRONT AXLE, 11R22.5 FRONT TIRES, ALUM BUDS, 21,000-LB REAR AXLE, 11R22.5 REAR TIRES, ALUM	
			BUDS, BUSINESS CLASS M2, AIR RIDE SUSPENSION, (LIC G50660), C/W CUSTOM VAC SERVICES	
			PRESSURE TANK UNIT S/N 29811206, 10,329-LITRE CAPACITY, TC407 TANK, 3 COMPARTMENT,	
			TRIPLEX PRESSURE PUMP, HOSES, REEL, WATER PRODUCT PUMP [NO BATTERY]	
PS102	2007	EREIGHTLINER	PRESSURE TRUCK, S/A EXTENDED CAB, MERCEDES BENZ ENGINE, AUTOMATIC TRANSMISSION,	ED
. 5102	. 2007	THE OTHER	FRONT TIRES 11R22.5-ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 11R22.5-ALUM BUDS,	
		i	21,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES STEEL TANK	
			· · · · · · · · · · · · · · · · · · ·	
			S/N 29820107, TC407, 3 - COMPARTMENT 10,329-LITRE CAPACITY, 3" BOWIE FLUID PUMP, APLEX TRIPLEX PLUNGER PUMP MODEL SO 15H S/N 4879	
PS103	2007	FREIGHTLINER	PRESSURE TRUCK, 4X4	GP
13103	2007	INCIGINENCI	TRESSORE INVERT, 474	GI.
PT110	2007	INTERNATIONA	S/A PRESSURE TEST TRUCK W/ EXTENDED CAB, DT 466 ENGINE, EATON FULLER 10-SPEED	GP
11110	2007	L	TRANSMISSION, FRONT TIRES 11R22.5-ALUM BUD, 12,000-LB FRONT AXLE, REAR TIRES 11R22.5-	Gi
		"	l ' '	
			ALUM BUDS, 20,000-LB REAR AXLE, AIR-RIDE SUSPENSION, C/W INTERCONTINENTAL 20' VAN BODY	
			S/N 7261 BC, GARDNER DENVER T300M PUMP, TC SPEC 406, S/N 7238, 4-CUSTOM VAC TANKS, 426-	
PT111	2007	INTERNATIONA	LITRE S/N'S 2978A 2978B 2978C & 2978D S/A PRESSURE TEST TRUCK, EXT CAB, DT 466 ENGINE, 6-SPEED TRANSMISSION, FRONT TIRES	ED
LITTT	2007			ED
		L	11R22.5-ALUM BUDS-, 12,000-LB AXLE, REAR TIRES 11R22.5-ALUM BUDS, 21,000-LB AXLE, AIR RIDE	
			SUSPENSION, C/W ITB 20' VAN BODY, S/N 7262 BC, GARDNER DENVER T300M PUMP, S/N 7241, 4-	
	<u> </u>	1	TANKS	

Equip. #	Year	Make	Description	i-doption
PT113	2007	INTERNATIONA	S/A PRESSURE TEST TRUCK, EXT CAB, DT 446 ENGINE, EATON FULLER 6-SPEED TRANSMISSION,	GP
		L	12,000-LB FRONT AXLE, 11R22.5 FRONT TIRES, ALUM BUDS, 21,000-LB REAR AXLE, 11R22.5 REAR	
-			TIRES, ALUM BUDS, INTEGRAL SLEEPER, C/W INTERCONTINENTAL TRUCK BODY S/N 7264BO,	
			OPERATOR CONTROL ROOM, GARDNER DENVER TRIPLEX PUMP [REPORTED AS OUT OF SERVICE]	
			And the base of the second of	
ST210	2007		4X4 S/A EXTENDED CAB SERVICE TRUCK, DT 466 ENGINE, 6-SPEED TRANSMISSION, FRONT TIRES	GP
		1 1	11R22.5-ALUM BUDS, REAR TIRES 11R22.5-ALUM BUDS, C/W ALI ARC GUARD, 16'X8' SERVICE BODY	
1175044	2007		W/ JABCO MODEL 1817 CRANE S/N 12679	GP
UT311	2007	1	TRIDEM 50' SCISSORNECK LOW BOY TRAILER, 30,600-LB AXLES, TIRES 275/70R22.5, ALUM BUDS, AIR	Gr
	•		RIDE SUSPENSION, (LIC. W531-13), C/W AIR CENTER LIVE ROLL, HYDRAULIC REAR LIVE ROLL,	
CB907	2007	WESTERN STAR	FOLDING SIDE MARKERS TRI-DRIVE COMBO TRUCK, 32" SLEEPER, DETROIT SERIES 60 515-HP ENGINE, 18-SPEED, FRONT TIRES	CNRL
CDSU/	2007		445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 23000-LB X 3 AXLES,	HORIZON
			AIRLINER AIR RIDE SUSP, S/N 5KKPALCK57PX81057, C/W CARGO TANK MFG 2-COMPARTMENT	PROJECT
			TANK, FRONT 1,216-GALLON, REAR 2,978-GALLON, TC SPEC 407/412, S/N V22326C, HIBON VTB-820	ritoreci
			RIOWER HOTSY WR 9460 HEATER	
H728	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATONG FULLER 18-SPEED, FRONT	ED
11720	2007	! :	TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK, S/N	,
			2724076. 10.822-LITRE	
SK010	2007	FREIGHTLINER		ED
H737	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 14L 515-HP ENGINE, EATON FULLER 18	ED
		1	SPD,1 09,468-KILOMETERS INDICATED, 13,812-HOURS INDICATED, FRONT AXLE 445/65R22.5 ALUM	
			BUDS 20,000-LB, REAR AXLES 11R/24.5 ALUM BUDS 60,000-LB, C/W CUSTOM VAC UNIT, 10,822 L	
			CAPACITY S/N 27250706, HOTSY WATER HEATER, ROBUSCHI BLOWER, HYDRAULIC BOOM	
	2007	MATERIA CTAR	TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT DIESEL 60 SERIES 475-HP ENGINE, EATON FULLER 18-	ED
H747	2007		SPEED, 20,400-LB FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES,	LD
			REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC STEEL TANK S/N 27940906, 10,822 L,	
			TC407/412, ROBUSCHI BLOWER MODEL RB-DV-145 SP/UN, S/N 06-30185, WATERBLAST HEATER,	
			HYDRAUTIC BOOM	
H750	2007	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES DIESEL ENGINE, EATON FULLER 18-SPEED,	CNRL
			FRONT AXLE 445/65R22.5 ALUM BUDS 20,000-LB, REAR AXLES 11R24.5 ALUM BUDS-20 61,200-LB,	
			AIR RIDE SUSPENSION, C/W CUSTOM VAC HYDRO-VAC UNIT, TC407/412, 10,822 L CAP, S/N	
			28951066, HIBON BLOWER, HOTSY WATER BLAST WB9460 HEATER, HYDRAULIC BOOM	
P271	2007	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, DIESEL ENGINE, AUTO, TIRES 265/75R16, C/W LEATHER, SUNROOF,	ED
			BUCKETS W/ CONSOLE, WOOD BOX LINER, RUNNING BOARDS	
P277	2007	CHEVROLET	PICKUP TRUCK, 3/4-TON, CREW CAB, MANUAL TRANSMISSION, 255,436-KILOMETERS INDICATED,	ED
			2,599 ENGINE HOURS, 4WD, SUNROOF, OIL/GAS TANK W/ PUMP, DURAMAX DIESEL ENGINE	
V558	2007	WESTERNISTAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	ED
V558	2007	WESTERNSTAR	445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W PROGRESS TANK, S/N V22066-C,	נט
			, , , , , , , , , , , , , , , , , , , ,	
NP011	2007	WESTERN STAR	15,906-LITRE TRI-DRIVE NITROGEN PUMP UNIT, SLEEPER/CONTROL CAB COMBO, CAT ENGINE, EATON FULLER 18-	ED
MLOTT	2007	MEDIEWKAIWK	SPEED, FRONT TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, AIR RIDE SUSPENSION, C/W	
			WELLQUIP NITROGEN PUMP UNIT, TANK S/N 30160, 11,234 LITRE	
TR014	2007	WESTERN STAR	T/A TRACTOR W/ 46" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ED
111027	2007	1.7.501.5111.01741	FRONT TIRES 11R24.5-ALUM BUDS, 14,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS,	
			46,000-LB REAR AXLES, NEWAY AIR RIDE SUSPENSION, S/N 5KIJALAV17PY07157, C/W WET KIT,	
			SLIDING 5TH WHEEL	
H735	2007	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 14L 515-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR	
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, S/N 5KKPALCK87P81053, C/W CUSTOM VAC SERVICES	
		1		
			HYDRO-VAC UNIT S/N 27730706, HOTSY WATER BLAST, 10,822-LITRE CAPACITY, TC 407/412 TANK,	

Equip. # Year Description H744 2007 WESTERN STAR TRI-AXLE HYDRO-VACUUM TRUCK, SLEEPER, DETROIT 60 SERIES ENGINE, EATON FULLER 18-SPEED ED TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR AXLES, 11R24.5 REAR TIRES; ALUM BUDS, AIR RIDE SUSPENSION, (LIC G45174), C/W CUSTOM VAC SERVICES UNIT S/N 27920906, ROBUSHI BLOWER TC 407/412 TANK, 10,822-LITRE CAPACITY, HOTSY STEAMER & WATER BLAST SYSTEM, HYDRAULIC BOOM WESTERN STAR TRI-AXLE NITROGEN PUMPER TRUCK, SLEEPER/CONTROL CAB COMBO, CAT C15 ACERT 550-HP 2007 NP010 ΕD ENGINE, 18-SPEED TRANSMISSION, 192,152-KILOMETERS INDICATED, 10,692-HOURS INDICATED, FRONT TIRES 445/65R22.5-ALUMINUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-, CRYOGENIC VESSEL ALTERNATIVES TANK S/N 30158 (2008) W/ 2,964-GALLON CAPACITY NTR005 2007 WESTERN STAR TANDEM-AXLE HIGHWAY TRACTOR W/ SLEEPER, CAT C15.475-HP ENGINE, EATON FULLER ED TRANSMISSION, FRONT TIRES 425/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, AIR LINER AIR RIDE SUSPENSION, (LIC G19616), C/W ALUM TOOL BOX, 298,851-KILOMETERS INDICATED, 9.174-HOURS INDICATED WESTERN STAR T/A HIGHWAY TRACTOR, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION,16,000-NTR007 2007 ED LB FRONT AXLE, 385/65R22.5 FRONT TIRES, ALUM BUDS, 46,000-LB REAR AXLES, 11R24.5 REAR TIRES, ALUM BUDS, SLEEPER, (LIC G19649), 232,131-KILOMETERS INDICATED, 7,073-HOURS INDICATED NTR008 2007 WESTERN STAR T/A HIGHWAY TRACTOR, CAT C15 475-HP, EATON FULLER 18-SPEED, 385/65R22.5 FRONT, 11R24.5 ED REAR, ALUM BUDS, AIR RIDE วกกว WESTERN STAR TRUCK TRACTOR, CATERPILLAR C15 475-HP, EATON FULLER 18-SPEED TRANSMISSION, 24" SLEEPER NTR010 FD CAB, 388,338-KILOMETERS INDICATED, 11,082-HOURS INDICATED CB910 2007 WESTERN STAR TRI-DRIVE COMBO TRUCK W/ 26" SLEEPER, CAT C15 550-HP ENGINE, EATON FULLER 18-SPEED ĠΡ TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W PROGRESS TANK S/N V223660, TC407/412, 2 COMPARTMENT - 8,002-LITRE/11,185-LITRE CAPACITY, HIBON BLOWER, HOTSY WATER BLAST CB914 2007 WESTERN STAR TRI-DRIVE COMBO VACUUM TRUCK W/ 36" SLEEPER, CAT C15 550-HP ENGINE, EATON FULLER 18-GP SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5 ALUM BUDS, 69,000-LB AXLES, AIRLINER AIR RIDE SUSPENSION, C/W PROGRESS 2 COMPARTMENT TANK S/N V224860, DOT407/412, 11.227-LITRE CAPACITY, HIBON VTB 820 BLOWER, WATER BLAST HEATER FREIGHTLINER PRESSURE TRUCK, S/A, 4X4, EXTENDED CAB, MERCEDES BENZ ENGINE, AUTOMATIC TRANSMISSION, PS101 2007 RL FRONT TIRES 385/65R22.5-ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 42X25.00-20, 21,000-LB REAR AXLES, SPRING SUSPENSION, C/W CUSTOM VAC SERVICES STEEL TANK S/N 29830107, TC407, 3 COMPARTMENT 10,329-LITRE CAPACITY, GENERAL IND. PUMP, TRIPLEX PUMP, 2" T&E FLUID PUMP INO BATTERYL INTERNATIONA S/A PRESSURE TEST TRUCK W/ EXTENDED CAB, DT466 ENGINE, EATON FULLER 6-SPEED PT112 2007 RL TRANSMISSION, FRONT TIRES 11R22.5-ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 11R22.5-ALUM BUDS, 21,000-LB REAR AXLE, (LIC G20941), C/W INTERCONTINENTAL 20' VAN BODY, GARDNER DENVER TRIPLEX PUMP, 4 TANKS, OPERATOR CONTROL ROOM H748 2007 WESTERN STAR TRI-DRIVE HYDRO VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 20,000-LB ED FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G49531), C/W CUSTOM VAC S/N 28941006, TCRN Z.05.437.05.04, TC 407/412, 10,822-LITRE, 309,139-KILOMETERS INDICATED, 13,451-HOURS INDICATED H739 2007 WESTERN STAR TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 20,000-LB ZC FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 62,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC HYDRO-VAC UNIT, TC407/412. TCRN Z-05-437-05-04, 10,822 L, S/N 25380706, ROBUSCHI BLOWER, HOTSY, WATER BLAST, HYDRAULIC BOOM WESTERN STAR TRI-DRIVE HYDRO-VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED, FRONT TIRES H752 2007 ED 445/65R22.5, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G-50654), C/W CUSTOM VAC UNIT, S/N 28981106, TC407/412, 10,822 L, HIBON BLOWER, HOTSY WATER BLAST WB9460 HEATER, BOOM 2007 STEPHENS QUAD WATER TRAILER, 34,000-LITRE CAPACITY, 2 COMPARTMENT, ALUMINUM 0006 ZAMA CITY



Equip.#	T	Make		Locatio
TR012	2007	1	T/A TRACTOR W/ 39" SLEEPER, CAT C15 475HP ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	MEDICIN
			11R24.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, AIRLINER AIR RIDE SUSPENSION, C/W WET	HAT
			KIT, MAGNUM ALUM BUMPER	
H741	2007	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, FULLER 18-SPEED TRANS, CUSTOM VAC UNIT S/N 27280806,	SWIFT
			11,915L CAP	CURREN
H751 20	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, CAT ENGINE, 8-SPEED, FRONT TIRES 445/65R22.5, ALUM BUDS,	SWIFT
			REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK S/N 28961106, TC407/412, 10,822-LITRE	CURREN
			CAPACITY	
H753	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT	SWIFT
			TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC UNIT, S/N	CURREN
			28971106, 10,822 L, 192,225-KILOMETERS INDICATED, 13,057-HOURS INDICATED	
HU002	2007	FREIGHTLINER	FLUID HEATER UNIT W/ SLEEPER, TRI AXLE, CAT C15 ENGINE, EATON FULLER 18-SPEED	SWIFT
			TRANSMISSION, FRONT TIRES 425/65R22.5-ALUM BUDS, 16,000-LB AXLE, REAR TIRES 11R24.5-ALUM	CURREN
		1	BUDS, 52,000-LB AXLES, (LIC H-26759) C/W 12 MILLION BTU BURNER, 3 COMPARTMENT TANK 1,300-	
			LITRE, 4,900-LITRE, AND 3,800-LITRE CAPACITY, 141,898-KILOMETERS INDICATED, 1,448-HOURS	
			INDICATED	
TRO13	2007	WESTERN STAR	T/A TRUCK TRACTOR W/ 48" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	SWIFT
			TRANSMISSION, FRONT TIRES 11R24.5-ALUM BUDS, 14,000-LB FRONT AXLE, REAR TIRES 11R24.5-	CURREN
			ALUM BUDS, 46,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W PTO DRIVE, 3" FLUID	COMMEN
			PUMP. 357.747-KILOMETERS INDICATED. 17.087-HOURS INDICATED	
CT2062	2007	KENWORTH	TANDEM-AXLE STEER / TRI-AXLE COILED TUBING UNIT, CAT C15 ENGINE, EATON FULLER 18-SPEED	ED
C12002	2007	REINWORTH	TRANSMISSION, 425/65R22.5 FRONT TIRES, ALUM BUDS, 11R24.5 REAR TIRES, ALUM BUDS, NEWAY	
			, ,	
			AIR RIDE SUSPENSION, C/W C-TECH COILED TUBING UNIT, HYDRA RIG RT25 INJECTOR, 60"	
			GOOSENECK, SPICER 792, ALLIANCE ALUM CONTROL CABIN W/HYDRAULIC LIFT, FASSI 7-TON PICKER	
CDD12	2007	MUCCTERM CTAR	E150AC24_SIDE_OUTRIGGERS_S/N 5159_ COMBINATION UNIT, TRI-DRIVE HYDRO-VACUUM TRUCK, W/ WASH SYSTEM, DETROIT SERIES 60	MEDICI
CB913	2007			
			515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, AIR RIDE SUSPENSION, SLEEPER, W// 10-	HAT
11740	2007	MICCEPALCEAR	CU. METER DEBRIS, AND 4-CU. METER WATER SECTIONS	MEDICI
H749	2007		TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT	
	ŀ		TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, C/W	HAT
			CUSTOM VAC UNIT, TC407/412, 10,822 L, ROBUSHI BLOWER, HOTSY, WATER BLAST WB9460	
	200=	14/555551 5745	HEATER, HYDRAULIC BOOM, S/N 28931006	
H743	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	SC
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	
			BUDS, 60,000-LB AXLES, AIR RIDE SUSPENSION, 36" SLEEPER, C/W CUSTOM VAC SERVICES, 10,822-	
, , .			LITRE	
H725	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DDC 60 515-HP ENGINE, EATON FULLER 18-SPEED	SC
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
		-	SERVICES STEEL TANK S/N 25360606, TC407/412, 2 COMPARTMENT - 10,822-LITRE CAPACITY,	
			ROBUSCHUBLOWER WATERBLAST HEATER	
HU004	2007	GMC	HEATER TRUCK, 4X4, S/A, 6.6 DURAMAX DIESEL ENGINE, 6-SPEED ALLISON TRANSMISSION, FRONT	GP
			TIRES 245/70R19.5, 8,000-LB AXLE, REAR TIRES 245/70R19.5, 13,000-LB AXLE, (LIC H-26761) C/W 3	
			TANKS, 568-LITRE CAPACITY , 568-LITRE CAPACITY AND 1,200-LITRE CAPACITY, 3,500,000-BTU	
			BURNER	
UT107	2007	EDMONTON	TRIDEM SCISSORNECK TRAILER, TIRES 275/70R22.5-ALUM BUDS, AIR RIDE SUSPENSION, C/W 10'X35'	GP
		TRAILER	WORKING DECK, 40-TON, CENTER LIVE ROLL, FLIPOVER REAR LIVE ROLL, BEAVERTAILS	
H745	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENG!NE, EATON FULLER 18-SPEED	
			TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 69,000-LB REAR	
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
			SERVICES VAC UNIT S/N 28921006, 10,822-LITRE CAPACITY, TC407/412 TANK, ROBUSCHI RB DV	
		I .	145/SP UN BLOWER S/N 06-30190, HOTSY WB9460 WATER BLAST HEATER S/N 60430806, BOOM	
P278	2007	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, DURAMAX DIESEL ENGINE, PUMP TANK	
· · ·	•••		. , , ,	
ED014	2008	NORTH	TRIDEM END DUMP TRAILER, TIRES 11R24.5, STEEL BUDS, C/W ROLL UP TARP, SEALED END GATE,	ED

				le estion
Section of the Control of the Contro			Description	
Q009	2008	STEPHENS	QUAD WATER TRAILER, 34,000-LITRE CAPACITY, 2 COMPARTMENT, ALUMINUM	SWIFT CURRENT
VTR002	2009	WESTERN STAR	COMPLETE SEMI VAC UNIT, TANDEM AXLE TRACTOR, SLEEPER CAB, FULLER 18-SPEED	GP
	1		TRANSMISSION, 14,230-LB FRONTS, 46,000-LB REARS, CUSTOM VAC TRI-AXLE VAC TRAILER, S/N	
	ļ		36290807, 24,991-LITRE CAPACITY	
HO014	2009	WESTERN STAR	TRI-DRIVE HOT OILER TRUCK, SLEEPER, ALLISON AUTO TRANS, 385/70R22.5 FRONT, ALUM BUDS,	GP
			11R24.5 REAR, ALUM BUDS, C/W 2008 WELLQUIP HOT OILER UNIT, 2008 FABMASTER 5 MMBTU OIL	
			BURNER S/N 709, 2008 HAMMS TC406-MS-03-EL STEEL TANK 3-COMPARTMENT (28/24/8 BBL) S/N	
			46ANM063M81444012, 2006 SPM MODEL TWS600S TRIPLEX PUMP, S/N F1006-3885, OPERATORS	
			CONTRO! PANEL HOSE REFL	
SU014	2009	WESTERN STAR	SOUR SEAL TRUCK - TC407 SEALED UNIT, W/ SCRUBBER - TANK S/N 4899-09-09	ZC
SVT002	2009	N/A	COMPLETE SEMI-VAC UNIT	SWIFT
				CURRENT
P309	2011	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	GP
		<u> </u>		
P316	2011	DODGE	PICKUP TRUCK	ED

P285	2011	DODGE	PICKUP TRUCK, CREW CAB, HEMI 5.75 HEMI ENGINE, 4WD	ED
P308	2011	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	ED
B04.0	2044	- FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	DI.
P312	2011	FÖRD	PICKUP TRUCK, CREW CAB, 4WD, TURBU DIESEL ENGINE	RL
naec	2011	DODGE	PICKUP TRUCK, CREW CAB, LONG BED, 5.7L HEMI ENGINE, 4WD, OIL/FUEL TANK W/ PUMP, 144,425-	SWIFT
P286	2011	DODGE		CURRENT
D21.4	2011	FORD	KILOMETERS INDICATED PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	ZAMA CITY
P314	2011	FORD	PICKOP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	ZAMIA CITT
P313	2011	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	
1313	2011	1 10112	Trooty Chen Gray Troy Total Distriction	
P288	2012	DODGE	PICKUP TRUCK, CREW CAB, SHORT BED, S.7L HEMI ENGINE, 4WD	ED
			, , , , , , , , , , , , , , , , , , , ,	
P311	2012	FORD	PICKUP TRUCK, SUPER DUTY, 4-DOOR, MEGA CAB, LONG BED, C/W FUEL TANK / TOOL BOX, W/ 12-	ED
			VOLT PUMP	
P287	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	GP
P293	2012	DODGE	PICKUP TRUCK, CREW CAB, CUMMINS DIESEL ENGINE, 4WD, 641-KILOMETERS INDICATED	GP
P296	2012	DODGE	PICKUP TRUCK, CREW CAB, CUMMINS DIESEL ENGINE, 4WD	GP
P315	2012	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	, ZC
P302	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	CNRL
			· *	HORIZON
D204	2012	PODGE	DICKLID TOLICK CREW CAR, AWD	PROJECT ED
P304	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	50
P295	2012	DODGE	PICKUP TRUCK, CREW CAB, CUMMINS DIESEL ENGINE, 4WD, FUEL/GAS TANK W/ PUMP, 931-	MEDICINE
'233		2000	KILOMETERS INDICATED	HAT
NP013	2012	N/A	NITROGEN PUMPER [ASSUMED TRUCK MOUNTED]	ED
P289	2012	DODGE	PICKUP TRUCK, CREW CAB, HEMI 5.7L HEMI ENGINE, 4WD, 2,508-KILOMETERS INDICATED	ED
P292	2012	DODGE	PICKUP TRUCK, CREW CAB, HEMI 5.71 HEMI ENGINE, 4WD, 2,159-KILOMETERS INDICATED	ED
P298	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	ED

Equip: i	Year	Make	Description	Location
P299	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	ED
P300	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	CNRL.
P301	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	ED
P303	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	CNRL
P306	2012	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	ED
P307	2012	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	ED
P310	2012	FORD	PICKUP TRUCK, SUPER DUTY, 4-DOOR, MEGA CAB, LONG BED, C/W FUEL TANK / TOOL BOX, W/ 12- VOLT PUMP	ED
NTT001	2012	MICRON	NITROGEN TUBE TRAILER	. ED
P290	2012	DODGE	PICKUP TRUCK, CREW CAB, SHORT BED, 5.7L HEMI ENGINE, 135,546-KILOMETERS INDICATED	SWIFT
P305	2012	DODGE	PICKUP TRUCK, SHORT BED, CREW CAB, 4WD, 114,507- KILOMETERS INDICATED	CURRENT SWIFT
TR016	2012	KENWORTH	TRUCK TRACTOR, TANDEM AXLE	SWIFT
P297	2012	DODGE	PICKUP TRUCK, LONG BED EXTENSION, CREW CAB, 4WD	CURRENT ED
HU007	2012	OVERLAND	FRAC HEATER UNIT, 35-MM	GP
P291	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	
P294	2012	DODGE	PICKUP TRUCK, CREW CAB, CUMMINS DIESEL ENGINE, 4WD, 532-KILOMETERS INDICATED	
NTR011	2013	WESTERN STAR	TRUCK TRACTOR, TRI-AXLE	GP
H758	2013	WESTERN STAR	HYDRO-VAC, TRI-AXLE, W/ TORNADO	ED
H757	2013	INTERNATIONA L	HYDRO-VACUUM TRUCK, TRI-AXLE, W/ OPERATORS COMPARTMENT, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5, REAR 11R24.5, INTERNATIONAL ENGINE, HOTSY STEAMER, W/ WATER BLAST SYSTEM, HYDRAULIC BOOM	SC .
H756	2013	INTERNATIONA L	HYDRO-VACUUM TRUCK, TRI-AXLE, EATON FULLER 18-SPEED TRANSMISSION, HOTSY STEAMER, W/ WATER BLAST SYSYTEM, HYDRAULIC BOOM, 65,695-KILOMETERS INDICATED, 3,288-HOURS INDICATED	SWIFT CURRENT
NP013	2013	N/A	N2 PUMPER HT RECOVERY	ED
H760	2014	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, DETAILS NOT PROVIDED	CNRL
NP015	2014	N/A	NITROGEN PUMPER TRAILER	ED
P326	2014	DODGE	PICKUP TRUCK, CREW CAB, 4WD	GP
P324	2014	DODGE	PICKUP TRUCK, CREW CAB, 4WD	ED
P327	2014	DODGE	PICKUP TRUCK, CREW CAB, 4WD	ED
H759	2014	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, DETAILS NOT PROVIDED	SWIFT CURRENT
H761	2014	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, DETAILS NOT PROVIDED	SWIFT



Big Eagle Limited Partnership Asset List* GA Global Partners Report

Fquip #	Year			Location
P323	2014	DODGE	PICKUP TRUCK, CREW CAB, 4WD	GP
P325	2014	DODGE	PICKUP TRUCK, CREW CAB, 4WD	MH
PTI109	2014	KENWORTH	PRESSURE TRUCK, 600-HP, 6X4, WITH TRIPLEX PUMP, VESSEL, OPERATORS CAB [NEW IRON ADDED IN 2015]	GP
H762	2014	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCĶ	
H763	2014	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, DETAILS NOT PROVIDED	
RN089		N/A	TRAILER, CREW SHACK	ED
RN093		N/A	TRAILER, CREW SHACK	ED
RN097		N/A	TRAILER, CREW SHACK	ED
RN100		N/A	TRAILER, CREW SHACK	ED
RN108		N/A	TRAILER, CREW SHACK	ED
RN110		N/A	TRAILER, CREW SHACK	ED
RN112		N/A	TRAILER, CREW SHACK	ED
UT359		TRAILMOBILE	DRY VAN TRAILER	GP
PSF108		N/A	SKID UNIT, PRESSURE TEST UNIT, W/ PUMP AND STORAGE SYSTEMS	ED
MV046	+	N/A	TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 2-BEDROOMS]	ED
RN081		N/A	TRAILER, CREW SHACK	ED
RN082		N/A	TRAILER, CREW SHACK	, ED
XW073		MACK	TRUCK TRACTOR, TANDEM AXLE [INSUFFICIENT INFORMATION]	ED
UT360	 	MOND	STORAGE TRAILER, 53'	ED
UT361		+	STORAGE TRAILER, 53'	ED
CT2071-2		+	SPOOLER, W/ GEAR DRIVE	ED
CSC003		N/A	TUBING SPOOLER	ED
UT358		FRUEHAUF	DRY VAN TRAILER	ED
RN035		N/A	PROPANE TANKS, ESTIMATED 50-GALLON	ED
RN051		N/A	TRAILER, CREW SHACK	ED
UT009		N/A	UTILITY TRAILER, FLATDECK, ESTIMATED 20'	ED
UTC318		N/A	TRAILER, FLATDECK, 18'	
UTH050		N/A	TRAILER, FLATDECK	
MV002		MOUNTAIN VIEW	TRI-AXLE CREW SHACK TRAILER, PLATE #S377-88	
UT108		N/A	BOOSTER FOR UT107	

Additional Assets

V065	N/A	N/A	VAC TRUCK	
V068	N/A	N/A	VAC TRUCK	
V063	N/A	N/A	VAC TRUCK	



Big Eagle Limited Partnership Asset List* GA Global Partners Report

Equip: #	Year	Make	Description Location
SSK004	N/A	N/A	SKID STEER
S005 °	N/A	N/A	STEAM TRUCK
Q009	N/A	N/A	CUSTOM QUAS VAC
CC2071	N/A	N/A	CATCH TANK
W077	N/A	N/A	WATER TRUCK
TRP802	N/A	N/A	PICKER TRUCK

Schedule 2.18 - Bill of Sale

THIS AGREEMENT made as of the _	day of	
BETWEEN:		

DELOITTE RESTRUCTURING INC. in its capacity as Receiver and Manager of the property, assets and undertaking of Big Eagle Hydro-Vac Inc., Big Eagle Limited Partnership, Big Eagle Hydro-Vac (North) Inc., and Big Eagle (North) Limited Partnership and not in its personal capacity

hereinafter called the "Vendor"

- and --

GREAT AMERICAN GLOBAL PARTNERS, LLC

hereinafter called the "Purchaser"

RECITALS

- A. The Vendor was appointed as receiver and manager of Big Eagle Hydro-Vac Inc., Big Eagle Limited Partnership, Big Eagle Hydro-Vac (North) Inc., and Big Eagle (North) Limited Partnership (the "Debtor") pursuant to an order of the Court made on June 8, 2015 (the "Order").
- B. The Vendor and the Purchaser have entered into an auction services agreement dated as of June 24, 2015 whereby the Purchaser was authorized to sell the assets of the Debtor as agent for the Vendor (the "Auction Services Agreement"). All terms not otherwise defined herein have the meanings given to them in the Auction Services Agreement.
- C. The delivery of this bill of sale is required upon the Vendor exercising its option to transfer title to any unsold assets of the Debtor to the Purchaser pursuant to Section 2.18 of the Auction Services Agreement.
- D. On [], the Authorization and Approval Order (as such term is defined in the Auction Sales Agreement) was granted by the Court.
- E. The Vendor has agreed to sell to the Purchaser all of the Debtor right, title and interest, if any, in and to the assets of the Debtor more particularly described in Schedule "A" hereto (the "Assets") in accordance with Section 2.18 of the Auction Services Agreement.

NOW THEREFORE THIS AGREEMENT WITNESSETH that in consideration of the premises and mutual covenants and agreements contained herein, the sum of Ten Dollars (\$10.00) now paid to the Vendor and other good and valuable consideration, the receipt and sufficiency of which each party hereby acknowledges, the parties do covenant and agree as follows:

1.1 SALE OF ASSETS

- (1) The Assets are being sold on an "as is, where is basis", at the Purchaser's risk and peril. The Purchaser has had an opportunity to inspect the Assets and to investigate the title thereto and the right of the Vendor to sell same. The Purchaser has conducted such inspections and investigations concerning those matters as the Purchaser decided were appropriate, and has satisfied itself concerning those matters and concerning the existence of and the condition of the Assets. No representation, warranty or condition either express or implied has been or will be given by the Vendor as to the existence, description, condition, cost, size, quality, fitness for purpose or merchantability of the Assets. Without limiting the foregoing, any and all representations, conditions or warranties express or implied pursuant to the Sale of Goods Act of Alberta and similar legislation do not apply hereto and have been waived by the Purchaser. The description of the Assets contained in Schedule "A" annexed hereto is for the purposes of identification only and no representation, condition or warranty has or will be given by the Vendor concerning the accuracy of that description.
- (2) The Vendor hereby sells, assigns and transfers to the Purchaser, pursuant to the power of sale contained in the Order and the Authorization and Approval Order, all of the Debtor's right, title and interest, if any, in and to the Assets, to hold such right, title and interest of the Debtor in and to the Assets to and to the uses of the Purchaser.
- (3) The Vendor hereby remises, releases and forever discharges to the Purchaser all of its claims and demands upon the Assets.
- (4) The Purchaser hereby covenants to pay any Sales Taxes, and agrees to indemnify and save the Vendor harmless from any claims or demands arising from or connected to any failure to pay such Sales Taxes.
- (5) The Purchaser agrees to take possession of the Assets at their existing locations (the "Premises") and to remove the Assets from the Premises by the close of business on []. The Purchaser further agrees to indemnify the Vendor with respect to any damages caused to the Premises upon the removal of the Assets by the Purchaser, its employees or agents. The Purchaser agrees that the Vendor has no obligation to deliver the Assets to the Purchaser.

2.0 MISCELLANEOUS

- (1) This indenture and all of its provisions shall enure to the benefit of and be binding upon the parties and their respective successors and assigns.
- (2) This indenture shall be governed by and construed in accordance with the laws of the Province of Alberta and the laws of Canada applicable therein.

(3) This Agreement may be executed in counterparts, each of which when so executed shall be an original but all such counterparts shall together constitute one and the same instrument.

IN WITNESS WHEREOF the parties hereto have executed this agreement.

The parties have executed this Agreement.

undertaking of the Debtor and not in its personal capacity	
Ву:	
Name:	
Title:	
GREAT AMERICAN GLOBAL PARTNERS, LLC By:	
Name:	
Title:	

DELOITTE RESTRUCTURING INC., in its capacity as Receiver of the property, assets and

SCHEDULE B

Auction and Asset Purchase Solicitation Procedures

AUCTION AND ASSET PURCHASE SOLICITATION PROCEDURES

BIG EAGLE HYDRO-VAC INC., BIG EAGLE LIMITED PARTNERSHIP, BIG EAGLE HYDRO-VAC (NORTH) INC., AND BIG EAGLE (NORTH) LIMITED PARTNERSHIP

RECITALS

- A. Pursuant to an order of the Honourable Madam Justice Strekaf sitting as a Justice of the Court of Queen's Bench of Alberta (the "Court") dated June 8, 2015 (the "Receivership Order"), Deloitte Restructuring Inc. (the "Receiver") was appointed as receiver and receiver and manager of all of the assets, undertaking and property of Big Eagle Hydro-Vac Inc., Big Eagle Limited Partnership, Big Eagle Hydro-Vac (North) Inc., and Big Eagle (North) Limited Partnership (collectively, the "Debtor") pursuant to section 243(1) of the Bankruptcy and Insolvency Act and section 13(2) of the Judicature Act (Alberta) under Court File No. 1501-06320.
- B. On June 29, 2015, the Court granted an order (the "AASP Approval Order") approving an auction and asset purchase solicitation process (the "AASP") and the AASP procedures set forth herein (these "AASP Procedures").
- C. All dollar amounts expressed herein, unless otherwise noted, are in Canadian currency. Unless otherwise indicated herein any event that occurs on a day that is not a **Business Day** shall be deemed to occur on the next Business Day.

ARTICLE 1 - DEFINED TERMS

In these AASP Procedures:

- (1) Approval Hearing has the meaning ascribed to it in Section 5.8(1);
- (2) Approval Order has the meaning ascribed to it in Section 5.8(1);
- (3) Asset Purchase Bid has the meaning ascribed to it in Section 2.1;
- (4) Assets means the right, title and interest of the Debtor and the Receiver in and to (i) the vehicles, rolling stock, machinery and equipment located at the Debtor's Premises or used by the Debtor in carrying on its Business listed on Schedule "A", (ii) the office equipment and office furniture located at the Debtor's Premises, and (iii) all trademarks, patents, industrial designs, trade names, domain names and other intellectual property, but, for greater certainty, does not include the Excluded Assets;
- (5) Auction has the meaning ascribed to it in Section 5.6;

- (6) Auction Bid has the meaning ascribed to it in Section 2.1;
- (7) Auction Date means the date on which the Auction takes place and shall be no later than forty-five (45) days after the granting of the Approval Order or such later date as may be agreed to in writing by the Receiver;
- (8) AASP has the meaning ascribed to it in Recital B;
- (9) AASP Approval Order has the meaning ascribed to it in Recital B;
- (10) AASP Procedures has the meaning ascribed to it in Recital B;
- (11) **Backup Bid** has the meaning ascribed to it in Section 5.7;
- (12) Backup Bid Expiration Date has the meaning ascribed to it in Section 5.7;
- (13) Backup Bidder has the meaning ascribed to it in Section 5.7;
- (14) Break-Up Fee means the amount of one percent (1%) of the Net Minimum Guaranteed Amount;
- (15) Business Day means a day on which banks are open for business in the City of Calgary but does not include a Saturday, Sunday or statutory holiday in the Province of Alberta;
- (16) **Debtor** has the meaning ascribed to it in Recital A;
- (17) **Debtor's Premises** means the following premises:
 - (a) 5916 36 Street East, Edmonton International Airport, Leduc, Alberta;
 - (b) 9319 81 Avenue, Clairmont, Alberta; and
 - (c) 2747 North Service Road West, Swift Current, Saskatchewan;
- (18) *Court* has the meaning ascribed to it in Recital A;
- (19) Excluded Assets means (i) all of the Debtor's real property; (ii) all fixtures including, all components of the building's electrical (other than, with the consent of the Receiver acting reasonably, any electrical components installed by the Debtor and exclusively dedicated to the Assets), mechanical, plumbing, heating and air-conditioning systems, fire extinguishers and other fire safety equipment, and (iii) all motor vehicles and office equipment leased by the Debtor;
- (20) GA means Great American Global Partners, LLC;
- (21) GA Auction Agreement means the auction services agreement dated June 24, 2015 between Great American Global Partners, LLC, and the Receiver;
- (22) Known Potential Bidders has the meaning ascribed to it in Section 2.3;

- (23) Net Minimum Guaranteed Amount means \$18,700,000.00, subject to adjustment in accordance with the GA Auction Agreement;
- (24) Participating Bidder has the meaning ascribed to it in Section 3.1(2);
- (25) **Process Letter** has the meaning ascribed to it in Section 2.3;
- (26) Purchase Price has the meaning ascribed to it in Section 5.3;
- (27) Qualified Asset Purchase Bid has the meaning ascribed to it in Section 5.3;
- (28) Qualified Auction Bid has the meaning ascribed to it in Section 5.2;
- (29) Qualified Bid or Qualified Bids has the meaning ascribed to it in Section 5.4;
- (30) Qualified Bid Deadline has the meaning ascribed to it in Section 5.1;
- (31) Qualified Bidder has the meaning ascribed to it in Section 5.4;
- (32) Receiver has the meaning ascribed to it in Recital A;
- (33) Receivership Order has the meaning ascribed to it in Recital A;
- (34) Solicitation Process has the meaning ascribed to it in Section 2.1;
- (35) Successful Bid has the meaning ascribed to it in Section 5.7;
- (36) Successful Bidder has the meaning ascribed to it in Section 5.7;
- Superior Bid shall mean (a) in the case of an Auction Bid, a credible offer that (i) is a higher or better bid than set out in the GA Auction Agreement provided that the net minimum guaranteed amount in such bid exceeds the Net Minimum Guaranteed Amount by not less than the amount of the Break-Up Fee plus \$200,000 and the appraisal costs in the amount of \$50,000 payable to GA under the GA Agreement with no materially unfavourable changes having been made to the other economic terms in the GA Auction Agreement (ii) made by a recognized, reputable and credit worthy North American liquidator, (iii) otherwise incorporates all or substantially all of the terms and conditions of the GA Auction Agreement, and (iv) otherwise complies with the AASP, and (2) in the case of a superior Asset Purchase Bid, a credible offer to acquire all or substantially all of the Assets (i) that is an all cash offer the Receiver reasonably believes should realize net cash proceeds greater than the net cash proceeds reasonably expected to be realized from the transactions contemplated by the GA Auction Agreement, (ii) made by a creditworthy bidder determined by the Receiver to be qualified to participate in the process by the Receiver, (iii) is on a "as is, where is" basis with no more onerous representations or warranties by the Receiver than contained in the GA Auction Agreement and contains no material holdbacks or downward purchase price adjustments that could lead to a lower realization of net proceeds than contemplated by the GA Auction Agreement, (iv) is reasonably expected by the Receiver to be completed prior to the Termination Date;
- (38) Target Asset Purchase Closing Date means July 27, 2015; and

(39) Termination Date means 30 business days after the Auction Date.

ARTICLE 2- SOLICITATION

Section 2.1 Solicitation Process

- (1) These AASP Procedures describe the process for soliciting and selecting bids (i) for a minimum guarantee auction bid in respect of the Assets
- (2) or (ii) to purchase the Assets (an "Asset Purchase Bid") and the approval thereof by the Court (collectively, the "Solicitation Process").

Section 2.2 "As Is, Where Is"

Any Auction Bid or Asset Purchase Bid will be on an "as is, where is" basis and without (1)surviving representations, warranties, covenants or indemnities of any kind, nature, or description, including without limitation, with respect to merchantability, physical or financial condition, description, or fitness for a particular purpose, by the Debtor or the Receiver or any of their agents, estates, advisors, professionals or otherwise, except to the extent set forth in the relevant agreement with the Successful Bidder. By participating in this Solicitation Process every Participating Bidder is deemed to have acknowledged and agreed that it: (i) has relied solely upon its own independent review, investigation and/or inspection of any documents and/or the assets to be acquired and liabilities to be assumed in making its bid; (ii) did not rely upon any written or oral statements, representations, promises, warranties or guaranties whatsoever, whether express or implied (by operation of law or otherwise), regarding the assets to be acquired or liabilities to be assumed or the completeness of any information provided in connection therewith, including by the Receiver, or any of its advisors, except as expressly stated in the definitive document submitted by it; (iii) is a sophisticated party capable of making its own assessments in respect of making its bid; and (iv) has had the benefit of independent legal advice in connection with the preparation and submission of its bid in accordance with these Solicitation Procedures.

Section 2.3 Solicitation of Interest

- (1) Forthwith upon the granting of the AASP Approval Order, the Receiver will,
 - (a) prepare a list of parties that may be interested in making an Auction Bid or a Asset Purchase Bid (the "Known Potential Bidders"); and
 - (b) send a letter to the Known Potential Bidders in substantially the form attached hereto as Schedule "B" (the "Process Letter") notifying Known Potential Bidders of the existence of the Solicitation Process and inviting the Known Potential Bidders to make an Auction Bid or an Asset Purchase Bid.

ARTICLE 3- PARTICIPATION REQUIREMENT

Section 3.1 Participation Requirements

- (1) In order for an interested party to participate in the Solicitation Process, the interested party must deliver the following to the Receiver unless the Receiver waives the requirement for such information, acting reasonably:
 - (a) a specific indication of the anticipated sources of capital for such Potential Bidder and preliminary evidence of the availability of such capital, or such other form of financial disclosure and credit support or enhancement that will allow the Receiver and its legal advisors, to make, in their reasonable business or professional judgment, a reasonable determination as to the Potential Bidder's financial and other capabilities to consummate an Auction Bid or an Asset Purchase Bid;
 - (b) a letter setting forth the identity of the Potential Bidder, the contact information for such Potential Bidder, full disclosure of the direct and indirect owners of the Potential Bidder and their principals;
 - (c) an indication of whether the Potential Bidder intends to make an Auction Bid or an Asset Purchase Bid;
 - (d) a written acknowledgment of receipt of a copy of the AASP Approval Order (including these AASP Procedures) and agreeing to accept and be bound by the provisions contained therein.
- (2) If the Receiver is satisfied with the reputation, creditworthiness and ability of the interested party to complete a transaction prior to the earlier Termination Date or Target Asset Purchase Closing Date, as applicable, then the interested party will be deemed to be a "Participating Bidder" and the Receiver will notify the interested party that they have qualified as a Participating Bidder.

ARTICLE 4- DUE DILIGENCE AND SITE VISITS

Section 4.1 Due Diligence

(1) The Receiver will provide Participating Bidders with information concerning the Assets and arrange for site visits at the **Debtor's Premises** and the opportunity to for the inspection of the Assets at times acceptable to the Receiver.

ARTICLE 5- SUBMISSION OF QUALIFIED BIDS

Section 5.1 Continued Participation in the Solicitation Process

(1) A Participating Bidder must deliver a Qualified Auction Bid or Qualified Asset Purchase Bid (or both) to the Receiver and such Qualified Auction Bid or Qualified Asset Purchase Bid, as applicable, must be received by the Receiver by no later than 5:00 p.m. (Calgary time) on July

15, 2015, or such later date or time as the Receiver may determine appropriate (the "Qualified Bid Deadline").

Section 5.2 Qualified Auction Bid

- (1) An Auction Bid submitted by a Participating Bidder will be considered a "Qualified Auction Bid" only if the Auction Bid complies with all of the following:
 - (a) it is a Superior Bid;
 - (b) it includes a letter stating that the Auction Bid is irrevocable until the earlier of (a) the approval by the Court of a Successful Bid, and (b) 45 Business Days following the Bid Deadline; provided, however, that if such Auction Bid is selected as the Successful Bid or the Backup Bid, it shall remain irrevocable until the closing of the Successful Bid or the Backup Bid, as the case may be;
 - (c) it includes a duly authorized and executed auction services agreement that is in substantially the same form as and incorporates all or substantially all of the same terms and conditions as the Great American Auction Services Agreement except as otherwise prescribed in these AASP Procedures;
 - (d) it does not include any request or entitlement to any break-fee, expense reimbursement or similar type of payment in the event that it is not the Successful Bid;
 - (e) it is not conditioned on (i) the outcome of unperformed due diligence by the bidder and/or (ii) obtaining any financing capital;
 - it is accompanied by a refundable deposit in the form of a wire transfer (to a trust account specified by the Receiver), or such other form acceptable to the Receiver, payable to the order of Deloitte Restructuring Inc., in trust, in an amount equal to 20% of the net minimum guaranteed amount as set out in the applicable auction services agreement included in such Auction Bid, to be held and dealt with in accordance with these AASP Procedures;
 - (g) it contains other information reasonably requested by the Receiver; and
 - (h) it is received by no later than the Qualified Bid Deadline.

Section 5.3 Qualified Asset Purchase Bid

- (1) An Asset Purchase Bid submitted by a Participating Bidder will be considered a "Qualified Asset Purchase Bid" only if the Asset Purchase Bid complies with all of the following:
 - (a) it is a Superior Bid;
 - (b) it includes a letter stating that the Asset Purchase Bid is irrevocable until the earlier of (a) the approval by the Court of a Successful Bid, and (b) 45 Business Days following the Bid Deadline; provided, however, that if such Asset Purchase

Bid is selected as the Successful Bid or the Backup Bid, it shall remain irrevocable until the closing of the Successful Bid or the Backup Bid, as the case may be;

- (c) it includes a duly authorized and executed asset purchase agreement substantially consistent with the terms of the Receiver's form of asset purchase agreement provided to such bidder specifying the purchase price, expressed in Canadian dollars (the "Purchase Price"), together with all exhibits and schedules thereto, and such ancillary agreements as may be required by the Qualified Bidder with all exhibits and schedules thereto (or term sheets that describe the material terms and provisions of such ancillary agreements) and such ancillary agreements and the proposed orders to approve the sale by the Court;
- (d) it does not include any request or entitlement to any break-fee, expense reimbursement or similar type of payment in the event that it is not the Successful Bid.
- (e) it is not conditioned on (i) the outcome of unperformed due diligence by the bidder and/or (ii) obtaining any financing capital and includes an acknowledgement and representation that the bidder has had an opportunity to conduct any and all required due diligence prior to making its Asset Purchase Bid;
- (f) it fully discloses the identity of each entity that is bidding or otherwise that will be sponsoring or participating in the Asset Purchase Bid, including the identification of the Qualified Bidder's direct and indirect owners and their principals, and the complete terms of any such participation;
- (g) it is accompanied by a refundable deposit in the form of a wire transfer (to a trust account specified by the Receiver), or such other form acceptable to the Receiver, payable to the order of Deloitte Restructuring Inc., in trust, in an amount equal to 20% of the Purchase Price, to be held and dealt with in accordance with these AASP Procedures;
- (h) it provides for closing of the Asset Purchase Bid by no later than the Target Asset Purchase Closing Date;
- (i) if the Qualified Bidder is an entity newly formed for the purpose of the transaction, the bid shall contain an equity or debt commitment letter from the parent entity or sponsor, which is satisfactory to the Receiver, that names the Receiver as a third party beneficiary of any such commitment letter with recourse against such parent entity or sponsor;
- (j) it contains other information reasonably requested by the Receiver; and
- (k) it is received by no later than the Qualified Bid Deadline.

Section 5.4 Qualified Bids

- (1) Qualified Auction Bids and Qualified Asset Purchase Bids shall hereinafter be referred to as "Qualified Bids" and each a "Qualified Bid" and each bidder who has submitted a Qualified Bid shall hereinafter be referred to as a "Qualified Bidder".
- (2) Notwithstanding Section 5.2 and Section 5.3 hereof, the Receiver may waive compliance with any one or more of the Qualified Bid requirements specified herein, and deem such non-compliant bids to be Qualified Auction Bids and Qualified Asset Purchase Bids, as the case may be.

Section 5.5 No Qualified Bids

- (1) If the Receiver determines at any time after July 7, 2015, acting reasonably, that it is unlikely that there will be any Qualified Bidders or that any Qualified Bids will be received, the Receiver shall (a) forthwith terminate these AASP Procedures; (b) notify each Participating Bidder (if any) that these AASP Procedures have been terminated.
- (2) The Receiver will assess the Qualified Bids received, if any, and will determine whether it is likely that the transactions contemplated by such Qualified Bids are likely to be consummated and whether proceeding with these AASP Procedures is in the best interests of the Debtor and its stakeholders. Such assessments will be made as promptly as practicable but no later than five (5) Business Days after the Qualified Bid Deadline, provided that such time period may be extended by the Receiver upon prior written notice to Qualified Bidders and GA or order of the Court upon application and appropriate notice.
- (3) If the Receiver, in accordance with Section 5.5(1) and 5.5(2) above, determines that (a) it is unlikely that there will be any Qualified Bids or Qualified Bidders, (b) there are no Qualified Bids or Qualified Bidders, or (c) at least one Qualified Bid was received but it is not likely that the transactions contemplated in any such Qualified Bids will be consummated, the Receiver shall, within ten (10) Business Days of such determination, file an application with the Court for approval of the GA Auction Agreement.

Section 5.6 Auction

(1) In the event that the Receiver receives at least one Qualified Bid that is an Auction Bid or at least one Qualified Bid that is an Asset Purchase Bid prior to the Bid Deadline, the Receiver shall invite the Qualified Bidders that submitted such Qualified Bids to an auction with **GA** (the "Auction"). The Auction shall be conducted by the Receiver at the Receiver's offices on July 17, 2015 in accordance with auction procedures to be established by Receiver. Without limiting the foregoing, the Qualified Bid with the highest guaranteed net cash proceeds will form the base bid in the Auction and initial minimum bid increments will be set at \$200,000.

Section 5.7 Selection Criteria

(1) The Receiver may select Qualified Bids for further negotiation and/or clarification of any terms or conditions of such Qualified Bids, including the amounts offered, before identifying the highest or otherwise best Qualified Bid(s) received (the "Successful Bid").

- (2) The Receiver will identify the Successful Bid and may identify a next highest or otherwise best Qualified Bid received (such offer, the "Backup Bid"). The Qualified Bidders(s) who made the Successful Bid is/are the "Successful Bidder" and the Qualified Bidder(s) who made the Backup Bid (if a Backup Bid is identified in accordance with these AASP Procedures) is/are the "Backup Bidder"). The Receiver will notify the Successful Bidder and any Backup Bidder that they are, respectively, the Successful Bidder and the Backup Bidder.
- (3) The Receiver will finalize definitive agreements in respect of the Successful Bid and the Backup Bid (if a Backup Bid is identified in accordance with these AASP Procedures), if any, conditional upon approval by the Court.
- (4) If a Backup Bid is identified in accordance with these AASP Procedures, then such Backup Bid shall remain open until the consummation of the transaction contemplated by the Successful Bid (the "Backup Bid Expiration Date").
- (5) All Qualified Bids (other than the Successful Bid and any Backup Bid identified in accordance with these AASP Procedures) shall be deemed rejected by the Receiver on and as of the date of approval of the Successful Bid or any Backup Bid by the Court.

Section 5.8 Approval Hearing

- (1) After definitive agreements in respect of a Successful Bid and Backup Bid (if a Backup Bid is identified in accordance with these AASP Procedures) have been finalized, in the case of the Successful Bid, signed (conditional on court approval) and, in the case of the Backup Bid signed (conditional on non-completion of the Successful Bid and on court approval) in accordance with these AASP Procedures, the Receiver shall seek a hearing as soon as practicable on a date to be scheduled by the Court that will permit not less than 2 full Business Days' notice to the service list (the "Approval Hearing") to approve the Successful Bid and the Backup Bid (such order, the "Approval Order"). The Approval Hearing may be adjourned or rescheduled by the Receiver in its sole discretion.
- (2) If, following approval of the Successful Bid transaction by the Court, the Successful Bidder fails to consummate the transaction for any reason, then the Backup Bid, if there is one, will be deemed to be the Successful Bid hereunder and the Receiver shall effectuate a transaction with the Backup Bidder subject to the terms of the Backup Bid, without further order of the Court.

Section 5.9 Deposits

(1) All deposits provided pursuant to Section 5.2 and Section 5.3 shall be retained by the Receiver and invested in an interest bearing trust account in a Schedule I Bank in Canada. If there is a Successful Bid, the Deposit (plus accrued interest) paid by the Successful Bidder whose bid is approved pursuant to the Approval Hearing shall be applied to the purchase price to be paid or investment amount to be made by the Successful Bidder upon closing of the Successful Bid. The Deposit (plus accrued interest) paid by the Backup Bidder, if there is one, shall be retained by the Receiver until the Backup Bid Expiration Date or, if the Backup Bid becomes the Successful Bid, shall be applied to the purchase price to be paid or investment amount to be made by the Backup Bidder upon closing of the Backup Bid. The Deposits (plus applicable interest) of all Qualified Bidders not selected as the Successful Bidder or Backup

Bidder shall be returned to such bidders with interest within five (5) Business Days of the later of the date upon which the Successful Bid and any Backup Bid are approved by the Court. If these AASP Procedures are terminated in accordance with the provisions hereof, all Deposits shall be returned to the bidders with interest within five (5) Business Days of the date upon which these AASP Procedures are terminated.

(2) If an entity selected as the Successful Bidder or Backup Bidder breaches its obligations to close or to pay the net minimum guarantee set out in its definitive agreement, it shall forfeit its Deposit to the Receiver; provided, however, that the forfeiture of such Deposit shall be in addition to, and not in lieu of, any other rights in law or equity that the Receiver has or may have against such breaching entity.

Section 5.10 Notice to the Receiver

(1) Any notice or other communication to be given to the Receiver in connection with this AASP shall be given in writing and shall be given by personal delivery (in which case it shall be left with a responsible officer of the recipient) or by electronic communication addressed to the Receiver as follows:

Deloitte Restructuring Inc. 700, 850 - 2 Street SW Calgary, Alberta T2P OR8

Attention: Jeff Keeble, CA, CIRP, CBV Email: jkeeble@deloitte.caAttention:

Tel No.: 403-503-1458

with a copy to:

Gowling Lafleur Henderson LLP Suite 1600, 421 7th Avenue S.W. Calgary, Alberta T2P 4K9

Attention: Jeffrey Oliver

Telephone No.: (403) 298-1818 Facsimile No.: (403) 695-3558

Email: jeffrey.oliver@gowlings.com

Section 5.11 Reservation of Rights

(1) The Receiver may (a) reject, at any time any bid that is (i) inadequate or insufficient, (ii) not in conformity with the requirements of these AASP Procedures or any orders of the Court applicable to the Debtor, or (iii) contrary to the best interests of the Debtor, its estate, and stakeholders as determined by the Receiver; (b) in accordance with the terms hereof accept bids not in conformity with these AASP Procedures to the extent that the Receiver determines, in its reasonable business judgment, that doing so would benefit the Debtor, its estate, and stakeholders; and (c) in accordance with the terms hereof extend the Qualified Bid Deadline and the Qualified Bid Deadline; and (d) reject all bids. The Receiver shall not be required to accept

the highest bid, but shall be entitled to recommend to the Court a transaction that in its view maximizes value for all stakeholders.

(2) These AASP Procedures do not, and shall not be interpreted to, create any contractual or other legal relationship between the Receiver on the one hand and any Known Potential Bidder, Potential Bidder, Participating Bidder, Qualified Bidder, Successful Bidder or Backup Bidder, on the other hand, except as specifically set forth in definitive agreements that may be executed by the Receiver.

Section 5.12 Further Orders

(1) At any time during the AASP, the Receiver may apply to the Court for directions with respect to the discharge of its powers and duties hereunder.

SCHEDULE A

Asset List

Year	Make	Description	tejeatiens
1974	RUSSELL	T/A ALUM NITROGEN TANK TRAILER, CGA-341, 61350-LB CAPACITY, TANK S/N 976, INSULATED, 4	ED
		CTLINDER KOBOTA DIESEL ENGINE, TORSO POWIF, 11022.5 TIMES-ALGIVI BODS, (LIC V310-33)	
1977	RUSSELL	, and the second	
1981	CHAMPION	V510-94J CGA-341, 68,209-LB TANK MOTOR GRADER	ED
1981	MACK	PICKER TRUCK, 12.5-TON	ED
1988	BOBCAT	SKID STEER LOADER, TIRES 12-16.5 NHS	MEDICINE
1990	WABASH	T/A WATER PUP, TIRES 11R24.5-STEEL BUDS, NON-SPEC STEEL TANK, (LIC F77605), 12,000-LITRE	HAT MEDICINE
2550		CAPACITY	HAT
1990	WEST TANK		SWIFT
1990	WEST TANK		CURRENT MH
1330	1	,	*****
		INSPECTION]	
1993	BEALL	, , , , , , , , , , , , , , , , , , , ,	MH
1994	ROUGHNECK		ED .
1994	FREIGHTLINER	· · · · · · · · · · · · · · · · · · ·	ED
1994	WABASH	TRI/AXLE TANK PUP TRAILER, 160-BARREL, TIRES 11R24.5-ALUM BUDS, AIR RIDE SUSPENSION	MEDICINE
4004	11/40 401/	TA CHEE DOLE ALLIMATICATED TANK BUD TRAILED TIDES (44D24 F. CTCF) DANTON CROING	HAT
1994	WABASH		SWIFT CURRENT
1995	WESTERN STAR		ZAMA CITY
		TIRES 11R24.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, CHALMERS AIR RIDE SUSPENSION,	
		l '	
		TANK S/N 030205278, TC WATER 3" FILID PLMP	
1995	ADVANCE	QUAD PUP, TC306 CRUDE, TIRES 11R24.5-STEEL BUDS, SPRING SUSPENSION, 2 COMPARTMENT,	SWIFT
		15,000-LITRE, 10,000-LITRE CAPACITY, FULL TURN TABLE, (LIC V547-77)	CURRENT
			SC
1550	'','	CENTIFIED OTHERS TO THE RECEIVED TO SEE STATE OF THE SECOND OF THE SECON	
1997	N/A	SIDE SPOOLER CUSTOM BUILT, SKID MOUNTED	ED
1998	POLAR	TRIDEM ALUM TANK TRAILER, 34M3 CAP, 11R24.5, ALUM BUDS, SPRING SUSPENSION	SWIFT CURRENT
1998	ADVANCE	TRIDEM TC306 ALUM TANK TRAILER, TIRES 11R24.5-STEEL BUDS, 37,000-LITRE CAPACITY, AIR RIDE	SWIFT
		SUSPENSION, (LIC V547-91)	CURRENT
1998	KENWORTH		МН
1998	VOLVO/AUTOC	AUTOCAR TANDEM-AXLE STEER/TRI-AXLE DRIVE COILED TUBING UNIT, CAT 500-HP, 18-SPEED	ED
	AR	l '	
			•
		· · · · · · · · · · · · · · · · · · ·	
1998	ADVANCE	TRIDEM TC306 ALUM TANK TRAILER 34M3 CAP, 11R24.5, STEEL DAYTON, SPRING SUSPENSION, TRI- AXLE, 30,000-LITRE CAPACITY	
	1974 1977 1981 1981 1988 1990 1990 1990 1994 1994 1994 1994 1995 1995 1995 1996 1997 1998 1998	1974 RUSSELL 1977 RUSSELL 1981 CHAMPION 1988 BOBCAT 1990 WABASH 1990 WEST TANK 1990 WEST TANK 1991 ROUGHNECK 1994 ROUGHNECK 1994 WABASH 1994 WABASH 1995 WESTERN STAR 1995 N/A 1996 N/A 1997 N/A 1998 ADVANCE 1998 ADVANCE 1998 KENWORTH 1998 VOLVO/AUTOC AR	1974 RUSSELL T/A ALUM NITROGEN TANK TRAILER, CGA-341, 5150-16 CAPACITY, TANK S/N 976, INSULATED, 4 CYLINDER KUBOTA DIESEL ENGINE, TURBO PUMP, 11872.5 TIRES -ALUM BUDS, (ILC V510-93) 1977 RUSSELL ALUM TANDEM-AXLE NITROGEN TRAILER, TIRES 255/75822.5, ALUM BUDS, S/N NA10026/643 (ILC V510-94) CGA-341, 68, 200-18 TANK MOTOR GRADER 1981 MACK PICKER TRUCK, 12.5-TON 1988 BOBCAT SKID STEFR LOADER, TRES 12-16.5 NHS 1990 WARASH 1/A WATER PUP, TIRES 1172.4-5-TEEL BUDS, NON-SPEC STEEL TANK, (ILC F77605), 12,000-11 TRE CAPACITY 1990 WEST TANK WEST TANK WEST TANK GUAD WAGON TANK PUP TRAILER, TIRES 1172.5, STEEL BUDS, SPRING SUSPENSION, 12306, 223/3 CAP, S/N WANTENSON, PULL TURN TABLE, (ILC V547-80) (NOT IN SERVICE AT TIME OF NSPECTION) 1990 BEALL 1991 WEST TANK WEST TANK QUAD WAGON TANK PUP TRAILER, TIRES 1172.4-5, SPRING SUSPENSION, PULL TURN TABLE, (ILC V547-80) (NOT IN SERVICE AT TIME OF NSPECTION) 1994 BEALL TANK QUAD WAGON PUP TRAILER, MC406, 34M3 CAPACITY, TIRES 1172.4-5, ALUM BUDS, SPRING SUSPENSION, LICL V547-80) (NOT IN SERVICE AT TIME OF NSPECTION) 1994 ROUGHNECK TANDEM-AXIE, FRONT TIRES 1272.5-ALUM BUDS, REAR TIRES 1172.4-5, ALUM BUDS, SPRING SUSPENSION, LICL V547-80 (NOT IN SERVICE AT TIME OF NSPECTION) 1994 WABASH TRI/AXIE TANK PUP TRAILER, IGO-BARREL, TIRES 1172.4-5, ALUM BUDS, SPRING SUSPENSION, LICL G10813) 1994 WABASH TRI/AXIE TANK PUP TRAILER, 160-BARREL, TIRES 1172.4-5, ALUM BUDS, AIR RIDE SUSPENSION 1994 WABASH TRI/AXIE TANK PUP TRAILER, 160-BARREL, TIRES 1172.4-5, STEEL DAYTON, SPRING SUSPENSION, C/W HAMMS 100 BARREL STEEL TANK S/N 3000705778, CW WAFF, SPRING SUSPENSION, C/W HAMMS 100 BARREL STEEL TANK S/N 3000705778, CW WAFF, SPRING SUSPENSION, PUP TRAILER, SPEED TRAINSMISSION, FRONT TIRES 1172.4-5-ALUM BUDS, REAR TIRES 1172.4-5-ALUM BUDS, SPRING SUSPENSION, C/W HAMMS 100 BARREL STEEL TANK S/N 3000705778, TO WAFF, SPRING SUSPENSION, PUP TRAILER, 160-BARREL STEEL TANK S/N 3000705778, CW WAFF, SPRING SUSPENSION, PUP TAILER, 160-BARREL STEEL TANK S/N 3000705778, AURUS S/N 3000705778, CW WAFF, SPRING SUSPENS

Equip. #		Make	Description	- mentilon
CB902	1999	1	T/A COMBO UNIT, M11-370 ESP ENGINE, EATON FULLER 8-SPEED TRANSMISSION, 445/65R22.5	ED
			FRONT TIRES, ALUM BUDS, 11R24.5 REAR TIRES, ALUM BUDS, (LIC F96607), C/W CUSTOM VAC	
			SERVICES COMBO UNIT S/N 0498026, 10,862-LITRE CAPACITY, TC350, HOTSY STEAM, WATER BLAST,	
			BLOWER, TRIPLEX PUMP, 271,389-KILOMETERS INDICATED, 5,541-HOURS INDICATED	
TWPK004	1999	JASPER	T/A STIFF POLE TANK TRAILER PUP, TIRES 11R24.5-STEEL DAYTONS, C/W JASPER SERVICE CARGO	ED
			TANK S/N 2J9JT1827YS031104, TC307, 14,956-LITRE CAPACITY	
FKLC003	1999	LINDE	LIFT TRUCK, 6000-LB CAPACITY, LPG, DOUBLE MAST, 192" MAXIMUM LIFTING HEIGHT, SIDE SHIFT, 5'	ED
			FORKS, PNEUMATIC TIRES, 9,996-HOURS INDICATED	
FH002	2000	INTERNATIONA	FUEL TANKER TRUCK, SINGLE AXLE, DT 466E ENGINE, 6-SPEED, FRONT TIRES 11R22.5, STEEL BUDS,	CNRL
	-	L	REAR TIRES 11R22.5, STEEL BUDS, C/W 1989 HUTCHINSON IND. TANK, 4 COMPARTMENTS - 4,000 L,	
			3,000 L, 2,500 L, 2,000 L, S/N 89-6386	
SSS003	2000	CATERPILLAR	SKID STEER LOADER, W/ GENERAL PURPOSE BUCKET, SNOW PLOW, 1914-HOURS INDICATED	SWIFT
				CURRENT
TRS112	2000	PETERBILT	T/A HIGHWAY TRACTOR W/ 63" STAND UP SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED	SWIFT
			TRANSMISSION, FRONT TIRES 11R24.5-ALUM BUDS, 12,000-LB AXLE, REAR TIRES 11R24.5-ALUM	CURRENT
		ļ	BUDS, 46,000-LB AXLES,238" WB, (LIC G10811), [NOT IN SERVICE AT TIME OF INSPECTION, ROLL	
			OVER DAMAGE	
CT201	2000	VOLVO/AUTOC	AUTOCAR TANDEM FRONT/REAR-AXLE DRIVE COILED TUBING UNIT, CUMMINS N14 ENGINE, EATON	ED
		AR	FULLER 18-SPEED, 37,400-LB FRONT AXLES, FRONT TIRES 445/65R22.5, ALUM BUDS, 46,000-LB REAR	
			AXLES, REAR TIRES 11R24.5, ALUM BUDS, (LIC G20898), C/W COILED TUBING UNIT, 3100M OF 1 1/4 $^{"}$	
			TUBING, HURRICANE COMPRESSOR 903-84 S/N 80394, HYDRARIG RT25 !NJECTOR W/ARCH	
			GOOSENECK, ATLAS CRANE KNUCKLE PICKER AK1401-A4 S/N 104700, HYDRAULIC SPOOL, CONTROL	
			CAR_CAT_4517_TRIPLEX_PUMP	
PD118	2000	VOLVO	PICKER PILE DRIVER	ED
ATK007	2000	JASPER	TRIDEM TANK TRAILER, TIRES 11R24.5-ALUM BUDS, SPRING SUSPENSION, C/W 2 COMPARTMENT	ZAMA CIT
			STEEL TANK - 32,750-LITRE CAPACITY, DOT 412	
W033	2001	1	WATER TRUCK, T/A, CAT C12 355-HP ENGINE, EATON FULLER 13-SPEED TRANSMISSION, FRONT	ED
			TIRES 425/65R22,5-ALUM BUDS, 14,000-LB FRONT AXLE, REAR TIRES 11R24.5-STEEL BUDS, 40,000-	
			LB REAR AXLES, AIR RIDE SUSPENSION, C/W ADVANCE WATER TANK S/N 2AESTHB042E000150,	
			TC306-MS CRUDE, 15,900-LITRE CAPACITY, BOWIE 4" PRODUCTION PUMP, 1,975-KILOMETERS	
	<u> </u>		SHOWING (NEW GALIGE). 9.829-HOURS REPORTED	
W039	2001	WESTERN STAR	4964FX TANDEM-AXLE WATER TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED, 16000-LB FRONT	ED
			/ 23000-LB X 2 REAR AXLES, FRONT 425/65R22.5-ALUM BUDS, REAR 11R24.5-STEEL BUDS, AIR RIDE	
			SUSPENSION, C/W ADVANCE CRUDE TANK, 15900-LITRE, S/N 2AESTHB032E000169 [COLLISION	
			DAMAGED : SALVAGE ONLY)	
SSK004	2001	MACK	T/A SOUR SEAL TRUCK W/ 46" SLEEPER, E7 460 XT ENGINE, EATON FULLER 13-SPEED	GP
		1	TRANSMISSION, FRONT TIRES 315/80R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, AIR RIDE	
			SUSPENSION, C/W JASPER STEEL TANK S/N 219JT1837Y031017, TC307, 14,956-LITRE CAPACITY, 3"	
			FLUID PUMP.	
ST200	2001	KENWORTH	HD SERVICE TRUCK W/ SLEEPER, EATON FULLER TRANSMISSION, PALFINGER 22' BOOM W/ 4,450-LB.	ED
	ļ		CAPACITY, WELDER/GENERATOR, AIR COMPRESSOR	
NP016	2001	ASPEN	NITROGEN PUMP TRAILER, INCLUDES PUMP, DRIVEN BY DIESEL ENGINE, NITROGEN TANK, AND	ED
			OPERATORS CAB, UNITIZED ON TRI-DEM TRAILER, PURCHASED USED IN 2014, APPEARS REBUILT	
ATK046	2001	JASPER	TRIDEM TANK TRAILER, TIRES 11R24.5-ALUM BUDS, SPRING SUSPENSION, C/W 2 COMPARTMENT -	ZAMA CIT
			32,750-LITRE, TC407/412	
SSK045	2001	WESTERN STAR	T/A SOUR SEAL TRUCK W/ 48" SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ZAMA CI
		1	FRONT TIRES 385/65R22.5-ALUM BUDS, 14,600-LB FRONT AXLE, REAR TIRES 40X31.00-20 NHS,	
			46,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W JASPER STEEL TANK, TC412, 14,900-LITRE	
			CAPACITY, 3" T&E FLUID PUMP	
V029	2001	WESTERN STAR	T/A VACUUM TRUCK, 4964SX, CAT C12, EATON FULLER 18-SPEED, 425/65R22.5 FRONT, ALUM BUDS,	MEDICIN
			48X31.00-20 FLOATATION TIRES, STEEL RIMS, TC350 CRUDE TANK, ESTIMATED 12,000-LITRE	HAT
	1		CAPACITY, (LIC G-19561)	

Equip. #	Year		Description	£0(e)(i)010
WS108	2001	i i	T/A TANK TRUCK W/ SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	SWIFT
)	TIRES 385/65R22.5-ALUM BUDS, 14,600-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB	CURRENT
		1	AXLES, AIR RIDE SUSPENSION, 236" WB, PINTLE HITCH, SPRAY BAR, (LIC G10812), C/W WABASH	
			TC306 ALUM 16M3 TANK, S/N 63022121D16001275, HEATED SIDE CABINET W/HYDRAULICS, 4" T&E	
			HYD PUMP, 517,739-KILOMETERS INDICATED, 18,918-HOURS INDICATED	
WS118	2001	KENWORTH	T/A TANK TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, 12,000-LB FRONT AXLE, FRONT	SWIFT
}			TIRES 11R24.5, ALUM BUDS, 46,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE	CURRENT
			SUSPENSION, 236" WB, (LIC G10808), C/W TC406 CRUDE TANK, SPRAY BAR, PINTLE HITCH, 16M3	
			CAP, HEATED SIDE CABINETS W/HYDRAULICS, 4" T&E HYD PUMP, 726,868-KILOMETERS INDICATED,	
UTC332	2001	ROUGHNECK	17.721-HOURS INDICATED 32' TANDEM-AXLE GOOSENECK FLATDECK TRAILER, TIRES 235/85R16-STEEL BUDS, 40,000-LB AXLE	ED
			THE COLUMN TWO IS NOT THE PARTY OF THE COLUMN TWO IS NOT THE COLUM	C.D.
V020	2001	WESTERN STAR	TANDEM-AXLE VACUUM TRUCK, CAT C12 ENGINE, EATON FULLER 18-SPEED, 150,939-KILOMETERS	GP
			INDICATED, 20,685-HOURS INDICATED, FRONT AXLE 425/65R22.5 ALUM BUDS 16,000-LB, REAR	
			AXLES 48X31.00-20 STEEL BUDS 46,000-LB, C/W VACUUM TECH UNIT TC 350, TCRN 25-383, 13,229 L	
			CAPACITY S/N 1000029, HIBON BLOWER	
STD209	2001	FORD	SERVICE TRUCK, 4,X4, DUALLY, DIESEL ENGINE, 6-SPEED TRANSMISSION, TIRES 245/70R19.5, C/W	ED
			GENERAL BODY & EQUIPMENT LTD. 11'X8' SERVICE BODY W/ LIFTMORE 3200 10,000-LB TRUCK	
			CRANE, UMAC UNDERHOOD ROTARY AIR COMPRESSOR	
SSK005	2001	WESTERN STAR	T/A SOUR SEAL TRUCK W/ 36" SLEEPER, CUMMINS N14 ENGINE, EATON FULLER 18-SPEED	
			TRANSMISSION, FRONT TIRES 385/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, C/W JASPER STEEL CARGO TANK S/N	
55304	2004	C TECH	219)T18371S031073, TC307, 14.956-LITRE CAPACITY	
CC201	2001		CATCH TANK, TANDEM-AXLE, 7.5R16 TIRES, 10000-LB X 2 AXLES	ZANAA CITV
TWPK003	2002	JASPER	TRIDEM TANK TRAILER, TIRES 11R24.5-ALUM BUDS, SPRING SUSPENSION, S/N 2J9JT00371S031001, C/W 2 COMPARTMENT STEEL TANK, 32,750-LITRE CAPACITY, DOT 412	ZAMA CITY
BT001	2002	CUSTOM	TRAILER, FLATBED, 24' WOOD DECK, TANDEM-AXLE	MEDICINE
RIOOT	2002	MANUFACTUR	TRAILER, FLATBED, 24 WOOD DECK, TAINDEIVI-AALC	HAT
				1941
VR101	2002	ED KENWORTH	T/A VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	MEDICINE
ANTOT	2002	KENVORIII	TIRES 385/65R22.5-ALUM BUDS,19,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-	HAT
			LB REAR AXLES, CHALMERS SUSPENSION, C/W PROXY WELD INDUSTRIES S/N 03-0611, TC407/412,	,,,,,
			13.033-LITRE CAPACITY, HIBON VTB 800 BLOWER	
RN034	2002	ROUGHNECK	T/A 28' RV TRAILER, TIRES 235/85R16, COMPLETE	ED
101037	2002	ROGGINATOR		
RV039	2002	CITATION	TRAVEL TRAILER, 26', CREW SHACK	ED
RN027	2002	ROUGHNECK	28' T/A RV TRAILER, TIRES LT235/85R16-50%, FURNACE, WATER HEATER, RANGE, REFRIGERATOR,	
			A/C, TOILET, SHOWER [REPORTED AS PERMANENTLY OUT OF SERVICE]	
RV050	2002	CITATION	26' TRAVEL TRAILER, TIRES LT235/85R16, STEEL BUDS	MH
V082	2003	WESTERN STAR	TRI-AXLE VACUUM TRUCK, CAT C15, EATON FULLER 18-SPEED, 445/65R22.5 FRONT, 11R24.5 REAR,	ED
V002	2005	WESTERNSTAN	ALUM BUDS, AIR RIDE, (LIC G-10823), C/W CUSTOM VAC UNIT, TC412, 19193 L CAP, HIBON VTB820,	
		ļ		
W072	2003	WESTERN STAR	S/N 12471203, 234,152-KILOMETERS INDICATED, 22,384-HOURS INDICATED WATER TRUCK, TANDEM	ED
W072	2003	WESTERNSTAR	WALES MOCK, IAMOCK	
PTI105	2003	INTERNATIONA	S/A PRESSURE TEST TRUCK, DT 466 ENGINE, EATON FULLER 10-SPEED ENGINE, FRONT TIRES 11R22.5-	GP
		L	ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 11R22.5-ALUM BUDS, 20,000-LB REAR AXLE, AIR	
			RIDE SUSPENSION, (LIC G03344), C/W INTERCONTINENTAL 20' TRUCK BODY S/N 4685,	
			BUTTERWORTH TRIPLEX PUMP, 4 ALUM TANKS, OPERATOR CONTROL ROOM [REPORTED AS OUT OF	
			SERVICE1	
S001	2003	WESTERN STAR	T/A STEAMER UNIT TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, FRONT AXLE	CNRL
			425/65R22.5 ALUM BUDS 16,000-LB, REAR AXLES 11R22.5 ALUM BUDS 46,000-LB, C/W THERMO	
			KING VAN BODY S/N VB3190, HOTSY 9480 STEAMER, ESTIMATED 8,000-LITRE CAPACITY TANK	
		1		

Eguip-#	Year	Make	Description	ligentien.
V055	2003		TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 375-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
			SERVICES S/N 19179803, TC412, 15,934-LITRE CAPACITY, HIBON BLOWER, TIRE PRESSURE	
			MONITORS	
V056	2003	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB AXLES, AIR	
			RIDE SUSPENSION, (LIC F16261) C/W CUSTOM VAC SERVICES, TC SPEC 412, 19,200-LITRE CAPACITY -	
			S/N 191799-03. HIBON BLOWER- S/N 15571551	
V057	2003	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	ED
			445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK, S/N 349461-	
50400	2000	WEED LET COLL	03, 18,280 L, 271209-KILOMETERS INDICATED	70
SR130	2003		SINGLE AXLE STEAM TRUCK, EATON FULLER 10-SPEED TRANSMISSION, 11R22.5 TIRES, ALUM BUDS,	ZC
		L	C/W INTERCONTINENTAL TRUCK BODY S/N ITB0211836, HOTSY STEAMER, 122,421-KILOMETERS	
DEC.O.O.	7007	CNC	INDICATED, 20,828-HOURS INDICATED	ED.
PF302	2003	GMC	PICKUP TRUCK, EXTENDED CAB, 4X4	ED
SSK003	2003	WESTERN STAR	TRI-DRIVE SOUR SEAL TANK TRUCK W/ 36' SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-	ZAMA CITY
		1	SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 2D,000-LB FRONT AXLES, REAR TIRES	
			11R24.5-ALUM BUDS, 42,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W JASPER TANK, 3"	
			FLUID PUMP	
Q002	2003	CUSTOM VAC	CUSTOM VAC QUAD WAGON VAC TANK PUP TRAILER, TIRES 11R22.5, ALUM BUDS, AIR RIDE	SC
		SERVICES	SUSPENSION, C/W CUSTOM VAC TANK S/N 10110703, 21,746-LITRE, TC412	
V059	2003	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	MEDICINE
			TIRES 65R22,5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, SLEEPER, C/W CUSTOM VAC	HAT
			SERVICES TANK, 18,200-LITRE, S/N 349473-03	
CD099	2003	DODGE	SERVICE TRUCK [CONVERTED PICKUP TRUCK] FULL UTILITY BODY, QUAD CAB, 4WD [ENGINE	SWIFT
			REPLACED), 361,324-KILOMETERS INDICATED	CURRENT
TWPS010.	2003	ALLWELD	TRIDEM TC406 ALUM TANK TRAILER, 74,000-LB AXLES, TIRES 11R24.5, ALUM BUDS, AIR RIDE	SWIFT
!			SUSPENSION, SIDE FLOATS, (LIC V547-85)	CURRENT
UT001	2003	SOUTHLAND	T/A DUMP TRAILER, LT235/85R15 TIRES, STEEL BUDS, (LIC R300-39)	ED
	2000	VENNICOTIL	THE DOLLES WERE A SHALL A TRUCK WE CALL CLEEPER OF THE LIP FAIGHT. FATOMER HAD BEEN A	F.D.
HR120	2003	KENWORTH	TRI-DRIVE HYDRO-VACUUM TRUCK W/ 36" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-	ED
			SPEED, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM	
			BUDS, 58,000-LB REAR AXLES, NEW WAY AIR RIDE SUSPENSION, C/W WESTECH VAC SYSTEM HYDRO-	
			VAC SYSTEM, 11,166-LITRE CAPACITY, DOT 407, S/N 1202801, HIBON SIAV 822 BLOWER, S/N	
			046346732, HOTSY WATER BLAST WB 9460 HEATER, S/N 51880903, 16,465-HOURS INDICATED	
			[SPEEDOMETER REPLACED; KILOMETERS ESTIMATED TO BE 360,000]	
MV005	2003	MOUNTAIN	TRIDEM WELLSITE TRAILER, TIRES LT265/75R16	ED
		VIEW		
ST205	2003	FORD	SERVICE BODY TRUCK, SUPER CREW, 4X4, POWER STROKE V8 DIESEL ENGINE, AUTOMATIC	
			TRANSMISSION, FRONT TIRES 225/70R19.5 ALUM BUDS, REAR TIRES 225/70R19.5 STEEL BUDS, C/W	
		·	WESTERN TRUCK BODY SERVICE BODY, GRILL GUARD	
\$002	2003	WESTERN STAR	T/A STEAMER UNIT, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, 16,000-LB FRONT AXLE,	MH
			FRONT TIRES 425/65R22.5, ALUM BUDS, 46,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS,	
			AIR RIDE SUSPENSION, C/W THERMO KING 4-13248 VAN BODY S/N VB3186, HOTSY 9480 STEAMER,	
			WATER BLAST, HOSE REELS, 3 POLY TANKS, ESTIMATED 5,000-LITRE CAPACITY TANK	
) NIC440	2000	11100	T/A TANIK TRUCK NAACK ACO NAAKITOROUE 40 CDEED 4480A E EDOUT 4000UR EDOUT	
WS110	2003	MACK	T/A TANK TRUCK, MACK 460, MAXITORQUE 18-SPEED, 11R24.5 FRONT, 12000LB FRONT, ALUM	
			BUDS, 11R24.5 REAR, 46000-LB REAR, AIR RIDE, 236" WB, C/W TC406 CRUDE TANK W/SPRAY BAR,	
			PINTLE HITCH, PTO BOWEE PUMP, 16M3 CAP, TANK S/N 2AESTHB044S000184 [REPORTED AS	
1/004	2004	WESTERN ST.	PERMANENTLY OUT OF SERVICE	MEDICINE
V084	2004	WESTERN STAR	TRI-AXLE VACUUM TRUCK, CAT ENGINE, EATON FULLER 3-SPEED TRANSMISSION, FRONT TIRES	MEDICINE
1			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 54,000-LB AXLES, AIR	HAT
			RIDE SUSPENSION, (LIC F32691) C/W CUSTOM VAC, TC SPEC 412, 19,193-LITRE CAPACITY -	
	L		ISN17841204, HIBON BLOWER, TIREBOSS TIRE PRESSURE CONTROL	

Equip.#	Year	Wake	Description	Lecation
CD201	2004	DODGE	PICKUP TRUCK, QUAD CAB 4WD, 1-TON DUALLY, 5.9L CUMMINS TURBO DIESEL, AUTO, PW/PDL,	CNRL
			235/80R17 TIRES, C/W 10' SERVICE DECK, CABINETS, DIESEL/METHANOL TANKS	•
MV017	2004	MOUNTAIN	WELLSITE TRAILER, TRI/AXLE, TIRES 75R16	ED
		VIEW		
MV019	2004	MOUNTAIN	TRI-AXLE WELLSITE TRAILER, TIRES 265/75R16-STEEL DAYTONS, (LIC T17913)	ED .
		VIEW		
V061	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			445/65R22.5-ALUM BUDS, 16,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 58,000-LB AXLES, AIR	
		1	RIDE SUSPENSION, (LIC G20935) C/W CUSTOM VAC, TC SPEC 412, 18,280-LITRE CAPACITY - S/N	
			1349475-03. HIBON BLOWER	
V064	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, FRONT AXLE	ED
			65R22.5 ALUM BUDS 20,000-LB, REAR AXLES 11R24.5 ALUM BUDS 78,000-LB, AIR RIDE SUSPENSION,	
		1	C/W CUSTOM VAC SERVICES TANK, TC 412, 19,193-LITRE S/N 10781003, HIBON 820 BLOWER	
			C/W COSTOWI VAC SERVICES TANK, TO 412, 13,135-LITRE 5/N 10/81005, HIBON 820 BLOWER	
W071	2004	WESTERN STAR	STRAIGHT T/A WATER TRUCK, CAT C12 ENGINE, 18-SPEED TRANSMISSION, FRONT TIRES	ED
77071	1 2001	11231211131111	425/65R22.5-ALUM BUDS, 16,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB AXLES,	LD
			CHALMERS SUSPENSION, S/N 5KKHALLASS4M16154, C/W CUSTOM VAC SERVICES 16,528-LITRE	
			TANK, TC SPEC 412, S/N 10240703, BOWIE PUMP, 221,085-KILOMETERS INDICATED, 23,587-HOURS	
W074	2004	VA/ECTEDNI CTAD	INDICATED T/A WATER TRUCK, CAT C12 430-HP ENGINE, EATON FULLER 18-SPEED, 16,000-LB FRONT AXLE,	- FD
WU74	2004	WESTERNSTAR		ED
			FRONT TIRES 425/65R22.5, ALUM BUDS, 46,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS,	
10001	2004		C/W CUSTOM VAC STEEL TANK, S/N 10260903, 16,528 L, TC412, BOWIE PUMP	
MV004	2004	MOUNTAIN	TRIDEM WELLSITE TRAILER, 10'X30', TIRES 265/75R16, C/W LPG PIG, FRIDGE, MICROWAVE, STOVE,	ED
		VIEW	WASHROOM, SHOWER, FOLD UP STEP, RAILING	
PS118	2004	KENWORTH	S/A PRESSURE TRUCK W/ 40" SLEEPER, CUMMINS ISC 315-HP ENGINE, EATON FULLER 10-SPEED	CNRL
		-	TRANSMISSION, FRONT TIRES 11R22.5 TIRES-ALUM BUDS, REAR TIRES 11R22.5-ALUM BUDS, AIR	
			RIDE SUSPENSION, (LIC G50667), C/W HAMM'S TANK S/N 46ANM062M31029424, TC406, 2	
			COMPARTMENT - 4,045-LITRE/4,055-LITRE CAPACITY, GARDNER DENVER TRIPLEX PUMP	
D71400	2001	***************************************	A POSSELLES TROUBLE OF THE STANDARD STA	
PTI108	2004	1	S/A PRESSURE TEST TRUCK, DT 466 ENGINE, EATON FULLER 10-SPEED ENGINE, 11R22.5 TIRES, ALUM	GP
		L	BUDS, (LIC G03323), C/W TRUCK BODY S/N ITB312253, OPERATOR CONTROL CABIN, TRIPLEX PUMP	
			AND TANKS, 10,000-PSI, 40-L/M, -KILOMETERS REPORTED	
SSP404	2004	THOMAS	SKID STEER LOADER, TIRES 10-16.5, C/W FORKS, 3206-HOURS INDICATED	ED
TR001	2004	WESTERN STAR	T/A HIGHWAY TRACTOR W/ 34" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	GP
			TRANSMISSION, FRONT TIRES 425/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, (LIC G10843), C/W	
			WET KIT, SLIDING 5TH WHEEL, 424,021-KILOMETERS INDICATED, 13,093-HOURS INDICATED	
UT003	2004	SOUTHLAND	DUMP TRAILER, LIC. #T17915	GP
W075	2004	WESTERN STAR	T/A TANK TRUCK, CAT C12 430-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	CNRL
			425/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB	HORIZON
			REAR AXLES, C/W CUSTOM VAC SERVICES S/N 10270903, TC412, 16,528-LITRE CAPACITY, BOWIE 3"	PROJECT
			PRODUCTION PUMP	
W309	2004	WESTERN STAR	T/A WATER TRUCK, EATON FULLER, 425/65R22.5 FRONT, 11R24.5 REAR, ALUM BUDS, 23000-LB X 2	ED
			AXLES, (LIC G-19583), C/W CUSTOM TANK 16418L CAP, S/N 15331004	
V062	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
		1	445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 58,000-LB AXLES, AIR	
		1	RIDE SUSPENSION, (LIC G19595) C/W CUSTOM VAC SERVICES, TC SPEC 41.2, 18,355-LITRE CAPACITY -	
			IS/N 10390803. HIBON BLOWER	
V067	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB AXLES, AIR	
		[RIDE SUSPENSION, (LIC G19593) C/W CUSTOM VAC, TC SPEC 412, 19,193-LITRE CAPACITY - S/N	
	<u> </u>		11511003. HIBON BLOWER.	

Еритр. 4	Year	Make	Description	Location
V069	2004	1	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, 475-HP ENGINE, EATON FULLER 18-SPEED	ED
		1.	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 60,000-LB REAR AXLES, (LIC F32688), C/W CUSTOM VAC SERVICES VAC UNIT	
			S/N 11551103, 19,193-LITRE CAPACITY, TC412 SPEC, HIBON BLOWER, HYDRAULIC HOIST, REAR	
			DOOR, VIBRATOR, DLIMP GATE	
V085	2004		TRI-AXLE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED
			44S/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL BUDS, 56,000-LB AXLES, AIR	
			RIDE SUSPENSION, (LIC F32692), C/W CUSTOM VAC TANK, 19,193-LITRE CAPACITY - S/N 13510204	
V089	2004	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT SERIES 60 ENGINE, 515-HP, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W	İ
			CUSTOM VAC SERVICES TANK, 19,193-LITRE CAPACITY, S/N 13540304, 14.9-PSI RATING, HYDRAULIC	
			BOOM, 165,450-KILOMETERS INDICATED, 17,507-HOURS INDICATED	
MV003	2004	MOUNTAIN	TRI-AXLE WELLSITE TRAILER, TIRES 265/75R16-STEEL DAYTONS, PROPANE TANK, (LIC S561-54)	ED
		VIEW		
MV042	2004	MOUNTAIN	TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 2-BEDROOMS], TIRES 265/75R16-STEEL	ED
		VIEW	DAYTONS, PROPANE TANK, (LIC T260-43)	
W308	2004		T/A WATER TRUCK, DETROIT 60 SERIES ENGINE, EATON FULLER 18-SPEED, 18000-LB FRONT AXLE,	ED
			445/65R22.5 FRONT TIRES, ALUM BUDS, 46000-LB REAR AXLES, 11R22.5 REAR TIRES, STEEL BUDS,	
			AIR RIDE SUSPENSION, EST!MATED 16,000-LITRE CAPACITY (LIC. G10829), C/W STEEL TANK,	
			PRODUCT PUMP	
TR002	2004	W/ESTERN STAR	T/A TRACTOR W/ SLEEPER, DETROIT SERIES 60 500-HP ENGINE, EATON FULLER 18-SPEED	ZAMA CITY
111002	2.00+	WESTERN BITTO	TRANSMISSION, FRONT TIRES 425/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	EATH CIT
			11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W WET KIT, SLIDING STH	
		i		
MV018	2004		WHEEL. ROPER 4" PRODUCTION PUMP TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 1-BEDROOM], 265/75R16 TIRES PROPANE PIG,	MEDICINE
INIANTA	2004	1	• • • • • • • • • • • • • • • • • • • •	
	D11047 0004		KITCHEN, BATH, BEDROOM, (LICS891-96)	HAT
RV047	2004	HAULMARK	TRI-AXLE GOOSE NECK ENCLOSED 38' SITE TRAILER, LT 235/85R16 TIRES, ALUM BUDS, W/ REAR	MEDICINE
		<u> </u>	DOOR, FRONT LIVING QUARTERS, SIDE DOOR, GENSET, AWNINGS, A/C, (LIC T070-85)	HAT
V066	2004		TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES	ED ·
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB AXLES, AIR	
-			RIDE SUSPENSION, (LIC F32656), C/W CUSTOM VAC S/N 11491003, TCRN Z.05.437.03.04, TC 412,	
			19,193-LITRE CAPACITY, HIBON BLOWER, TIRE BOSS PRESSURE GAUGE	
V083	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	MEDICINE
			TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W CUSTOM VAC SERVICES	HAT
			TANK, 19,193-LITRE, S/N 2530104 [IN ACCIDENT ON 8/27/10; WHOLE FRONT END TO BE REPLACED	
			ALONG W/PTO, FUEL TANKS, HOSES. EXTENSIVE BODY WORK TO BE PERFORMED; CONSIDERED	
			REPAIREDL	
AT001	2004	HEIL	TRIDEM ALUM WATER TANKER, 60,000-LB AXLES, TIRES 11R24.5, STEEL BUDS, (LIC T055-68), C/W	SWIFT
			ALUM TANK, 37,854 L, TC407	CURRENT
P136	2004	CHEVROLET	PICKUP TRUCK, CREW CAB, 3/4-TON, DURAMAX DIESEL, 4WD, LEATHER SEATS (NOT IN SERVICE AT	SWIFT
			TIME OF INSPECTION]	CURRENT
TR003	2004	WESTERN STAR	PRODUCTION WATER TRUCK, BULKER, 36" SLEEPER CAB, 279,837-KILOMETERS INDICATED	MH
V065	2004	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, 475-HP, EATON FULLER 18-SPEED TRANSMISSION,	ED
VC0.5	2004	WESTERNSIAR	FRONT TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W CUSTOM VAC	
			SERVICES VAC UNIT, 19,193-LITRE CAPACITY, TC412 SPEC, S/N 11481003, C/W HIBON BLOWER,	
V068	2004	MECTERNICTAR	HYDRAULIC HOIST, REAR DOOR, VIBRATOR, DUMP GATE & KNIFE VALVE	ED
VUOR	2004	1	TRI-AXLE VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, 20,000-LB FRONT	""
		1	AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 60,000-LB REAR AXLES, REAR TIRES 11R24.5, ALUM	
			BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC STEEL TANK, 19,193-LITRE, TC412, S/N 11531003,	
1,105			HIBON BLOWER, AIR BOSS TIRE INFLATION SYSTEM	
V086	2004	WESTERN STAR	T/A VACUUM TRUCK, CAT C12 ENGINE, EATON FULLER 18-SPEED, FRONT AXLE 425/65R22.5 ALUM	ED
			BUDS 16,000-LB, REAR AXLES 11R24.5 ALUM BUDS 46,000-LB, C/W CUSTOM VAC TANK, TC350, S/N]
	l		13200204, 19,193-LITRE CAPACITY, HIBON BLOWER, 181,321-KILOMETERS INDICATED, 14,505-	
		1	2320020 17 23/233 ETTE OF THOSE DEOTECT ADJUSTED INCOME TEND IN THE CONTENT ADJUSTED ADJUSTED IN THE CONTENT ADJUSTED IN THE CONTENT ADJUSTED IN THE CONTENT ADJUSTED IN THE CONTENT ADJUSTED IN THE CONTENT ADJUSTED ADJUSTED IN THE CONTENT ADJUSTED ADJUSTED IN THE CONTENT ADJUSTED AD	3

Equip-#	Year	Make	Description	Lacation
S003	2004	Complete Com	STEAM TRUCK, 24" SLEEPER CAB, EATON FULLER 13-SPEED TRANSMISSION [NEW STEAMER ADDED	MH
3003	200-1	TEST EITH STAIN	IN 2015]	17111
V063	2004	WESTERN STAR	TRI-AXLE VACUUM TRUCK, CAT ENGINE, EATON FULLER 3-SPEED TRANSMISSION, FRONT TIRES	ĘD
¥003	2004	1123,211113,711	445/65R22,5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 54,000-LB AXLES, AIR	45
			RIDE SUSPENSION, C/W CUSTOM VAC, TC SPEC 412, 19,193-LITRE CAPACITY, S/N 10771003, HIBON	
V088	2004	WESTERN STAR	BLOWER. 146.650-KILOMETERS INDICATED. 18.573-HOURS INDICATED TRI-DRIVE VACUUM TRUCK, CAT 3637 ENGINE, EATON FULLER 13-SPEED TRANSMISSION, FRONT	МН
VV60	2004	WESTERN STAR	TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 57,000-LB	IVIE 1
			· · · · · · · · · · · · · · · · · · ·	
			AXLES, AIR RIDE SUSPENSION, (LIC F32681) C/W CUSTOM VAC, TC SPEC 412, 18,262-LITRE CAPACITY	
N 40 (00.7	2004	ACCUNITAIN	S/N 13210204. H/BON BLOWER	CD
MV007	2004	MOUNTAIN	WELLSITE TRAILER, TRI/AXLE, TIRES 75R16	GP
LITTECO 4.4	2004	VIEW	VALOOF AALTAA INDIAA AVII E OCEICE TRAILER DOE TOOME ALLIAAAA GE CIDE DOOR OSAR RADAL	
UTC314	2004	AGASSIZ	V-NOSE 14' TANDEM-AXLE OFFICE TRAILER, 225/75R15, ALUM MAGS, SIDE DOOR, REAR BARN	
		JUNCTION	DOORS, HEATER, CARRIER AIRV AC, C/W SUNMAR EXCEL TOILET, SYLVANIA MW, BUILT-IN	
			DESK/COUNTER, SOFA, OFFICE CHAIR, (LIC W060-64)	
UTC316	2004	SL TRAILERS	TANDEM-AXLE 16' FLATDECK TRAILER, TIRES 235/80R16-STEEL BUDS, (LIC W060-65)	
MV041	2004	MOUNTAIN	TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 1-BEDROOM], TIRES 75R16	ED
		VIEW		
AT002	2004	HEIL	TRIDEM WATER TANKER, TIRES 11R24.5, STEEL BUDS, AIR RIDE SUSPENSION, S/N	ED
			5HTDL423245J74948, C/W 1 COMPARTMENT ALUM TANK, 37,854 L, TC407	
MV012	2004	MOUNTAIN	WELLSITE TRAILER, TRI/AXLE	MEDICINE
		VŧEW		HAT
MV014	2004	MOUNTAIN	TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 1-BEDROOM], TIRES 75R16	MEDICINE
		VIEW		HAT
UTK010	2004	JASPER	UTILITY TRAILER	
CSC001	2005	N/A	SIDE SPOOLER CUSTOM BUILT TANDEM-AXLE TRAILER MOUNTED, 215/75R17.5 TIRES, 20000-LB	ED
			AXLES, C/W 1978 INTERNATIONAL DT466 200-HP DIESEL ENGINE, 2018-HOURS INDICATED	
PC014	2005	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE, 3/4-TON, 288,892-KILOMETERS INDICATED	ED
PF303	2005	FORD	PICKUP TRUCK, LARIAT KING RANCH QUAD CAB, 4X4, TURBO DIESEL ENGINE, TIRES 285/65R18,	ED
			FULLY LOADED, SHORT BED, (LIC G-19621), C/W ALUM TOOL BOX	
ST202	2005	INTERNATIONA	SINGLE AXLE SERVICE TRUCK, SLEEPER, EATON FULLER 10-SPEED TRANSMISSION, 10,433-LB. FRONT	CNRL
		L	AXLE, 23,000-LB. REAR AXLES, SERVICE BODY W/STORAGE BOXES, WELDER/GENERATOR, AIR	
		_	COMPRESSOR, AUTO CRANE 22' BOOM CRANE W/ 3,200-LB CAPACITY, 181,613-KILOMETERS	
			INDICATED, 8,910-HOURS INDICATED	
V502	2005	WESTERN STAR	VACUUM TRUCK, TRI-AXLE, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ED
1302	2005	Western Stylin	CUSTOM VAC 19,193-LITRE TANK S/N 15230704, HYDRAULIC BOOM, HIBON BLOWER, -KILOMETERS	
			INDICATED, 22,310-HOURS INDICATED	
V503	2005	MESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT SERIES 60 575-HP ENGINE, EATON FULLER 18-SPEED	ED
V 303	2005	VVCSTERRY STAR	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	LD
			· · · · · · · · · · · · · · · · · · ·	
			11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES	
			S/N 15240704, TC407/412, 19,193-LITRE CAPACITY, HIBON BLOWER, 138,076-KILOMTERES	
1504	2005	WESTERN STAR	REPORTED 15 493-HOURS REPORTED	
V504	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT DIESEL SERIES 60 ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL	
			BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION, (LIC F24231), C/W CUSTOM VAC- S/N 15250704.	
		ļ <u>.</u>	19.193-HOURS INDICATED	
V505	2005	WESTERN STAR	VACUUM TRUCK, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	ED
			TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB	
			AXLES, S/N 5KKPALCK25PW69021, C/W CUSTOM VAC SERVICES TANK, 19,193-LITRE, S/N 15260704	
	2005	WESTERN STAR	VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 15,777-KILOMETERS	ED
V506	2005	TALEST ENTRE STATE		
V506	2005		INDICATED, 11,576-HOURS INDICATED, FRONT TIRES 445/65R22.5 ALUM BUDS, REAR TIRES 11R24.5	
V506	2005		INDICATED, 11,576-HOURS INDICATED, FRONT TIRES 445/65R22.5 ALUM BUDS, REAR TIRES 11R24.5 ALUM BUDS-10, W/ SLEEPER, TIRE PRESSURE MONITOR, ALUM CABINETS, C/W CUSTOM VAC UNIT	-



Equip. #	Vear	Make	Description	Locatioi
V508	2005		TRI-AXLE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
	i .		11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
			SERVICES S/N 15751204, TC407/412, 19,193-LITRE CAPACITY, HIBON BLOWER, 24,375-KILOMETERS	
			INDICATED, 4.018-HOURS INDICATED	
V511	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT DIESEL SERIES 60 ENGINE, EATON FULLER 18-SPEED	ED
VJII	2005	WESTERN STAN	TRANSMISSION, FRONT 20,000-LB AXLE, REAR 61,000-LB AXLES, (LIC F49363) S/N	
			SKKPALCK65PV23785, C/W CUSTOM VAC TCRN Z.05.437.03.04 TC SPEC 407/412, 19,193-LITRE	
			, ,	
14557	3005	WESTERN STAR	CAPACITY, 144.120-KILOMETERS INDICATED, 18.922-HOURS INDICATED TRI-AXLE VACUUM TRUCK, DETROIT DIESEL, EATON FULLER 18-SPEED, 445/65R22.5 FRONT, 11R24.5	ED
V517	2005			ED
			REAR, ALUM BUDS, AIR RIDE, 20400-LB X 3 AXLES, TIRE AIR PRESSURE CONTROL, (LIC F-49375), C/W	
		ļ	CUSTOM VAC UNIT, HIBON VTB820, TC407/412, 19193 L CAP, S/N 15731104	
V518	2005	WESTERN STAR	TRI-DRIVE VAC TRUCK, DETROIT SERIES 60 DIESEL ENGINE, 515-HP, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W	
			CUSTOM VAC SERVICES VAC UNIT, 18,262-LITRE CAPACITY, S/N 17831204, C/W HIBON BLOWER,	
	1		HYDRAULIC HOIST, REAR DOOR, VIBRATOR, DUMP GATE, 167,730-KILOMETERS INDICATED, 12,140-	
	1		HOURS INDICATED	
W302	2005	WESTERN STAR	TRI-AXLE TANK TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	GP
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
			, ,	
		[SERVICES S/N 10290404, TC407/412, 20,123-LITRE CAPACITY, BOWIE 4" PRODUCTION PUMP	
W304	2005	MECTEDNICTAD	PRODUCTION WATER TRUCK, BULKER	ED
W504	2003	AA EQ LEVIA 2 LAK	PRODUCTION WATER TROCK, BOLKER	LU
WINE	2005	WESTERNI STAR	TRI-DRIVE WATER TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ED
W305 2	2005	WESTERN STAR	l ' '	LL
			FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,200-	
			LB AXLES, AIR RIDE SUSPENSION, (LIC F24219), C/W CUSTOM VAC S/N 10320504, TC 407/412, 20,123	
			LITTE CAPACITY	ED
W307	2005	WESTERN STAR	TRI-AXLE WATER TRUCK, DETROIT 60 SERIES ENGINE, EATON FULLER 18-SPEED, 20000-LB FRONT /	ED
			20400-LB X 3 REAR AXLES, FRONT 445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE	
			SUSPENSION, C/W CUSTOM VAC SERVICES STEEL TANK, TC 406, 21123 L CAPACITY S/N 10340604	
	<u> </u>	1		
W310	2005	1	WATER TRUCK, TANDEM-AXLE, EATON FULLER 18-SPEED TRANSMISSION, CUSTOM VAC TANK S/N	ED
			15341004, 16,418-LITRE CAPACITY, 16,967-KILOMETERS INDICATED, 1,900-HOURS INDICATED	
W312	2005		TANDEM-AXLE WATER TRUCK, CAT, EATON FULLER 18-SPEED, 385/65R22.5 FRONT, 11R24.5 REAR,	ED
			ALUM BUDS, AIR RIDE, 23000-LB X 2 AXLES, (LIC F-65461), C/W ADVANCE WATER TANK, 16000-LITRE	
			CAPACITY, 134,597-KILOMETERS INDICATED, 18,931-HOURS INDICATED	
W313	2005	WESTERN STAR	TRI-DRIVE WATER TRUCK, DETROIT DIESEL, 425/65R22.5 FRONT, 11R24.5 REAR, ALUM BUDS, AIR	ED
	-	1	RIDE, 19400-LB X 3 AXLES (LIC F-65474), C/W CUSTOM WATER TANK 20123L CAP, TC407, S/N	
			19990505	
W314	2005	STERLING	T/A WATER TRUCK, DETROIT 60 SERIES DIESEL ENGINE, EATON FULLER 18-SPEED, 13000-LB FRONT /	ED
			20000-LB X 2 REAR AXLES, FRONT 11R24.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, ESTIMATED	
			15,000-LITRE, FLOAT TIRES	
AT005	2005	HEIL	TRIDEM WATER TANKER, TIRES 11R24.5-STEEL BUDS, AIR RIDE SUSPENSION, C/W FIBERGLASS	GP
WIOOD	2003	1 '''''	COATED TANK	C,
HOKOAZ	2005	WICETON CTAD	HOT OIL TRUCK, 186,994-KILOMETERS INDICATED, 5,249-HOURS INDICATED	GP
HOK047	2005	ANESTELIN STAK	THOT OIL TROCK, 100,334+NILOWIETERS INDICATED, 3,243-HOURS INDICATED	GP.
LILIODA	2000	(NITEDALATIONA)	T/A FILED DEATED HAD TANACE FEDER CHAMMANICION FAIGNE FATON FULLED 40 CDFFD	GP
HU001	2005		T/A FLUID HEATER UNIT W/ SLEEPER, CUMMINS ISM ENGINE, EATON FULLER 18-SPEED	GP
		L	TRANSMISSION, FRONT TIRES 12R24.5-ALUM BUDS, 14,000-LB AXLE, REAR TIRES 11R24.5-STEEL	
			BUDS, 40,000-LB AXLES, (LIC H-26755), C/W CUSTOM WELDING TANK- S/N TN00038-2144 MDIN	
			D000190, 8,840-LITRE CAPACITY TANK, 8,000,000- BTU HEATER, 3 COMPARTMENT TANK 1,400-	
			LITRE 1.400-LITRE AND 6.040-LITRE CAPACITY	
PR206	2005	DODGE	PICKUP TRUCK, QUAD CAB, 4WD, TURBO DIESEL ENGINE,179,590-KILOMETERS INDICATED	ED
V521	· 2005	WESTERN STAR	VACUUM TRUCK, TRI-DRIVE	· GP

Eguip, 8	Young	na i a	Description	s server viere
CB904	2005		TRI-DRIVE COMBO TRUCK, DETROIT 60 SERIES 14L 515-HP ENGINE, EATON FULLER 18-SPEED, FRONT	ED
LB904	2005	1	TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC UNIT, S/N	בט
	·			
- W205	2005	LUECTEDAL CTAD	16210904, 12,682-LITRE	CNDI
W306	2005	WESTERN STAR	TRI-DRIVE WATER TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-	CNRL
			ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,000-LB AXLES, AIR RIDE	HORIZON
			SUSPENSION, (LIC F24220), C/W CUSTOM VAC SERVICES LTD TANK, TC SPEC 406, 21,123-LITRE	PROJECT
		<u> </u>	CAPACITY - S/N 10330604	
V513	2005	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, EATON FULLER 18-SPEED TRANSMISSION, AIR RIDE SUSPENSION, C/W	ED
			CUSTOM VAC TANK	
V522	2005	1	TRI-AXLE VACUUM TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-	ED
			ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,000-LB AXLES, AIR RIDE	, i
			SUSPENSION, (LIC F24220) C/W CUSTOM VAC, TC SPEC 407/412, 19,193-LITRE CAPACITY -	
			SN17841204. HIBON BLOWER	
V523	2005	1	TRI-AXLE VACUUM TRUCK, DDC 60 SERIES ENGINE, EATON FULLER 18-SPEED, 20,000-LB FRONT	ED
			AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR AXLES, 11R24.5 REAR TIRES, ALUM	
i			BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC S/N 17871204, 19,193-LITRE CAPACITY, TC	
• •			407/412. ROBUSCHI BLOWER	
RN063	2005	ROUGHNECK	T/A 28' RV TRAILER, TIRES 235/80R16, COMPLETE	ED
PR203	2005	FÓRD	PICKUP TRUCK, CREW CAB, 3/4-TON, SERVICE TRUCK	ED
PTF104	2005	INTERNATIONA	S/A PRESSURE TEST TRUCK W/ SLEEPER, DT466 ENGINE, EATON FULLER 6-SPEED TRANSMISSION,	ED
		L	FRONT TIRES 1R22.5 TIRES-ALUM BUDS, REAR TIRES 11R22.5-ALUM BUDS, AIR RIDE SUSPENSION,	
			C/W INTERCONTINENTAL 16' TRUCK BODY S/N 6575, TRIPLEX PUMP, OPERATOR CONTROL ROOM	
		j	INOT IN SERVICE ON DATE OF INSPECTION]	
RN053	2005	ROUGHNECK	TRAILER, CREW SHACK	MH
RN056	2005	ROUGHNECK	T/A RV TRAILER, (LIC 1537-40), C/W PROPANE PIG	ED
RN065	2005	ROUGHNECK	TANDEM-AXLE RV TRAILER, W/ PINTLE HITCH EXTENSION, PIG, 100-LB	ED
MV045	2005	MOUNTAIN	TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 2-BEDROOMS AND CENTRAL AIR], TIRES	ED
		VIEW	265/75R16, STEEL DAYTONS, PROPANE TANK	
RN062	2005	ROUGHNECK	T/A RV TRAILER, 235/80R16 TIRES, KITCHEN, BED, 100-LB PROPANE PIG	ED
	•			
NP001	2005	WESTERN STAR	TRI-AXLE NITROGEN PUMP UNIT TRUCK, SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-	ED
			SPEED, FRONT TIRES 425/65R22.5, ALUM BUDS, REAR TIRES11R24.5, ALUM BUDS, AIR RIDE	
			SUSPENSION, C/W HYDRA RIG NITROGEN PUMP UNIT, TC 338, TCRN Z-05-507-06-05 S/N	
			HRC180K10KHR110003-1, WESSINGTON TANK TC338 S/N 460, CAIN HEAT RECOVERY SYSTEM,	
	<u>'</u>		TRIPLEX PUMP, CONTROL CAB, 364,745-KILOMETERS INDICATED, 14,874-HOURS INDICATED	
			,	
NP002	2005	WESTERN STAR	TRI-AXLE NITROGEN PUMP UNIT, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, 20000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61200-LB REAR	1
	1	,	AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, SLEEPER, (LIC G19596), C/W HYDRA	l
		İ	RIG NITROGEN PUMP UNIT, CAIN HEAT RECOVERY SYSTEM S/N 4183, WESSINGTON TANK ID# 464,	l
		i	TC338 TANK, CONTROL CAB, BACKEND S/N HRC180K10KHR110004-2	
		<u> </u>	· · · · · · · · · · · · · · · · · · ·	
NP004	2005	WESTERN STAR	TRI-AXLE NITROGEN PUMP UNIT TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, FRONT	ED
		1	TIRES 445/65R22.5 ALUM BUDS, REAR TIRES 11R24.5 ALUM BUDS, AIR RIDE SUSPENSION, C/W	1
		1	HYDRA RIG N2 PUMPER UNIT TC 338, TCRN Z-05-507-06-05 S/N HRC180K10KHR110004-3,]
		1	WESSINGTON CRYOGENICS TANK S/N 448. SLEEPER	
NT001	2005	WESTERN STAR	TRI-AXLE NITROGEN TRANSPORTER TRUCK W/ SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-	ED
		1	SPEED, 20,000-LB FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES,	1
			REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC F65500), C/W HYDRA RIG NITROGEN	
		1	TANK SETUP, WESSINGTON CRYOGENIC TANK, 15,897 L, TC338 TANK, MODEL RTV 4200 S/N 465,	
			1352 760-KH OMETERS INDICATED 8 180-HOURS INDICATED	
	1	<u> </u>	1332 ADEAN COMPLEAS INDICATED A TALEBOOKS INDICATED	1

Enlith:	Year	Make	Description	Laterifica
CB901	2005	KENWORTH	TRI-DRIVE COMBO UNIT, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, 19705-LB FRONT AXLE,	RL.
			445/65R22.5 FRONT TIRES, ALUM BUDS, 58000-LB REAR AXLES, 11R24.5 REAR TIRES, ALUM BUDS,	
			NEWAY AIR RIDE SUSPENSION, SLEEPER, (LIC F96606), C/W CUSTOM VAC SERVICES COMBO UNIT	
			S/N 16421004, 13,147-LITRE CAPACITY, TC407/412 DUAL COMPARTMENT TANK, ROBUSCHI	
			BLOWER, HOTSY STEAMER, WATER BLAST, HYDRAULIC BOOM, DUMP GATE VALVES	
PS120	2005	FORD	PRESSURE TRUCK, TESTER	RL
PTI102	2005	FORD	SUPER DUTY 4X4 PRESSURE TEST TRUCK, POWER STROKE DIESEL ENGINE, 6-SPEED TRANSMISSION,	RL
	1		FRONT TIRES 225/70R19.5-ALUM BUDS, REAR TIRES 225/70R19.5-STEEL BUDS, SPRING SUSPENSION,	
			(LIC G03331), C/W INTERCONTINENTAL 16' TRUCK BODY S/N ITBO111279,2 STAINLESS STEEL TANKS,	
			3 ALUM TANKS, NATIONAL 330 3000-PS! TRIPLEX PUMP S/N 12616-D, OPERATOR CONTROL ROOM	
CB905	2005	WESTERN STAR	TRIDRIVE COMBO VACUUM TRUCK, DDC 60 490-HP ENGINE, AUTOMATIC TRANSMISSION, FRONT	GP
00000	2005	THE STEIN STAN	TIRES 445/65R22.5-ALUM BUDS, 18,000-LB FRONT AXLE, REAR TIRES 11R24.5-AUMINUM BUDS,	ui ui
		1	57,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES TANK S/N 16230105,	
			TC406/412, 2 COMPARTMENT - 10,146-LITRE CAPACITY, HIBON BLOWER, WATERBLAST WASHER	
			10400/412, 2 COMPARTIMENT - 10,140-LITRE CAPACITY, FIBON BLOWER, WATERBLAST WASHER	
CB906	2005	WESTERN STAR	TRIDRIVE COMBO VACUUM TRUCK, DDC 60 515-HP ENGINE, AUTOMATIC TRANSMISSION, FRONT	GP
	1		TIRES 445/65R22.5-ALUM BUDS, 18,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-	
			LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES STEEL TANK S/N 16240205,	
			TC407/412, 2 COMPARTMENT - 12,682-LITRE CAPACITY, HIBON BLOWER, WATERBLAST HEATER	
MV044	2005	MOUNTAIN	TRIDEM WELLSITE TRAILER [LIVING QUARTERS W/ 1-BEDROOM], TIRES 265/75R16-STEEL DAYTONS,	AAFDICINE
1414044	2005	VIEW	PROPANE TANK, (LICT260-45)	MEDICINE HAT
RN057	2005		T/A RV TRAILER, TIRES 235/80R16-STEEL BUDS, PROPANE PIG, (LIC T537-37)	MEDICINE
			, , , , , , , , , , , , , , , , , , , ,	HAT
RN058	2005	ROUGHNECK	T/A RV TRAILER, CREW SHACK	MEDICINE
				HAT
ST203	2005	STERLING	S/A SERVICE BODY TRUCK, ACTERRA MERCEDES BENZ ENGINE, STANDARD TRANSMISSION, FRONT	MEDICINE
			TIRES 11R22.5-STEEL BUDS, 12,000-LB AXLE, REAR TIRES 11R22.5-STEEL BUDS, 21,000-LB AXLE, (LIC	HAT
			G19600), C/W BRUTUS SERVICE DECK, STORAGE COMPARTMENT, TIGER MODEL 6541E 4-TON	
. TR004	2005	VALECTE DAL CTAD	CRANE, AIR COMPRESSOR	MEDICINE
. IKUU4	2005	WESTERN STAR	T/A TRACTOR, 24" SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT	MEDICINE
			TIRES 65R22.5-ALUM BUDS, 16,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB AXLES, 372,058-KILOMETERS INDICATED	HAT
TR006	2005	WESTERN STAR	T/A TRACTOR, SLEEPER, DETROIT SERIES 60 ENGINE, 515-HP, EATON FULLER 18-SPEED	MEDICINE
111000	1005	THE STERM STAR	TRANSMISSION, 367,108-KILOMETERS INDICATED	HAT
TWP006	2005	ADVANCE	TRIDEM VAC STIFF POLE PUP TRAILER, 24,000-LITRE, TIRES 11R24.5-ALUM BUDS, AIR RIDE	MEDICINE
			SUSPENSION	HAT
V514	2005	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED, FRONT TIRES 445/65R22.5,	MEDICINE
			ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W 20005 CUSTOM VAC TANK, S/N 17821204,	HAT
		4	17,333 L, TC407/412	
V524	2005	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 DIESEL ENGINE, 515-HP, EATON FULLER 18-SPEED	MEDICINE
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, REAR TIRES 11R24.5-ALUM BUDS, C/W	HAT
		1	CUSTOM VAC SERVICES VACUUM TRUCK, 18,252-LITRE CAPACITY, S/N 17890105, C/W HIBON	
110422	2005	KENNADDELL	BLOWER, REAR DOOR, VIBRATOR, DUMP GATE	
HR123	2005	KENWORTH	TRI-DRIVE HYDRO-VACUUM TRUCK, INTEGRAL SLEEPER, CAT C15 ENGINE, EATON FULLER 18-SPEED	SWIFT
			TRANSMISSION, 20,000-LB FRONT AXLE, 425/65R22.5 FRONT TIRES, ALUM BUDS, 58,000-LB REAR	CURRENT
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G03328), C/W WESTECH VAC	
			SYSTEMS S/N WT0604934, 13,756-LITTE CAPACITY, HIBON BLOWER, HOTSY, DUMP GATE VALVES,	
			30,009-KILOMETERS INDICATED, 2,191-HOURS INDICATED	
Q003	2005	CUSTOM VAC	QUAD WAGON VAC TANK TRAILER, TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK S/N	SWIFT
		SERVICES	18850405, 21,820-LITRE	CURRENT
SVT001	2005	I .	QUAD WAGON VAC TANK TRAILER, TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK S/N	SWIFT
		SERVICES	18850405, 21,820-LITRE	CURRENT



		Viake	Description	- kecalilan
Equip. 9.			TRAILER, VACUUM TRAILER, TRI-AXLE, W/ PROGRESSIVE TANK 4,230-GALLON VACUUM TANK S/N V-	SWIFT
TWP015	2005	N/A		
TUDEON	2005	ADMANCE	21911C (2006), DOT 407/412, CUSTOMVAC BLOWER	CURRENT
TWPS021	2005	ADVANCE	ADVANCE TRIDEM OIL PUP, TIRES 285/75R24.5, ALUM BUDS, 15,900-LITRE CAPACITY	SWIFT
	2005	144555555155515	ALACHUMA TO LOW TO LOW TO LOW OUT ON A VACUABILITY AND ONCE TAKE OF THE	CURRENT
V515	2005	WESTERNSTAR	VACUUM TRUCK, TRI-DRIVE, CUSTOM VACUNIT, 18,000L [NOT IN SERVICE AT TIME OF INSPECTION]	МН
V516	2005	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 DIESEL ENGINE, 515-HP, EATON FULLER 18-SPEED	SWIFT
			TRANSMISSION, C/W CUSTOM VAC SERVICES VACUUM TRUCK	CURRENT
V519	2005	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-	SWIFT
			ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL BUDS, 61,000-LB AXLES, AIR RIDE	CURRENT
			SUSPENSION, (LIC F49396), C/W CUSTOM VAC TANK, TCRN Z.05.437.03.04, TC SPEC 407/412, 18,262-	
		•	LITRE CAPACITY	
VR103	2005	KENWORTH	T/A VACUUM TRUCK, CAT C15 ACERT 475-HP ENGINE, EATON FULLER 18-SPEED, 18740-LB FRONT /	MH
			23000-LB X 2 REAR AXLES, FRONT 385/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, NEWAY AIR	
•			RIDE SUSP, C/W PROXY WELD INDUSTRIES 13033-LITRE TANK, TC SPEC 407, S/N 04-0757, HIBON VTB-	
			820 BLOWER, INOT IN SERVICE AT TIME OF INSPECTION	
WTRS119	2005	MACK	T/A TANK TRUCK, MACK ENGINE, 18-SPEED, FRONT TIRES 11R24.5, ALUM BUDS, REAR TIRES	SWIFT
			11R24.5, ALUM BUDS, C/W ADVANCE TANK S/N 2AESTHBOX4S000206, 15,900-LITRE, 349,305-	CURRENT
			KILOMETERS INDICATED, 15,379-HOURS INDICATED	
PC010	2005	FORD	SPORT UTILITY VEHICLE, 4WD, TURBO DIESEL ENGINE V10	ED
			·	
P146	2005	CHEVROLET	CREW CA8 PICK UP, DURAMAX ENGINE, AUTO, TIRES 245/75R16	SC
1140	2005	CHEVROLET	CNEW CAS FICK OF, DORANNA ENGINE, ACTO, TINES 243/13R10	30
CT205	2005	KENWORTH	TANDEM-AXLE STEER/TRI-AXLE DRIVE COILED TUBING UNIT, CAT C15 550-HP ENGINE, EATON	ED
			FULLER 18-SPEED, 32,000-LB FRONT AXLES, FRONT TIRES 425/65R22.5, ALUM BUDS, 69,000-LB REAR	
		1	AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, SPICER 792, C/W COILED TUBING	
			UNIT, 3100M OF 1 1/4" TUBING, HYDRA RIG RT25 INJECTOR - 3/4" TO 2", CAT 4517 TRIPLEX PUMP,	
			50L/MIN @ 21 MPA, W/NORTHWEST MODEL 5450 COMPRESSOR, 18M3 @ 14 MPA, W/TEREX	
			ATLAS 145CS-A4 14-TON PICKER, 6.7M REACH, S/N 100104 (LIC G-20899)	
. T.O.O.	2005	(15)		NACO LOGINE
AT004	2005	HEIL	TRIDEM WATER TANKER, 10,000-GALLON TANK, TIRES 11R24.5-STEEL BUDS	MEDICINE HAT
AT006	2005	HEIL	TRIDEM ALUM WATER TRAILER, 37,854-LITRE CAPACITY, TIRES 11R24.5-STEEL BUOS, (LIC T894-75)	MEDICINE
711000	1000	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	THIS ENTITION THE CONTROL OF THE GRANT OF THE BELLE IN STREET OF THE STR	HAT
V507	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED, FRONT	ED
1301	2003	TVESTEIN SIAN	AXLE 445/65R22.5 ALUM BUDS, 20,000-LB, REAR AXLES 11R24.5 ALUM BUDS, 69,000-LB, AIRLINER	2.5
			AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES 19,193 L CAPACITY, TC SPEC 407/412, S/N	
	1			
			15741204, HIBON VTB-820 BLOWER, S/N N/A, TIRE PRESSURE CONTROL SYSTEM, HYDRAULIC	
CB903	2005	WESTERNISTAR	BOOM. 144.056-KILOMETERS INDICATED. 20.542-HOURS INDICATED TRI-AXLE COMBO TRUCK, DAY CAB, DETROIT 60 SERIES, 14L 515-HP ENGINE, AUTOMATIC	MH
CB303	2005	WEST ENIN STAIN	TRANSMISSION, 18,000-LB. FRONT AXLE, 18,850-LB. REAR AXLES, CUSTOM VAC TANK S/N 16220105	14117
			1	
FKL002	2005	DAEWOO	W/ 13,147-LITRE CAPACITY LIFT TRUCK, LPG, 4450-LB CAPACITY, 3-STAGE, 186" MAST, SIDE-SHIFT, SOLID TIRES, 956-HOURS	MEDICINE
FKLUUZ	2005	DAEWOO		
D1 F1	2005	CHEVROLET	INDICATED CREW CAB 4X4 S/B PICKUP, DURAMAX DIESEL, AUTO, TIRES 265/75R16, FULLY LOADED, C/W TIDY	HAT
P151	2005	LUEAKOTEI		
: DNIOCO	2005	ROUGHNECK	TANK, ALUM BOX, 349,302-KILOMETERS REPORTED AS OF AUGUST 2011 T/A RV TRAILER, (LIC T537-66), C/W PROPANE PIG	
RN060	2005	NOOGHNECK	T/A AV TIMILLIS (LIC 1337-00), C/W PROPANE FIG	
ST201	2005	GMC	S/A 4X4 SERVICE TRUCK, DURAMAX DIESEL ENGINE, AUTOMATIC TRANSMISSION, FRONT TIRES	
		1	245/70R19.5, REAR TIRES 245/70R19.5, C/W SERVICE BODY, SERVICE TRUCK INTERNATIONAL TIGER	
		1	PICKER, 10,000-LB CAPACITY, S/N 1688 35 FH, 269, 065-KILOMETERS INDICATED (NOT IN SERVICE	
	<u> </u>		ON DATE OF INSPECTION1	
UTC310	2005	TITAN	TANDEM-AXLE 12' UTILITY TRAILER, TIRES 285/85R16-STEEL BUDS, (LIC W060-63)	



Big Eagle Limited Partnership Asset List* ASSE LIX

Equip. #	Year	Make	Description	Location
V509	2005	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, 18-SPEED TRANSMISSION, FRONT	
			TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 69,000-LB	
	!		AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES 19,193-LITRE TANK, TC SPEC	
			407/412, S/N 15761204, HIBON VTB-820 BLOWER S/N N/A, TIRE PRESSURE CONTROL SYSTEM	
	1			
V510	2005	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL, 445/65R22.5 FRONT, 11R24.5 REAR, ALUM BUDS, AIR	
			RIDE, (LIC F-49380), C/W CUSTOM VAC UNIT, HIBON VTB820, TC407/412, S/N 15771204	
	İ		(10 E) (210 F 13300), 0, 17 003 70 11 11 10 011 17 10 020, 10 10 1, 122, 0, 11 22 1	
V525	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, SLEEPER, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED,	•
V 323	2003	TT LOTE MIN STAIN	20,000-LB FRONT AXLE, 23,000-LB X 3 REAR AXLES, FRONT 445/65R22.5 ALUM BUDS, REAR 11R24.5	
			ALUM BUDS, AIRLINER AIR RIDE SUSP, C/W CUSTOM VAC SERVICES LTD. 18,262-LITRE VAC TANK,	
			TC407/412, S/N 17920105, HIBON VTB-820 BLOWER, TIRE PRESSURE CONTROL SYSTEM, 161,903-	
			KILOMETERS INDICATED, 17 106-HOURS INDICATED	
W301	2005	WESTERN STAR	TRI-DRIVE TANK TRUCK, DETROIT SERIES 60 515-HP ENGINE, 18-SPEED TRANSMISSION, FRONT TIRES	
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 69,000-LB AXLES,	
			AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES LTD. 20,123-LITRE TANK, S/N	
			10280404, BOWIE PUMP	
W303	2005	WESTERN STAR	TRI-AXLE TANK TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM	
	1	1	BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION,	
			(LIC F38542), 166,271-KILOMETERS INDICATED, 14,193-HOURS INDICATED	
CC205	2005	CELTIC PRIDE	TANDEM-AXLE CATCH TANK, 10M3, AXLE RATING 10000LB X 2, 7.50R16, (LIC W060-61)	
COLOS	2000	0227707777	, , , , , , , , , , , , , , , , , , , ,	
AT003	2005	HEIL	TRIDEM ALUM 407 WATER TANKER, 37,854-LITRE CAPACITY, TIRES 11R24.5-STEEL BUDS, 60,000-LB	MEDICINE
ATOOS	2003	***************************************	AXLES, AIR RIDE SUSPENSION, (LIC T482-15)	HAT
DC110	2005	FOCICUTE INFO	PUMP TRUCK W/ SLEEPER, S/A, CAT C7 ENGINE, AUTOMATIC TRANSMISSION, FRONT TIRES 11R22.5-	GP
PS119	2005	FREIGHTLINER		GF.
			ALUM BUDS, 16,000-LB AXLE, REAR TIRES 11R22.5-ALUM BUDS, 23,000-LB AXLE, (LIC H-26758), C/W	
			HAMMS TANK, 2 COMPARTMENT 4,220-LITRE CAPACITY , 2,325-LITRE CAPACITY - S/N	
	1		46ANM062M80842815. MAWP 3.0-PSI. WABASH PUMP, 6000-PSI	
QR127	2005	ADVANCE	TRIDEM STIFF POLE VAC TANK TRAILER, 11R22.5 TIRES, ALUM BUDS, C/W WESTECH VAC SYSTEMS	SC
			S/N WT0705046, 14,600-LITRE CAPACITY TC407 TANK	
TWP007	2005	CUSTOM VAC	TRIDEM STIFF POLE TANK TRAILER PUP, TIRES 11R24.5-ALUM BUDS, C/W CUSTOM SERVICES VAC	
		SERVICES	STEEL TANK S/N 25610906, TC406, 20,123-LITRE CAPACITY	
V512	2005	WESTERN STAR	VACUUM TRUCK, TRI-DRIVE	GP
V520	2005	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, SLEEPER, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED,	
			20000-LB FRONT AXLE, 20400-LB X 3 REAR AXLES, FRONT 445/65R22.5 ALUM BUDS, REAR 11R24.5	
			ALUM BUDS, AIR RIDE SUSP, C/W CUSTOM VAC UNIT, HIBON BLOWER, AIR PRESSURE MONITOR	
			THEORY BODD, ANT REED ON FOR THE COST ON THE COST OF T	
V526	2005	WESTERN STAR	TRI-AXLE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	GP
*320	2003	WESTERN STAN	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	Ģ.
			BUDS, 61,200-LB AXLES, AIR RIDE SUSPENSION, (LIC F65514), C/W CUSTOM VAC S/N 17930105,	
			TCRN Z.05.427.03.04, TC 407/412, 18,262-LITRE CAPACITY, HIBON BLOWER	
	-			
RN106	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, PROPANE PIG	ED
P198	2006	DODGE	PICKUP TRUCK, 3/4-TON, CREW CAB, 4WD, FUEL PUMP, TURBO DIESEL ENGINE, SUNROOF, 155,813-	ED
		1	KILOMETERS INDICATED	
RN071	2006	ROUGHNECK	T/A RV TRAILER, PROPANE PIG, (LIC V068-29)	ED
	1			
RN092	2006	ROUGHNECK	XTREME DUTY TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG	ED
RN094	2006	N/A	RV TRAILER, TIRES 235/80R16, W/ OAK CABINETS, OAK LAMINATE, VINYL FLOORING, FRIDGE,	ED
111054		1 .,,,,	STOVE, BATHROOM, KITCHEN	
				EB
DNIOGE	2006	POHCHNECK	LIVANIENCEVALE BALLBANIER III. 136786618 11862 BRIDANIE DAZ KITULIENI BALLBANIENI DELIDITARE	
RN095	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, 20', 235/85R16 TIRES, PROPANE PIG, KITCHEN BATHROOM, BEDROOM	ED
RN095 RN096	2006	ROUGHNECK	T/A 28' RV TRAILER, TIRES 235/80R16, COMPLETE	ED

Equipati	Vasi	Make	Description	loesion :
RN099	2006		XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG	ED
	<u>.</u>			
RN101	2006	ROUGHNECK	T/A RV TRAILER, CREW SHACK	ED
RN109	2006	N/A	RV TRAILER, TIRES 235/80R16 STEEL BUDS, 30' LENGTH	ED
RN111	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, 20', 235/85R16 TIRES, PROPANE PIG, KITCHEN BATHROOM, BEDROOM	ED
	2000			
V529	2006	WESTERN STAR	VACUUM TRUCK, TRI-DRIVE	ED
V531	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, 18-SPEED, 71217-KILOMETERS	ED
			INDICATED, 6929-HOURS INDICATED, 20000-LB FRONT AXLE, 23000-LB X 3 REAR AXLES, FRONT	
			445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIRLINER AIR RIDE SUSP, C/W CUSTOM VAC	
			SERVICES 19193-LITRE TANK, TC407/412, S/N 20581005, HIBON VTB-820 BLOWER, S/N 1550103,	
V533	2000	MESTERNISTAD	147 089-KII OMETERS INDICATED 13 855-HOURS INDICATED	ED
V555	2006	WESTERNSTAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	50
			11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES	
			S/N 19711105, TC407/412, 19.193-LITRE CAPACITY, HIBON BLOWER	
V535	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED, 20000-LB	ED
			FRONT AXLE, 20400-LB X 3 REAR AXLES, FRONT 445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM	
•			BUDS, AIR RIDE SUSP, C/W CUSTOM VAC UNIT, TC407/412 TANK 19193 L CAPACITY, HIBON	
			BLOWER, S/N 20621005	
V537	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED TRANSMISSION, AIR	ED
			RIDE SUSPENSION, C/W CUSTOM VAC, 19,193-LITRE CAPACITY, HIBON BLOWER	
V538	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DDC 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED, 20,000-LB	ED
			FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR AXLES, 11R24.5 REAR TIRES,	
			ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC S/N 20651205, 19,193-LITRE CAPACITY,	
			HIBON BLOWER	
V539	2006	WESTERN STAR	TRI-AXLE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, 62,931-	ED
			KILOMETERS INDICATED, 5,703-HOURS INDICATED, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-	
			LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB AXLES, AIR RIDE SUSPENSION, W/TIRE PRES	
		WEGTERN GT I	CTRL, C/W CUSTOM VAC SERVICE TANK, 19.193-LITRE TANK, S/N 119731105	
V542	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DDC 60-14-515-HP ENGINE, EATON FULLER 18-SPEED, 20000-LB FRONT	ED
			/ 20333-LB X 3 REAR AXLES, FRONT 445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE	
			SUSP, C/W CUSTOM VAC UNIT, 19193L TANK CAPACITY, S/N 20680905, HIBON VTB-820 BLOWER, 167.263-KILOMETERS INDICATED. 17.284-HOURS INDICATED.	
V547	2006	WESTERN STAR	VACUUM TRUCK, TRI-DRIVE, 104,227-KILOMETERS INDICATED, 13,371-HOURS INDICATED	ED
			,	
V548	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, EATON FULLER 18-SPEED, 16600-LB FRONT / 18100-LB X 3 REAR AXLES,	ED
			FRONT 445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE SUSP, C/W CUSTOM VAC	
			UNIT, TC407/412 TANK 19193 L CAPACITY S/N 20661205, HIBON BLOWER	
V549	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, 18-SPEED TRANSMISSION, FRONT	ED
			TIRES 445/65R22.5-ALUMINUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUMINUM BUDS,	
			69,000-LB AXLES, 19,000-LITRE CAPACITY	
V552	2006	WESTERN STAR	TRI-AXLE VACUUM TRUCK, W/TIRE PRES CTRL[BAD VIN]	ED
W318	2006	WESTERN STAR	TRI-AXLE WATER TRUCK, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ED
			FRONT TIRES 415/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS,	
			60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, S/N 5KKPALCK36PY20332, C/W ADVANCE	
			TANK S/N 2AESTKB005E000227, TC406, 22,000-LITRE CAPACITY, 4" BOWIE FLUID PUMP	
W320	2006	STERLING	TANDEM-AXLE WATER TRUCK, FATON FULLER 18-SPEED TRANSMISSION, 20,000-LB FRONT AXLE,	ED
			23,000-LB. REAR AXLES, ADVANCE BODY SPEC 400 TANK S/N 2AESTGA086E000153 W/ 20,000-LITRE	
	<u> </u>	1	CAPACITY, 105,290-KILOMETERS INDICATED, 14,500-HOURS INDICATED	



Anti-rouge to the control	Appelle Commission of the Comm			or management and
Eggin 4		Make	Description	\$0(0)\$10
W324	2006	WESTERN STAR	WATER TRUCK, TRI-DRIVE	ED
W327	2006	WESTERN STAR	TRI-DRIVE TANK TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W ADVANCE TANK	
			S/N 2AESTKB046E000250, TC406, 22,000-LITRE CAPACITY, BOWIE 4" PRODUCTION PUMP, TOP	
AT013	2006	STEPHENS	SAFETY RAII TRIDEM WATER TANK TRAILER, TIRES 11r24.5, STEEL BUDS, 60,000-LB AXLES, ALUM 38M3 TANK,	GP
MIUTO	2000	STEPTIENS	(LIC W132-88)	GF
AT014	2006	STEPHENS	TRIDEM ALUM WATER TANK TRAILER, 38M3 CAPACITY, TIRES 11R24.5-ALUM BUDS	GP
AIULA .	2000	STEFFICAG	TRIDERS MEDIS WATER THAKER, SONS CAPACITY, THES TERZA-S ALONS 5005	GF
CB911	2006	WESTERN STAR	TRI-AXLE COMBO TRUCK W/ 36" SLEEPER, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED	GP
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	
			BUDS, 60,000-LB AXLES, AIR RIDE SUSPENSION, (LIC G20867), C/W PROGRESS TANK S/N V-22142C,	
			APPROXIMATELY 12,000-LITRE CAPACITY, HIBON BLOWER, 198,251-KILOMETERS INDICATED, 9,119-	
			HOURS INDICATED	
CB912	2006	WESTERN STAR	TRI-DRIVE COMBO TRUCK, DETROIT DIESEL SERIES 60 ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, 20000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61200-LB REAR	
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, INTEGRAL SLEEPER, C/W	
,			PROGRESS TANK S/N 22171C. HIBON BLOWER. BECKETT STEAMER	
NTR003 200	2006	WESTERN STAR	T/A TRACTOR W/ 36" SLEEPER, DETROIT 60 SERIES 14L 515-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 385/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W WET KIT, SLIDING 5TH	
		↓	WHEEL, 338,490-KILOMETERS INDICATED, 9,765-HOURS INDICATED	
NTROO4	2006	WESTERN STAR	T/A HIGHWAY TRACTOR, SLEEPER, CAT, 385/65R22.5 FRONT, 11R24.5 REAR, ALUM BUD, AIR RIDE,	GP
			(LIC F-96643)	
P\$115	2006	INTERNATIONA	S/A PRESSURE TRUCK, DT466 285-HP ENGINE, EATON FULLER 12-SPEED, FRONT TIRES 12,000-LB	GP
		L	FRONT AXLE, FRONT TIRES 11R22.5, ALUM BUDS, 21,000-LB REAR AXLE, REAR TIRES 11R22.5, ALUM	
			BUDS, AIR RIDE SUSPENSION, C/W JASPER 4250H STEEL TANK S/N 2J9JT10275S031041, 2	
			COMPARTMENTS - 2,100-LITRE/4,200 L,TC406, CAT TRIPLEX PUMP MODEL 3560, S/N 60406528,	
			339 502-KILOMETERS INDICATED: 10 896-HOURS INDICATED.	
RN084	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, (LIC V222-16), C/W PROPANE PIG	ED
ST204	2006	FREIGHTLINER	SERVICE BODY TRUCK W/ INTEGRAL SLEEPER, 4X4, S/A, MERCEDES OM 906 LA ENGINE, EATON	GP
		•	FULLER 9-SPEED TRANSMISSION, FRONT TIRES 11R22.5 TIRES-ALUM BUDS, 10,000-LB FRONT AXLE,	
		•	REAR TIRES 11R22.5-ALUMIUNUM BUDS, 21,000 REAR AXLE, AIR RIDE SUSPENSION, (EIC F96616),	
			C/W BRUTUS SERVICE DECK, TIGER CRANE 3-TON MODEL 6000EH, MILLER WELDER, UNDER HOOD	
			AIR COMPRESSOR 188 882-KILOMETERS INDICATED. 8.167-HOLIRS INDICATED	
TR007	2006	WESTERN STAR	T/A TRACTOR W/ 36" SLEEPER, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	GP
			TRANSMISSION, FRONT TIRES 385/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIR RIDE SUSPENSION, S/N 5KJJALCK76PY17783, C/W	
			WET KIT, SLIDING 5TH WHEEL, BOWIE 4" PRODUCTION PUMP	
TR008	2006	WESTERN STAR	T/A TRACTOR, SLEEPER, DETROIT SERIES 60 ENGINE, 515-HP, EATON FULLER 18-SPEED	GP
			TRANSMISSION	
TR009	2006	WESTERN STAR	T/A HIGHWAY TRACTOR, INTEGRAL SLEEPER, CAT ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	GP
			16,000-LB FRONT AXLE, 425/65R22.5 FRONT TIRES, ALUM BUDS, 46,000-LB REAR AXLES, 11R24.5	
			REAR TIRES, ALUM BUDS, C/W WET KIT, 199,029-KILOMETERS INDICATED, 8,275-HOURS INDICATED	
TR011	2006	WESTERN STAR	T/A TRACTOR, SLEEPER, DETROIT SERIES 60 ENGINE, 18-SPEED TRANSMISSION	GP
		1		
V540	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	GP
		1	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
		I	11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIR RIDE SUSPENSION, (LIC G-10766), C/W CUSTOM	
			VAC SERVICES S/N 20671205, TC407/412, 19,193-LITRE CAPACITY, HIBON BLOWER	



Fojeljo, #	Your	Make	Description	in en
V545	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED, 20000-LB FRONT AXLE,	ED
			20000-LB X 3 REAR AXLES, FRONT 445/65R22.5, ALUM BUDS, 11R24.5 REAR ALUM BUDS, AIR RIDE	
			SUSP, C/W CUSTOM VAC UNIT, TC407/412 TANK 19193 L CAPACITY S/N 19721105, HIBON BLOWER,	
			HOIST, AGITATOR, VIBRATOR (NOT IN SERVICE ON DATE OF INSPECTION)	
W322	2006	INTERNATIONA	6X4 WATER TRUCK, ENGINE, AUTOMATIC TRANSMISSION, FRONT TIRES 11R22.5-STEEL BUDS, REAR	ZC
		L	TIRES 11R22.5-STEEL BUDS, AIR RIDE SUSPENSION, C/W TREMCAR TANK, 16,000-LITRE, S/N 4962,	
			116,180-Kilometers indicated, 11,182-Hours indicated	
W323	2006	INTERNATIONA	T/A WATER TRUCK, DT 570 9.3 £ ENGINE, AUTO, FRONT TIRES 11R22.5, STEEL BUDS, REAR TIRES	ED
		Ļ	11R22.5, STEEL, C/W TREMCAR ALUM TANK, S/N 4961, 16,000-LITRE, BOWIE PUMP, 187,950-	
			KILOMETERS INDICATED	
V530	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ED
		1	FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS,	
			61,200-LB REAR AXLES, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES S/N 20560905,	
			TC407/412, 19.193-LITRE CAPACITY, HIBON BLOWER	
W325	2006	WESTERN STAR	TRI-AXLE TANK TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 441/65R22.5-ALUM	CNRL
			BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION,	HORIZO
			(LIC G10752) S/N 5KKPALCK86PY66724, C/W ADVANCE TANK, TC 406 SPEC, 22,000-LITRE CAPACITY -	PROJEC
			S/N 2AESTKB045E000229	
H709	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,000-LB REAR	
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES	
			HYDRO-VAC UNIT S/N 19640605, TC407/412 TANK, 10,822-LITRE CAPACITY, BOC EDWARDS	•
			BLOWER, HOTSY HEATER, HOTSY STEAMER, WATER BLAST SYSTEM	
H711	2006	WESTERN STAR	TRI-DRIVE HYDRO VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	
			BUDS, 61,200-LB AXLES, AIR RIDE SUSPENSION, (LIC F77613), C/W CUSTOM VAC S/N 19660805,	
			TCRN Z.05.438.05.04, TC 402/412, 10,822-LITRE CAPACITY, ROBUSCHI BLOWER S/N 05-06887	
1174.0	2005	MESTERN STAR	TO AVIE WORD WAS IN A TRUCK OUT OF STEAD FROM EATER OF SPEED FROM THE	# P.
H712	2006	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED, FRONT AXLE	ED
		İ	445/65R22.5 ALUM BUDS 20,000-LB, REAR AXLES 11R24.5 ALUM BUDS 61,200-LB, AIR RIDE	
			SUSPENSION, C/W CUSTOM VAC SERVICES HYDRO-VAC, 10,822 L CAPACITY, TC407/412 SPEC S/N	
			19670805, C/W HYDRAULIC HOIST, REAR DOOR, VIBRATOR, DUMP GATE & KNIFE VALVE, HOTSY,	
H724	2006	MECTEDNI CTAD	HYDRALILIC ROOM TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED,	CNRL
11724	2000	WESTERN STAR		CIVIL
			20,000-LB FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES, REAR TIRES	
			11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES HYDRO-VAC UNIT S/N	
			24590306, 10,822 L TANK, TC 407/412, EDWARDS BLOWER, HOTSY STEAMER & WATER BLAST	
			SYSTEM, HYDRAULIC BOOM, 140,019-KILOMETERS INDICATED, 11,187-HOURS INDICATED	
P168	2006	CHEVROLET	PICKUP TRUCK, 3/4-TON, CREW CAB, DURAMAX DIESEL	ED
RN077	2006	ROUGHNECK	T/A RV TRAILER, PROPANE PIG, KITCHEN, BATHROOM, BEDROOM, TIRES 235/80R16	ED
ST206	2006	FREIGHTLINFR	 SERVICE TRUCK W/ INTEGRAL SLEEPER, S/A, MERCEDES BENZ ENGINE, EATON FULLER 10-SPEED	ED
			TRANSMISSION, FRONT TIRES 11R22.5-ALUM BUDS, 10,000-LB AXLE, REAR TIRES 11R22.5-ALUM	
			BUDS, 21,000-LB AXLE, (LIC G10836), C/W TIGER CRANE MODEL 6031C, 3-TON BLOCK, BRUTUS	
		1	ISERVICE DECK	
	2000	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, CAT C15 ENGINE, EATON FULLER 18-SPEED, 141,368-KILOMETERS	ĘD
V544	2006		INDICATED, 14,690-HOURS INDICATED, 16600-LB FRONT AXLE, 17750-LB X 3 REAR AXLES, FRONT	
V544	2006			
V544	2006		445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE SUSP, C/W CUSTOM VACUALITY	
V544	2006		445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE SUSP, C/W CUSTOM VAC UNIT,	
V544 RN068	2006		445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE SUSP, C/W CUSTOM VAC UNIT, TC407/412 TANK 19193 L CAPACITY. S/N 23030106. HIBON VTB-820 BLOWER T/A RV TRAILER, (LIC V068-26), C/W PROPANE PIG	ED
		ROUGHNECK	TC407/412 TANK 19193 L CAPACITY, S/N 23030106, HIBON VTB-820 BLOWER	ED ED

Equip: #	Year	t/lake	Description	Location
RN075	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER	ED
RN078	2006	ROUGHNECK	TRAILER, CREW SHACK	ED
RN088	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, (LIC V381-00), C/W PROPANE PIG	ED
RN103	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, 235/85R16 TIRES, PROPANE PIG, KITCHEN BATHROOM, BEDROOM	ED
RN104	2006	ROUGHNECK	XTREME DUTY TANDEM-AXLE TRAILER, TIRES 235/80R16-STEEL BUDS, PROPANE PIG, (LIC P842-99)	ED
NTR002	2006		TANDEM-AXLE HIGHWAY TRACTOR W/SLEEPER, DETROIT DIESEL SERIES 60 14L 515-HP ENGINE, EATON FULLER 18-SPEED, FRONT TIRES 385/65R22.5, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC F-77618), 253,514-KILOMETERS INDICATED, 5,941-HOURS INDICATED	ED
RN087	2006	ROUGHNECK	T/A RV TRAILER, KITCHEN, BED, 100-LB PROPANE RIG	ED
RN090	2006	ROUGHNECK	T/A RV TRAILER, (LIC V381-02), C/W PROPANE PIG	ED
RN091	2006	ROUGHNECK	XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG, S/N 2TTNT30046R600431	ED
V528	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, 165,136-KILOMETERS INDICATED	ED
V534	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 445/65R22.5 FRONT TIRES, 11R24.5 REAR TIRES, AIR RIDE SUSPENSION, TIRE PRESSURE MONITOR, (LIC. F96632), C/W CUSTOM VAC UNIT, TC 407/412, 19,193-LITRE CAPACITY, S/N 20511005, HIBON BLOWER	ED
V543	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION	ED
W317	2006	WESTERN STAR	TRI-DRIVE TANK TRUCK, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-STEEL BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION, (LIC F96634), C/W ADVANCE TANK, TC SPEC 406, 22,000-LITRE CAPACITY - S/N 2AESTKB055E00024 [SUCKED-IN DENTED BACK PANEL DUE TO DISCHARGING WITHOUT VENTING; STILL USABLE]	ED
H721	2006	WESTERN STAR	HYDRO VACUUM TRUCK, CATERPILLAR C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, CUSTOM VAC SERVICES 10,822-LITRE TANK	CNRL
H726	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB AXLES, AIR RIDE SUSPENSION, 36" SLEEPER, C/W CUSTOM VAC SERVICES, 10,822-LITRE	CNRL .
H731	2006	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED, 20,000-LB FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G20885), C/W CUSTOM VAC SERVICES HYDRO-VAC UNIT S/N 25370606, 10,822 L TANK, TC 407/412, ROBUSHI BLOWER, HOTSY STEAMER & WATER BLAST SYSTEM HYDRALILIC BOOM NEW DROP BOX	ED
H734	2006	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED, FRONT TIRES 445/65R22.5, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G-19638), C/W CUSTOM VAC UNIT, S/N 25320406, 10.822 L. TC407/412	ED
NP009	2006	WESTERN STAR	TRI-AXLE NITROGEN PUMP UNIT TRUCK, CAT C15 ACERT 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 69,000-LB REAR AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIRLINER AIR RIDE SUSPENSION, SLEEPER/CONTROL CAB COMBO, (LIC G20901), C/W CRYOGENIC VESSEL ALTERNATIVES NITROGEN VESSEL S/N 30135, MC-338, 26,370-LB WATER CAPACITY, BLACKMORE MODEL TX03E PUMP, BUTECH PRESSURE SYSTEM, XCHANGER INC. MODEL C-250 COIL HEATER, 10 1"X10" JOINTS, 5 5" JOINTS, 195,421-KILOMETERS	GP
NST003	2006	АМКО	TRIDEM NITROGEN TANK TRAILER, 69,000-LB AXLES, TIRES 11R22S, ALUM BUDS, AIR RIDE SUSPENSION, (LIC N65174), C/W TC341 TANK S/N V4472, 36,877 L TANK	EĐ

Едиір, #	Year	Make	Description	Location
NST004	2006	AMKO	TRI-AXLE NITROGEN TANK TRAILER, 69,000-LB AXLES, TIRES 11R22.5, AIR RIDE SUSPENSION, C/W ALLOY TANK, TC341, 82361-LB CAPACITY, TCRN Z053590305, TANK S/N V4473, 9,742-GALLON CAPACITY	ED
RN079	2006	ROUGHNECK	TRAILER, CREW SHACK	ED
RN102	2006	ROUGHNECK	XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG	ED
RN105	2006	ROUGHNECK	XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG	ED
V551	2006	WESTERN STAR	TRIDRIVE VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES STEEL TANKS/N 23060106, TC SPEC 407/412, 19,193-LITRE CAPACITY, HIBON VTB-820 BLOWER S/N 37933510. TIRE PRESSURE CONTROL SYSTEM	CNRL
V557	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK W/ SLEEPER, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18- SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CARGO TANK, TC407/412. 15.959-LITRE CAPACITY. HIBON BLOWER	CNRŁ
AT007	2006	HEIL	TRIDEM ALUM 407 WATER TANKER, 37,854-LITRE CAPACITY, TIRES 11R24.5-STEEL BUDS, 60,000-LB AXLES. AIR RIDE SUSPENSION. (LIC V158-52)	GP
AT011	2006	N/A	TRIDEM WATER TANKER, 10,000-GALLON TANK	ZC
V532	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DDC 60 SERIES ENGINE, EATON FULLER 18-SPEED, 20,000-LB FRONT AXLE, 445/65R22.4 FRONT TIRES, ALUM BUDS, 61,200 REAR AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC, TC407/412 TANK 19193L CAPACITY S/N 20591005, HIBON BLOWFR	GP
H736	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT SERIES 60 ENGINE, EATON FULLER 18-SPEED TRANSMISSION, ROBUSHI BLOWER, HOTSY HOT WATER SYSTEM, ONAN HYDRAULIC BOOM, CUSTOM VAC TANK S/N 24700306 W/ 10.822-LITRE CAPACITY	ED
RN115	2006	ROUGHNECK	XTREME DUTY 20' TANDEM-AXLE TRAILER, TIRES 235/80R16, PROPANE PIG	ED
RN116	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, PROPANE PIG, KITCHEN, BATHROOM, BEDROOM, 235/80RTIRES	ED
H754	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, 18-SPEED TRANSMISSION, C/W CUSTOM VAC SERVICES STEEL TANK [DAMAGED IN ROLLOVER; NOW OPERATIONAL; UNIT APPRAISED AS IN REPAIRED CONDITION]	ED -
W328	2006	PETERBILT	T/A WATER TRUCK, CAT C7 ENGINE, EATON FULLER 8 TRANSMISSION, FRONT TIRES 11R22.5-ALUM BUDS, 12,000-LB AXLE, REAR TIRES 11R22.5-ALUM BUDS, 40,000-LB AXLES, AIR RIDE SUSPENSION, (LIC G57786), C/W ADVANCE TANK S/N 2AESTGBOX5S000136, 14,320-LITRE CAPACITY	ED
AT010	2006	HEIL	TRIDEM ALUM WATER TANKER, THES 11R24.5-STEEL BUDS, HENDRICKSON AIR RIDE SUSPENSION, C/W ALUM TANK, DOT 407, 37,850-LITRE CAPACITY, EXTERIOR INSULATION	ZAMA CITY
NTRO01	2006	WESTERN STAR	T/A HIGHWAY TRACTOR, SLEEPER, DETROIT DIESEL 14L 515-HP, EATON FULLER, 385/65R22.5 FRONT, 11R24.5 REAR, ALUM BUDS, 23000-LB X 2 AXLES, (LIC F-77617)	ZAMA CITY
SU012	2006	WESTERN STAR	TRI-DRIVE SOUR SEAL TRUCK W/ 32" SLEEPER, DDC 60 515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES 48X31.00-20 NHS ND, 60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES LTD STEEL TANK S/N 25600706, TC406, 20,123-LITRE CAPACITY, 3" T&E FLUID PUMP	ZAMA CITY
V550	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DDC 60-14 515-HP ENGINE, EATON FULLER 18-SPEED 20000-LB FRONT / 20333-LB X 3 REAR AXLES, FRONT 445/65R22.5, REAR 11R24.5, AIR RIDE SUSP, C/W CUSTOM VACUNIT, 19193 L CAPACITY TANK S/N 23040105, HIBON VTB-820 BLOWER	ZAMA CITY
W321	2006	INTERNATIONA L	T/A WATER TRUCK, ALLISON AUTO TRANS, FRONT 11R22.5 STEEL BUDS, REAR 11R22.5 STEEL BUDS, C/W TREMCAR ALUM TANK, 16000-LITRE CAPACITY S/N4963	GP
P161	2006	FORD	PICKUP TRUCK, CREW CAB, 3/4-TON, 75-GALLON WESTEEL DIESEL TANK W/ 12V DC PUMP, 287,237- KILOMETERS INDICATED	MEDICINE HAT

Found #	Veni	Make	Description	- koreal) je n
P178	2006		PICKUP TRUCK, 4X4 CREW CAB S/B, DURAMAX ENGINE, AUTOMATIC TRANSMISSION, TIRES	MEDICINE
1270	2000		LT265/75R16, MOTOROLA PHONE, CROSS OVER TOOL BOX, HITCH, 253,404-KILOMETERS INDICATED	HAT
P188	2006	DODGE	PICKUP TRUCK, MEGA CAB, 4WD, TURBO DIESEL ENGINE [NOT IN SERVICE ON DATE OF INSPECTION]	MEDICINE HAT
P191	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, TURBO DIESEL ENGINE	MEDICINE HAT
P196	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, ALLISON DURAMAX, 471-LITRE TANK W/ ELECTRIC PUMP	MEDICINE HAT
P197	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, DURAMAX DIESEL, 4X4, (LIC G10772)	MEDICINE HAT
P257	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, DURAMAX DIESEL	SC
RN076	. 2006	ROUGHNECK	T/A RV TRAILER, C/W PROPANE PIG	MEDICINE HAT
RN086	2006	ROUGHNECK	T/A RV TRAILER (LIC V222-18), C/W PROPANE PIG	MEDICINE HAT
S004	2006	PETERBILT	STEAM TRUCK, ESTIMATED 8,000-LITRE CAPACITY TANK, 24' BOX VAN, THERMO KING MODEL 4- 15822, S/N VB3601 BOX	SC
UT105	2006	CUSTOM MANUFACTUR ED	TRAILER, WELDING, 10' FLATDECK, W/ GENERATOR	MEDICINE HAT
AT009	2006	HEIL	TRIDEM WATER TANKER, 10,000-GALLON TANK	SWIFT CURRENT
H729	2006	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, FULLER 18-SPEED, 20,000-LB REARS, CUSTOM VAC UNIT S/N 25350506. 10.822L CAP. DUMP BED. WATER BLAST BOILER	SWIFT CURRENT
H732	2006		TRI-DRIVE HYDRO VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,000-LB AXLES, AIR RIDE SUSPENSION, (LIC G20877), C/W CUSTOM VAC S/N 25290606, TCRN Z.05.437.05.04, TC 407/412, 10,822-LITRE CAPACITY, ROBUSCHI BLOWER S/N 05-27026	SWIFT
H740	2006	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 20,000-LB FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC UNIT, TC407/412, 10,822 L, ROUSCHI BLOWER, HOTSY, WATER BLAST WB9460 HEATER, HYDRAULIC BOOM, S/N 27270806	SWIFT CURRENT
H742	2006	WESTERN STAR	TRI-DRIVE HYDRO VAC W/ SLEEPER, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,200-LB AXLES, AIR RIDE SUSPENSION, (LIC G20946), C/W CUSTOM VAC S/N 27910906, TCRN Z.05.437.05.04. TC 407/412. 10822-LITRE CAPACITY	SWIFT CURRENT
P256	2006	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, SHORT BED, DURAMAX DIESEL ENGINE, ALLISON AUTOMATIC, TIRES 265/75R16, FULLY LOADED, C/W SLIP TANK AND PUMP [NOT IN SERVICE AT TIME OF INSPECTION]	SWIFT CURRENT
P262	2006	GMC	PICKUP TRUCK, CREW CAB 4X4 S/B, DURAMAX DIESEL ENGINE, ALLISON TRANSMISSION, TIRES 265/75R16, C/W ALUM CABINET, 175,018-KILOMETERS INDICATED	SWIFT CURRENT
RN114	2006		T/A RV TRAILER, CREW SHACK	MH
ST208 -	2006	INTERNATIONA L	4X4 SINGLE-AXLE SERVICE TRUCK, D255 ENGINE, EATON FULLER 18-SPEED, FRONT TIRES 11R22.5, STEEL BUDS, REAR TIRES 11R22.5, STEEL BUDS, (LIC G19632), C/W CRANE & RIG BODY S/N CRT060801, CRANE RKi3200 ERX15 S/N 32003, COMPRESSOR S/N 174730606, GETEC 250-5.5 WELDER INOT IN SERVICE AT TIME OF INSPECTION]	SWIFT CURRENT
UT357	2006	sws	T/A FLATDECK GOOSENECK TRAILER, TIRES ST235/80R16-STEEL BUDS, 14,000-LB AXLES, (LIC V510- 82)	SWIFT CURRENT
V554	2006	KENWORTH	PRODUCTION VACUUM TRUCK, TANDEM, CW CUSTOM VAC TANK, S/N ZO543704TC407/412, 13,612- LITRE, 160,827-KILOMETERS, 8134-HOURS INDICATED	SWIFT CURRENT

V555 V556 VR127	2006		T/A VACUUM TRUCK, CUMMINS ISX-500 ENGINE, EATON FULLER 18-SPEED, FRONT TIRES 445/65R22.5 ALUM BUDS, REAR TIRES 11R24.5 ALUM BUDS, NEWAY AIR RIDE SUSPENSION, C/W	SWIFT CURRENT
V556	2006		· · · · · · · · · · · · · · · · · · ·	
	2006	ł I	,	
	2006		CUSTOM VAC UNIT, HIBON BLOWER, TC407/412, 13612 L CAP, S/N 23381205, 295,975-KILOMETERS,	
	2006		15.474-HOURS REPORTED	
	2000	WESTERN STAR	VACUUM TRUCK, TANDEM, CUSTOM VAC 13,612L VAC UNIT, 227,484-KILOMETERS, 15,488-HOURS	SWIFT
VR127		11723.21.11017.11	REPORTED	CURRENT
VIV.127	2006	KENWORTH	TRI-DRIVE VACUUM TRUCK W/ 36" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	SWIFT
	2000		TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	CURRENT
			11R24.5-ALUM BUDS, 52,000-LB REAR AXLES, NEWAY AIR RIDE SUSPENSION, C/W WESTECH VAC	COMMENT
1			· · · · · · · · · · · · · · · · · · ·	
			SYSTEM S/N WT0705045, TC407, 14,600-LITRE CAPACITY, HIBON BLOWER, 88,713-KILOMETERS	
VTR001	2006	M/ECTEDMICTAD	INDICATED 667-HOURS INDICATED SEMIVAC HIGHWAY TRACTOR, TRI-AXLE	GP
ALVOOT	2000	WESTERN STAR	SEIVILVAC FIGHTWAT TRACTOR, TRIPARLE	GF.
P250	2005	DODGE	PICKUP TRUCK, MEGA CAB, TIRES 265/70R17, C/W BUCKET SEATS, CUMMINS TURBO DIESEL	SC
			ENGINE, 4WD	
CT2061	2006	KENWORTH	TANDEM-AXLE STEER TRI-AXLE COILED TUBING UNIT, CAT ENGINE, EATON FULLER TRANSMISSION,	ED
		,	FRONT TIRES 425/65R22.5, ALUM BUDS, REAR TIRES 1.1R24.5, ALUM BUDS, AIR RIDE SUSPENSION,	
			C/W FASSI F150AC.24 7-TON PICKER S/N 4994, SIDE OUTRIGGERS, 60" GOOSENECK, HYDRA RIG RT25	
ŀ			INJECTOR, OPERATORS CABIN	
AT012	2006	STEPHENS	TRIDEM WATER TANK TRAILER, 60,000-LB AXLES, TIRES 11R24.5, ALUM BUDS, ALUM 38M3	MEDICINE
AIUIZ	2000	4	CAPACITY, (LIC V510-83)	HAT
V553	2006		TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	ED
4335	2000		TRANSMISSION, FRONT TIRES 445/65R22.5-ALUMINUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-	LL
H717	2006	WESTERN STAR	ALUMINUM BUDS TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	ZC
П/1/	2000		445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC UNIT, S/N	ZC
			, , , , , , , , , , , , , , , , , , , ,	
RN098	2006	BOHCHNECK	20491205, 10,822-LITRE T/A RV TRAILER, (LIC V381-30), C/W PROPANE PIG	мн
VIANDA	2006	ROUGHNECK	T/A NV TRAILER, (LIC VS81-SU), C/ W PROFAINE FIG	IVIT
RN107	2006	ROUGHNECK	T/A RV TRAILER (LIC V455-19), C/W PROPANE PIG	МН
TWP013	2006	CARGO	TRIDEM STIFF POLE TANK PUP TRAILER, TIRES 11R24.5-ALUM BUDS, AIR RIDE SUSPENSION, (LIC	ED
			G50681), C/W ALUM TANK S/N V-219106, TC407/412, 15,985-LITRE CAPACITY	
TRP802	2006	GMC	PICKER TRUCK, QUAD CAB, FLAT DECK, FRONT TIRES 245/70R19.5, ALUM BUDS, REAR TIRES	MH
			245/70R19.5, ALUM BUDS, C/W PALFINGER PK7501 KNUCKLE PICKER S/N 100041114, FLAT DECK,	
			CABINETS, 5TH WHEEL, SIDE OUTRIGGERS, 196,659-KILOMETERS AS OF JULY 2011	
UT202	2006	MUSTANG	TRAILER, GOOSENECK, STEP DECK, 10' TOP DECK, 30' WORKING DECK	MH
RN080	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER	
				·
V527	2006		TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	
			FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 61,000-	
			LB AXLE, AIR RIDE SUSPENSION, (LIC F77620), C/W CUSTOM VAC, TC SPEC 407/412, 19,193-LITRE	
			CAPACITY - S/N 20570905, HIBON BLOWER	
₽251	2006	CHEVROLET	PICKUP TRUCK, CREW CAB 4X4 SHORT BED, DURAMAX DIESEL ENGINE, LT 245/75R16 TIRES, ALUM	
		1	BUDS, C/W TIDY TANK, ALUM TOOL BOX, LEATHER, FIBERGLASS CANOPY	
V536	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, W/TIRE PRESSURE CONTROL, DETROIT 60 SERIES ENGINE, EATON	
		1	FULLER 18-SPEED, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR	
] !	AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, C/W CUSTOM VAC UNIT,	
			TC407/412, 19193 L CAP, HIBON VTB820, S/N 20631005	
W316	2006	WESTERN STAR	TRI-DRIVE TANK TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
]	11R24,5-ALUM BUDS, 61,000-LB REAR AXLES, AIR RIDE SUSPENSION, C/W ADVANCE TANK, TC403,	
		1	22,000-LITRE CAPACITY, BOWIE 4" PRODUCTION PUMP, TOP SAFETY RAIL, 138,175-KILOMETERS	
			INDICATED, 20.128-HOURS INDICATED	



-FigNips.#	1	Wake	Description	Location
W319	2006	WESTERN STAR	TRI-AXLE WATER TRUCK, DETROIT 60 SERIES ENGINE, EATON FULLER TRANSMISSION, FRONT	
			445/65R22.5 ALUM BUDS, REAR 11R24.5 ALUM BUDS, AIR RIDE SUSPENSION, C/W ADVANCE TANK	
			TC 406, 22,000-LITRE CAPACITY S/N 2AESTKB075E000225	
V541-A	2006	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, PURCHASED USED IN 2014, DETAILS NOT AVAILABLE, ASSUMED	
		····	SIMILAR TO LIKE ASSETS AND REBUILT BASED ON COST INFO PROVIDED	
AT015	2006	STEPHENS	TRIDEM ALUM WATER TRAILER, TIRES 11R24.5-STEEL BUDS, 60,000-LB AXLES, (LIC W132-94)	
HUT002	2006	NORBERTS	TRIDEM ENCLOSED GLYCOL HEATING UNIT, TIRES LT235/85R16-STEEL BUDS, 21,000-LB AXLES, (LIC	МН
			V510-90) S/N 2N9E7S7416G017303, C/W HEATER WATER BLAST MODEL WB9460-SN 61781206,	
			WATER PRESSURE WAND, 2-832-LITRE CAPACITY TANKS, 500-LITRE CAPACITY TANK, 4,542-LITRE	
			CAPACITY TANK, MITSUBISHI DIESEL GENERATOR	
RN072	2006	ROUGHNECK	TANDEM-AXLE RV TRAILER, PROPANE PIG	ED
V546	2006	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	
¥3+0	2000	VACOICAIVAIAN	TRANSMISSION	
ST207	2006	FREIGHTLINER	SERVICE TRUCK, S/A, 4X4, EXTENDED CAB, MERCEDES BENZ ENGINE, 8-SPEED TRANSMISSION,	ZAMA CITY
			FRONT TIRES 11R24.5-ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS,	
			21,000-LB REAR AXLE, C/W BRUTUS 8'X14' SERVICE BODY W/ TIGER 6541E TRUCK CRANE, 40,000-LB	
			IS/N 102441, FRONT BUMPER WINCH	
TR010	2006	WESTERN STAR	T/A TRACTOR, SLEEPER, DETROIT SERIES 60 DIESEL ENGINE, 515-HP, EATON FULLER 18-SPEED	ZAMA CITY
			TRANSMISSION	
TWP014	2006	CUSTOM VAC	TRIDEM STIFF POLE TANK TRAILER PUP, TIRES 11R24.5-ALUM BUDS, C/W CUSTOM SERVICES VAC	
		SERVICES	STEEL TANK S/N 28401106, TC407, 20,123-LITRE CAPACITY, TITAN LOGIX LEVEL METER	
CD2061	2007	FREIGHTLINER	SINGLE-AXLE BOOM TRUCK 4X4, MERCEDES BENZ ENGINE, ALLISON AUTOMATIC, FRONT TIRES	ED
			315/80R22.5, ALUM BUDS, C/W FASSI F110A-24 KNUCKLE BOOM CRANE, S/N 6779, SIDE	
			OUTRIGGERS, DECK	
CD2062	2007	FREIGHTLINER	BOOM TRUCK, SINGLE AXLE, 4X4, MERCEDES BENZ ENGINE, AUTOMATIC TRANSMISSION, 1LB FRONT	ED
			AXLE 11R24.5 FRONT TIRES, ALUM BUDS, 21000-LB REAR AXLE, 11R24.5 REAR TIRES, ALUM BUDS,	
			AIR RIDE SUSPENSION, INTEGRATED SLEEPER, C/W FASSI F110A-24 KNUCKLE BOOM CRANE, S/N	
			7076, SIDE OUTRIGGERS, DIESEL, METHANOL TANK, TOOL CABINET, 12' DECK, 150,485-KILOMETERS	
			INDICATED 5 490-HOURS INDICATED	
CD2072	2007	FREIGHTLINER	BOOM TRUCK, SINGLE-AXLE, 4X4, MERCEDES BENZ ENGINE, FRONT TIRES 11R22.5-STEEL BUDS,	ĘD
			12,000-LB AXLE, REAR TIRES 11R22.5-STEEL BUDS, 21,000-LB AXLE, (LIC G-57779) S/N	
			1FYDCXCS07HY24000, C/W FASS! F110A.24 KNUCKLE BOOM CRANE- S/N 6788, 12' DECK 110,904-	
			KILOMETERS INDICATED, 3.618-HOURS INDICATED	
CD2073	2007	FREIGHTLINER	BOOM TRUCK, SINGLE-AXLE, 4X4, W/INTEGRAL SLEEPER, MERCEDES BENZ ENGINE, ALLISON	ED
			TRANSMISSION, 12,000-LB FRONT AXLE, FRONT TIRES 11R24.5, STEEL BUDS, 21,000-LB REAR AXLE,	
			REAR TIRES 11R24.5, STEEL BUDS, (LIC G-64535) S/N 1FYDCXCSX7HX29699, C/W FASSI F100 A.24	
			KNUCKLE BOOM CRANE- S/N 7044, 12' DECK 89,600-KILOMETERS INDICATED, 3,837-HOURS	
			INDICATED	
CT2071	2007	KENWORTH	TANDEM-AXLE STEER/TRI-AXLE DRIVE COILED TUBING UNIT, CAT C15 550-HP ENGINE, EATON	ED
٠			FULLER 18-SPEED, 32,000-LB FRONT AXLES, FRONT TIRES 425/65R22.5, ALUM BUDS, 69,000-LB REAR	
			AXLES, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, SPICER 792, C/W COILED TUBING	
			UNIT, 3100M OF 1 1/4" TUBING, HYDRA RIG RT25 INJECTOR - 3/4" TO 2", CAT 4517 TRIPLEX PUMP,	
			50L/MIN @ 21 MPA, W/NORTHWEST MODEL 5450 COMPRESSOR, 18M3 @ 14 MPA, W/ AMCO	
			VEBA V815F-45 PICKER, 6.7M REACH, S/N 20159, 162,147-KILOMETERS INDICATED, 9,538-HOURS	
			INDICATED	
NP012	2007	WESTERN STAR	TRI-AXLE NITROGEN PUMP UNIT W/ SLEEPER/CONTROL CAB COMBO, CAT C15 ENGINE, EATON	ED
			FULLER 18-SPEED, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES, REAR TIRES	
		1	11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G-57750), C/W WELLQUIP NITROGEN PUMP UNIT	
	1	1	S/N 0506-89906, XCHANGER C-250 ACD RECIPROCATING TRIPLEX PUMP, CVA TANK 11,234 LITRE,	
		1	S/N 30202, TCRN 25670, TC341 TANK, C/W TOOLING & DOWNHOLE TOOLS, 218456-KILOMTERES	
		,	INDICATED 8064-HOURS INDICATED	
P270	2007	GMC	PICKUP TRUCK, 3/4-TON, CREW CAB, 4WD, FUEL PUMP, TURBO DIESEL ENGINE, 214,989-	ED
		1	KILOMETERS INDICATED	
P276	2007	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, SHORT BED, DURAMAX ENGINE, AUTOMATIC TRANSMISSION,	ED
	t	1	FULLY LOADED, AC, (LIC HWA172), 206,496-KILOMETERS REPORTED AS OF JUNE 2011	



Formp. #	Year	Make	Description	koestjon
CB908	2007		TRI-DRIVE COMBO TRUCK, 32" SLEEPER, DETROIT SERIES 60 515-HP ENGINE, 18-SPEED, FRONT TIRES 445/65R22.5-ALUM BUDS, 23000-LB X 3 AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CARGO TANK MFG 2-COMPARTMENT TANK, WATER FRONT 1,216-GALLON, 2,000-PSI WATER PUMP, REAR 2,978-GALLON VACUUM, 11,299-LITRE VACUUM	GP
			CAPACITY, TC SPEC 407/412, S/N V22326C, HIBON VTB-820 BLOWER, HOTSY WB 9460 HEATER	
CB909	2007		TRI-DRIVE COMBO VACUUM TRUCK W/ 36" SLEEPER, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 69,000-LB AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CARCO 2 COMPARTMENT TANK S/N V22222C, DOT407/412, 4,675-LITRE CAPACITY, 11,299-LITRE CAPACITY, HIBON VTB 820 BLOWER, WATER BLAST HEATER, 2,000-LITRE WATER PUMP CAPACITY [REPORTED]	GP
H738	2007		AS NO LONGER IN SERVICE! TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, 18-SPEED, 59,314- KILOMETERS INDICATED, 5,484-HOURS INDICATED, FRONT AXLE 445/65R22.5 ALUM BUDS 20,000- LB, REAR AXLES 11R24.5 ALUM BUDS 69,000-LB, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES LTD. 10,822-LITRE VACUUM CAPACITY, TC SPEC 407/412, S/N 27260806, ROBUSCHI RB DV 145 SP UN BLOWER, S/N 06-28427, HOTSY 9460 WATER BLAST HEATER, S/N 60160706; 2,000-PSI WATER PUMP, ESTIMATED 5,000-LITRE CAPACITY WATER TANK	GP
H746	2007		TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT TIRES 445/65R22.5 ALUM BUDS, REAR TIRES 11R24.5 ALUM BUDS-60, AIR RIDE SUSPENSION, C/W CUSTOM VAC UNIT, TC407/412, 10822 L CAP, ROBUSCHI BLOWER, HOTSY, WATER BLAST WB9460 HEATER. HYDRAULIC BOOM. S/N 28841006	GP
HU003	2007		FLUID HEATER TRUCK W/ 48" SLEEPER, TRI AXLE, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 425/65R22.5-ALUM BUDS, 16,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 52,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W HAMMS TANK S/N 46ANM063M61321828, TC406-MS-03-EL, 3 COMPARTMENT - 1,350-LITRE/1,350-LITRE/ 7,655-LITRE CAPACITY, CHANDLER MFG DIESEL FIRE HEATER S/N 4810, 12 MILLION BTU, FAN EQUIPMENT S/N 06-256B, SIZE 1-35, 3430 RPM (16,440-KILOMETERS INDICATED, 8750-HOURS INDICATED)	GP
NTR006	2007	WESTERN STAR	T/A TRACTOR W/ 44" SLEEPER, DETROIT 60 SERIES 14L 515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, FRONT TIRES 385/65R22.5-ALUM BUDS, 14,600-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS, 46,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, EXTENDED CHASSIS, C/W WET KIT. SLIDING.5TH WHEEL, PTO, HYDRAULIC PUMP	GP
PS100	2007	FREIGHTLINER	PRESSURE TRUCK, S/A, MERCEDES BENZ ENGINE, ALLISON AUTOMATIC TRANSMISSION, 12,000-LB FRONT AXLE, 11R22.5 FRONT TIRES, ALUM BUDS, 21,000-LB REAR AXLE, 11R22.5 REAR TIRES, ALUM BUDS, BUSINESS CLASS M2, AIR RIDE SUSPENSION, (LIC G50660), C/W CUSTOM VAC SERVICES PRESSURE TANK UNIT S/N 29811206, 10,329-LITRE CAPACITY, TC407 TANK, 3 COMPARTMENT, TRIPLEX PRESSURE PUMP, HOSES, REEL, WATER PRODUCT PUMP [NO BATTERY]	ED
PS102	2007	FREIGHTLINER	PRESSURE TRUCK, S/A EXTENDED CAB , MERCEDES BENZ ENGINE, AUTOMATIC TRANSMISSION, FRONT TIRES 11R22.5-ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 11R22.5-ALUM BUDS, 21,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC SERVICES STEEL TANK 5/N 29820107, TC407, 3 - COMPARTMENT 10,329-LITRE CAPACITY, 3" BOWIE FLUID PUMP, APLEX TRIPLEX PLUNGER PUMP MODEL SO 15H S/N 4879	ED
PS103	2007	FREIGHTLINER	PRESSURE TRUCK, 4X4	GP
PT110	2007	INTERNATIONA L	S/A PRESSURE TEST TRUCK W/ EXTENDED CAB, DT 466 ENGINE, EATON FULLER 10-SPEED TRANSMISSION, FRONT TIRES 11R22.5-ALUM BUD, 12,000-LB FRONT AXLE, REAR TIRES 11R22.5- ALUM BUDS, 20,000-LB REAR AXLE, AIR RIDE SUSPENSION, C/W INTERCONTINENTAL 20' VAN BODY S/N 7261 BC, GARDNER DENVER T300M PUMP, TC SPEC 406, S/N 7238, 4-CUSTOM VAC TANKS, 426- LITRE, S/N'S 2978A, 2978B, 2978C, & 2978D	GP
PT111 _.	2007	INTERNATIONA L	S/A PRESSURE TEST TRUCK, EXT CAB, DT 466 ENGINE, 6-SPEED TRANSMISSION, FRONT TIRES 11R22.5-ALUM BUDS-, 12,000-LB AXLE, REAR TIRES 11R22.5-ALUM BUDS, 21,000-LB AXLE, AIR RIDE SUSPENSION, C/W ITB 20' VAN BODY, S/N 7262 BC, GARDNER DENVER T300M PUMP, S/N 7241, 4- TANKS	ED



Equip: #	Үсаг	Make	Description	Lineathon
PT113	2007		S/A PRESSURE TEST TRUCK, EXT CAB, DT 446 ENGINE, EATON FULLER 6-SPEED TRANSMISSION,	GP
11113	2007	L	12,000-LB FRONT AXLE, 11R22.5 FRONT TIRES, ALUM BUDS, 21,000-LB REAR AXLE, 11R22.5 REAR	- Oi
		"	TIRES, ALUM BUDS, INTEGRAL SLEEPER, C/W INTERCONTINENTAL TRUCK BODY S/N 7264BO,	
			OPERATOR CONTROL ROOM, GARDNER DENVER TRIPLEX PUMP [REPORTED AS OUT OF SERVICE]	
			OPERATOR CONTROL ROOM, GARDNER DENVER TRIPLEX POMP [REPORTED AS OUT OF SERVICE]	
ST210	2007	INTERNATIONA	4X4 S/A EXTENDED CAB SERVICE TRUCK, DT 466 ENGINE, 6-SPEED TRANSMISSION, FRONT TIRES	GP
		L	11R22.5-ALUM BUDS, REAR TIRES 11R22.5-ALUM BUDS, C/W ALI ARC GUARD, 16'X8' SERVICE BODY	
			W/ JABCO MODEL 1817 CRANE S/N 12679	
UT311	2007	DOEPKER	TRIDEM 50' SCISSORNECK LOW BOY TRAILER, 30,600-LB AXLES, TIRES 275/70R22.5, ALUM BUDS, AIR	GP
			RIDE SUSPENSION, (LIC. W531-13), C/W AIR CENTER LIVE ROLL, HYDRAULIC REAR LIVE ROLL,	
			FOLDING SIDE MARKERS	
CB907	2007	WESTERN STAR	TRI-DRIVE COMBO TRUCK, 32" SLEEPER, DETROIT SERIES 60 515-HP ENGINE, 18-SPEED, FRONT TIRES	CNRL
			445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM BUDS, 23000-LB X 3 AXLES,	HORIZON
			AIRLINER AIR RIDE SUSP, S/N 5KKPALCK57PX81057, C/W CARGO TANK MFG 2-COMPARTMENT	PROJECT
			TANK, FRONT 1,216-GALLON, REAR 2,978-GALLON, TC SPEC 407/412, S/N V22326C, HIBON VTB-820	
			BLOWER, HOTSY WB 9460 HEATER	
H728	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATONG FULLER 18-SPEED, FRONT	ED
			TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK, S/N	
			2724076, 10,822-LITRE	
SK010	2007	FREIGHTLINER	ISTEAM TRUCK	ED
H737	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETRO!T 60 SERIES 14L 515-HP ENGINE, EATON FULLER 18	ED
	2007		SPD, 1 09, 468-KILOMETERS INDICATED, 13,812-HOURS INDICATED, FRONT AXLE 445/65R22.5 ALUM	
			BUDS 20,000-LB, REAR AXLES 11R/24.5 ALUM BUDS 60,000-LB, C/W CUSTOM VAC UNIT, 10,822 L	
			CAPACITY S/N 27250706, HOTSY WATER HEATER, ROBUSCHI BLOWER, HYDRAULIC BOOM	
	1		CAPACIT SIN 27250700, HOTST WATER HEATER, ROBOSCHI BLOWER, HTDRAGGE BOOM	
H747	2007	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT DIESEL 60 SERIES 475-HP ENGINE, EATON FULLER 18-	ED
			SPEED, 20,400-LB FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,200-LB REAR AXLES,	
			REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC STEEL TANK S/N 27940906, 10,822 L,	
			TC407/412, ROBUSCHI BLOWER MODEL RB-DV-145 SP/UN, S/N 06-30185, WATERBLAST HEATER,	
			HYDRAULIC BOOM	
H750	2007	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES DIESEL ENGINE, EATON FULLER 18-SPEED,	CNRL
			FRONT AXLE 445/65R22.5 ALUM BUDS 20,000-LB, REAR AXLES 11R24.5 ALUM BUDS-20 61,200-LB,	
			AIR RIDE SUSPENSION, C/W CUSTOM VAC HYDRO-VAC UNIT, TC407/412, 10,822 L CAP, S/N	
			28951066, HIBON BLOWER, HOTSY WATER BLAST WB9460 HEATER, HYDRAULIC BOOM	
P271	2007	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, DIESEL ENGINE, AUTO, TIRES 265/75R16, C/W LEATHER, SUNROOF,	ED
	2007	O / L VII G LL I	BUCKETS W/ CONSOLE, WOOD BOX LINER, RUNNING BOARDS	
P277	2007	CHEVROLET	PICKUP TRUCK, 3/4-TON, CREW CAB, MANUAL TRANSMISSION, 255,436-KILOMETERS INDICATED,	ED
	2007	J. J. J. J. J. J. J. J. J. J. J. J. J. J	2,599 ENGINE HOURS, 4WD, SUNROOF, OIL/GAS TANK W/ PUMP, DURAMAX DIESEL ENGINE	
			2,255 Erroll Erroll of Trop South Corp (Stop Stop France)	
V558	2007	WESTERN STAR	TRI-DRIVE VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	ED
			445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W PROGRESS TANK, S/N V22066-C,	
			15,906-LITRE	
NP011	2007	WESTERN STAR	TRI-DRIVE NITROGEN PUMP UNIT, SLEEPER/CONTROL CAB COMBO, CAT ENGINE, EATON FULLER 18-	ED
			SPEED, FRONT TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, AIR RIDE SUSPENSION, C/W	
		,	WELLQUIP NITROGEN PUMP UNIT, TANK S/N 30160, 11,234 LITRE	
TR014	2007	WESTERN STAR	T/A TRACTOR W/ 46" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION,	ED
		1	FRONT TIRES 11R24.5-ALUM BUDS, 14,000-LB FRONT AXLE, REAR TIRES 11R24.5-ALUM BUDS,	
			46,000-LB REAR AXLES, NEWAY AIR RIDE SUSPENSION, S/N 5KHALAV17PY07157, C/W WET KIT,	
			SLIDING 5TH WHEEL	
H735	2007	WESTERN STAR	TRI-AXLE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 14L 515-HP ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR	
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, S/N 5KKPALCK87P81053, C/W CUSTOM VAC SERVICES	
			HYDRO-VAC UNIT S/N 27730706, HOTSY WATER BLAST, 10,822-LITRE CAPACITY, TC 407/412 TANK,	
		1	ROBUSHI BI OWER, HYDRALII IC BOOM	

H752

Q006

2007

2007

STEPHENS

ED

ZAMA CITY

	************		Asset List* GA GIODAI PARTIERS I	mananan
Equip. #	Yen)1.	Make	Description	Location
H 744	2007		TRI-AXLE HYDRO-VACUUM TRUCK, SLEEPER, DETROIT 60 SERIES ENGINE, EATON FULLER 18-SPEED	ED
			TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 61,200-LB REAR	
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G45174), C/W CUSTOM VAC	
			SERVICES UNIT S/N 27920906, ROBUSHI BLOWER TC 407/412 TANK, 10,822-LITRE CAPACITY, HOTSY	
			STEAMER & WATER BLAST SYSTEM, HYDRAULIC BOOM	
NP010	2007	WESTERN STAR	TRI-AXLE NITROGEN PUMPER TRUCK, SLEEPER/CONTROL CAB COMBO, CAT C15 ACERT 550-HP	ED
			ENGINE, 18-SPEED TRANSMISSION, 192,152-KILOMETERS INDICATED, 10,692-HOURS INDICATED,	
			FRONT TIRES 445/65R22.5-ALUMINUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-, CRYOGENIC	
		•	VESSEL ALTERNATIVES TANK S/N 30158 (2008) W/ 2,964-GALLON CAPACITY	
NTR005	2007	WESTERN STAR	TANDEM-AXLE HIGHWAY TRACTOR W/ SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER	ED
			TRANSMISSION, FRONT TIRES 425/65R22.5, AŁUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, AIR	
			LINER AIR RIDE SUSPENSION, (LIC G19616), C/W ALUM TOOL BOX, 298,851-KILOMETERS INDICATED,	
			9.174-HOURS INDICATED	
NTR007	2007	1	T/A HIGHWAY TRACTOR, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION,16,000-	ED
			LB FRONT AXLE, 385/65R22.5 FRONT TIRES, ALUM BUDS, 46,000-LB REAR AXLES, 11R24.5 REAR	
+ 6			TIRES, ALUM BUDS, SLEEPER, (LIC G19649), 232,131-KILOMETERS INDICATED, 7,073-HOURS	
		<u> </u>	INDICATED	
NTRO08	2007	WESTERN STAR	T/A HIGHWAY TRACTOR, CAT C15 475-HP, EATON FULLER 18-SPEED, 385/65R22.5 FRONT, 11R24.5	ED
	2007	MIROTER N. CTAR	REAR, ALUM BUDS, AIR RIDE	FB.
NTR010	2007	WESTERN STAR	TRUCK TRACTOR, CATERPILLAR C15 475-HP, EATON FULLER 18-SPEED TRANSMISSION, 24" SLEEPER	ED
CD040	2007	LAUCCTE DALCTAD	CAB, 388,338-K!LOMETERS INDICATED, 11,082-HOURS INDICATED TRI-DRIVE COMBO TRUCK W/ 26" SLEEPER, CAT C15 550-HP ENGINE, EATON FULLER 18-SPEED	ĠР
CB910	2007	WESTERN STAR	l ' ' ' ' ' ' ' ' ' ' ' ' ' ' ' ' ' ' '	GP
			TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	
			11R24.5-ALUM BUDS, 61,000-L8 REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W PROGRESS TANK	
			S/N V223660, TC407/412, 2 COMPARTMENT - 8,002-LITRE/11,185-LITRE CAPACITY, HIBON BLOWER, HOTSY WATER BLAST	
CB914	2007	WESTERN STAR	TRI-DRIVE COMBO VACUUM TRUCK W/36" SLEEPER, CAT C15 550-HP ENGINE, EATON FULLER 18-	GP
			SPEED TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5	
			ALUM BUDS, 69,000-LB AXLES, AIRLINER AIR RIDE SUSPENSION, C/W PROGRESS 2 COMPARTMENT	
			TANK S/N V224860, DOT407/412,	
			11.227-LITRE CAPACITY, HIBON VTR 820 BLOWER, WATER BLAST HEATER	
PS101	2007	FREIGHTLINER	PRESSURE TRUCK, S/A, 4X4, EXTENDED CAB, MERCEDES BENZ ENGINE, AUTOMATIC TRANSMISSION,	RL
		1	FRONT TIRES 385/65R22.5-ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 42X25.00-20, 21,000-LB	
		1	REAR AXLES, SPRING SUSPENSION, C/W CUSTOM VAC SERVICES STEEL TANK S/N 29830107, TC407, 3	
			- COMPARTMENT 10,329-LITRE CAPACITY, GENERAL IND. PUMP, TRIPLEX PUMP, 2" T&E FLUID	
044711	ļ		PUMP INO BATTERY)	
PT112	2007	INTERNATIONA	S/A PRESSURE TEST TRUCK W/ EXTENDED CAB, DT466 ENGINE, EATON FULLER 6-SPEED	RL
		L	TRANSMISSION, FRONT TIRES 11R22.5-ALUM BUDS, 12,000-LB FRONT AXLE, REAR TIRES 11R22.5-	
			ALUM BUDS, 21,000-LB REAR AXLE, (LIC G20941), C/W INTERCONTINENTAL 20' VAN BODY,	
			GARDNER DENVER TRIPLEX PUMP, 4 TANKS, OPERATOR CONTROL ROOM	
H748	2007	WESTERN STAR	TRI-DRIVE HYDRO VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 20,000-LB	ED
			FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 61,000-LB REAR AXLES, REAR TIRES 11R24.5,	
			ALUM BUDS, AIR RIDE SUSPENSION, (LIC G49531), C/W CUSTOM VAC S/N 28941006, TCRN	
			Z.05,437.05.04, TC 407/412, 10,822-LİTRE, 309,139-KILOMETERS INDICATED, 13,451-HOURS	
LITTO	2007	MICCIENN CTAR	INDICATED	70
H739	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, 20,000-LB	ZC
			FRONT AXLE, FRONT TIRES 445/65R22.5, ALUM BUDS, 62,000-LB REAR AXLES, REAR TIRES 11R24.5,	
			ALUM BUDS, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC HYDRO-VAC UNIT, TC407/412,	
			TCRN Z-05-437-05-04, 10,822 L, S/N 25380706, ROBUSCHI BLOWER, HOTSY, WATER BLAST,	

WESTERN STAR TRI-DRIVE HYDRO-VACUUM TRUCK, CAT ENGINE, EATON FULLER 18-SPEED, FRONT TIRES

HEATER. BOOM
QUAD WATER TRAILER, 34,000-LITRE CAPACITY, 2 COMPARTMENT, ALUMINUM

445/65R22.5, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, (LIC G-50654), C/W CUSTOM VAC UNIT, S/N 28981.106, TC407/412, 10,822 L, HIBON BLOWER, HOTSY WATER BLAST WB9460

HYDRAULIC BOOM



	Vicar	Make	Description	Lacotton
TRO12	2007		T/A TRACTOR W/ 39" SLEEPER, CAT C15 475HP ENGINE, EATON FULLER 18-SPEED, FRONT TIRES	MEDICINE
771022	2007	The contract of the contract o	11R24.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, AIRLINER AIR RIDE SUSPENSION, C/W WET	HAT
		1	KIT. MAGNUM ALUM BUMPER	111.11
H741	2007	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, FULLER 18-SPEED TRANS, CUSTOM VAC UNIT S/N 27280806,	SWIFT
	2007		11,915L CAP	CURRENT
H751	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, CAT ENGINE, 8-SPEED, FRONT TIRES 445/65R22:5, ALUM BUDS,	SWIFT
	2007		REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC TANK S/N 28961106, TC407/412, 10,822-LITRE	CURRENT
			CAPACITY	COMMENT
H753	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT	SWIFT
,,,,,,,	2507	1	TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, C/W CUSTOM VAC UNIT, S/N	CURRENT
		1	28971105, 10,822 L, 192,225-KILOMETERS INDICATED, 13,057-HOURS INDICATED	COMMENT
HU002	2007	EREIGHTLINER	FLUID HEATER UNIT W/ SLEEPER, TRI AXLE, CAT C15 ENGINE, EATON FULLER 18-SPEED	SWIFT
	2007		TRANSMISSION, FRONT TIRES 425/65R22.5-ALUM BUDS, 16.000-LB AXLE, REAR TIRES 11R24.5-ALUM	CURRENT
			BUDS, 52,000-LB AXLES, (LIC H-26759) C/W 12 MILLION BTU BURNER, 3 COMPARTMENT TANK 1,300-	
			LITRE, 4,900-LITRE, AND 3,800-LITRE CAPACITY, 141,898-KILOMETERS INDICATED, 1,448-HOURS	
			INDICATED	
TR013	2007	WESTERN STAR	T/A TRUCK TRACTOR W/ 48" SLEEPER, CAT C15 475-HP ENGINE, EATON FULLER 18-SPEED	SWIFT
111023	2007		TRANSMISSION, FRONT TIRES 11R24.5-ALUM BUDS, 14,000-LB FRONT AXLE, REAR TIRES 11R24.5-	CURRENT
			ALUM BUDS, 46,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W PTO DRIVE, 3" FLUID	COMMENT
			PUMP, 357,747-KILOMETERS INDICATED, 17,087-HOURS INDICATED	
CT2062	2007	KENWORTH	TANDEM-AXLE STEER / TRI-AXLE COILED TUBING UNIT, CAT C15 ENGINE, EATON FULLER 18-SPEED	ED
012002	2007	N.C.IVVOIVIII	TRANSMISSION, 425/65R22.5 FRONT TIRES, ALUM BUDS, 11R24.5 REAR TIRES, ALUM BUDS, NEWAY	20
			AIR RIDE SUSPENSION, C/W C-TECH COILED TUBING UNIT, HYDRA RIG RT25 INJECTOR, 60"	
			GOOSENECK, SPICER 792, ALLIANCE ALUM CONTROL CABIN W/HYDRAULIC LIFT, FASSI 7-TON PICKER	
			· · · · · · · · · · · · · · · · · · ·	
CB913	2007	WESTERN STAR	E150AC24_SIDE_OUTRIGGERS_S/N 5159 COMBINATION UNIT, TRI-DRIVE HYDRO-VACUUM TRUCK, W/ WASH SYSTEM, DETROIT SERIES 60	MEDICINE
C D313	2007	l:	515-HP ENGINE, EATON FULLER 18-SPEED TRANSMISSION, AIR RIDE SUSPENSION, SLEEPER, W// 10-	HAT
			CU. METER DEBRIS, AND 4-CU. METER WATER SECTIONS	HAI
H749	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT DIESEL ENGINE, EATON FULLER 18-SPEED, FRONT	MEDIC:NE
11743	2007		TIRES 445/65R22.5, ALUM BUDS, REAR TIRES 11R24.5, ALUM BUDS, AIR RIDE SUSPENSION, C/W	HAT
		i	CUSTOM VAC UNIT, TC407/412, 10,822 L, ROBUSHI BLOWER, HOTSY, WATER BLAST WB9460	пат
		İ		
H743	2007	WESTERN STAR	HEATER, HYDRAULIC BOOM, S/N 28931006 TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT SERIES 60 515-HP ENGINE, EATON FULLER 18-SPEED	SC
117-13	2007	WESTERNSIAN	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB AXLE, REAR TIRES 11R24.5-ALUM	
			BUDS, 60,000-LB AXLES, AIR RIDE SUSPENSION, 36" SLEEPER, C/W CUSTOM VAC SERVICES, 10,822-	
			BODS, BUJUUU-LB AALES, AIR RIDE SUSPENSION, SO SLEEPER, C/W CUSTON VAC SERVICES, 10,822-	
H725	2007	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK, DDC 60 515-HP ENGINE, EATON FULLER 18-SPEED	SC
1:723	2007	1	TRANSMISSION, FRONT TIRES 445/65R22.5-ALUM BUDS, 20,000-LB FRONT AXLE, REAR TIRES	30
		1	11R24.5-ALUM BUDS, 60,000-LB REAR AXLES, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
		1	SERVICES STEEL TANK S/N 25360606, TC407/412, 2 COMPARTMENT - 10.822-LITRE CAPACITY.	
HU004	2007	GMC	ROBUSCHUBLOWER, WATERBLAST HEATER HEATER TRUCK, 4X4, S/A, 6.6 DURAMAX DIESEL ENGINE, 6-SPEED ALLISON TRANSMISSION, FRONT	GP
110004	2007	1	TIRES 245/70R19.5, 8,000-LB AXLE, REAR TIRES 245/70R19.5, 13,000-LB AXLE, (LIC H-26761) C/W 3	0,
			TANKS, 568-LITRE CAPACITY , 568-LITRE CAPACITY AND 1,200-LITRE CAPACITY, 3,500,000-BTU	
UT107	2007		BURNER TRIDEM SCISSORNECK TRAILER, TIRES 275/70R22.5-ALUM BUDS, AIR RIDE SUSPENSION, C/W 10'X35'	GP
01107	2007	1		Gr
H745	2007	TRAILER	WORKING DECK, 40-TON, CENTER LIVE ROLL, FLIPOVER REAR LIVE ROLL, BEAVERTAILS TRI-DRIVE HYDRO-VACUUM TRUCK, DETROIT 60 SERIES 515-HP ENGINE, EATON FULLER 18-SPEED	
Л/45	2007	1		
		1	TRANSMISSION, 20,000-LB FRONT AXLE, 445/65R22.5 FRONT TIRES, ALUM BUDS, 69,000-LB REAR	
			AXLES, 11R24.5 REAR TIRES, ALUM BUDS, AIRLINER AIR RIDE SUSPENSION, C/W CUSTOM VAC	
		1	SERVICES VAC UNIT S/N 28921006, 10,822-LITRE CAPACITY, TC407/412 TANK, ROBUSCHI RB DV	
			145/SP UN BLOWER S/N 06-30190, HOTSY WB9460 WATER BLAST HEATER S/N 60430806, BOOM	
0070	2007	CHEADOLE	DICKLID TOLICK CODIN CAD AVA DIDAMAN DICCI ENCINE DUMP TANK	
P278	2007	CHEVROLET	PICKUP TRUCK, CREW CAB, 4X4, DURAMAX DIESEL ENGINE, PUMP TANK	
ED04:	2000	NOSTU	TOINTENA CHID DUIMIN TRAILED TIDEC MADOME CYTEL DUIDE COMMON HID TARD CEMICS COMMON	
ED014	2008	NORTH	TRIDEM END DUMP TRAILER, TIRES 11R24.5, STEEL BUDS, C/W ROLL UP TARP, SEALED END GATE,	ED
		COUNTRY	[30' L	

			Description	Location
	1			
Q009	2008	STEPHENS	QUAD WATER TRAILER, 34,000-LITRE CAPACITY, 2 COMPARTMENT, ALUMINUM	SWIFT CURRENT
VTR002	2009	WESTERN STAR	COMPLETE SEMI VAC UNIT, TANDEM AXLE TRACTOR, SLEEPER CAB, FULLER 18-SPEED	GP
		1	TRANSMISSION, 14,230-LB FRONTS, 46,000-LB REARS, CUSTOM VAC TRI-AXLE VAC TRAILER, S/N	
			36290807, 24.991-LITRE CAPACITY	
HO014	2009	WESTERN STAR	TRI-DRIVE HOT OILER TRUCK, SLEEPER, ALLISON AUTO TRANS, 385/70R22.5 FRONT, ALUM BUDS,	GP
			11R24.5 REAR, ALUM BUDS, C/W 2008 WELLQUIP HOT OILER UNIT, 2008 FABMASTER 5 MMBTU OIL	
			BURNER S/N 709, 2008 HAMMS TC406-MS-03-EL STEEL TANK 3-COMPARTMENT (28/24/8 BBL) S/N	
			46ANM063M81444012, 2006 SPM MODEL TWS600\$ TRIPLEX PUMP, S/N F1006-3885, OPERATORS	
			CONTROL PANEL HOSE REEL	
SU014	2009	WESTERN STAR	SOUR SEAL TRUCK - TC407 SEALED UNIT, W/ SCRUBBER - TANK S/N 4899-09-09	ZC
SVT002	2009	N/A	COMPLETE SEMI-VAC UNIT	SWIFT
301002	2003	IN/A	CONTINUE TE SEIVIT-VAC ONTI	CURRENT
P309	2011	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	GP
P316	2011	DODGE	PICKUP TRUCK	EÐ .
			NA THE RESIDENCE OF THE PROPERTY OF THE PROPER	
P285	2011	DODGE	PICKUP TRUCK, CREW CAB, HEMI 5.7L HEMI ENGINE, 4WD	ED
5200	2011	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	ED
P308	2011	FORD	PICKOP TROCK, CREW CAB, 49VD, TORBO DIESEL ENGINE	ED
P312	2011	FORD	P!CKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	RL
1,312	2011	1000	There is the stay that a stay the stay that a stay the stay that a stay the stay that a st	
P286	2011	DODGE	PICKUP TRUCK, CREW CAB, LONG BED, 5.7L HEMI ENGINE, 4WD, OIL/FUEL TANK W/ PUMP, 144,425-	SWIFT
		1	KILOMETERS INDICATED	CURRENT
P314	2011	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	ZAMA CITY
P313	2011	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	
P288	2012	DODGE	PICKUP TRUCK, CREW CAB, SHORT BED, 5.7L HEM! ENGINE, 4WD	ĘD
. 2.00	1		,	
P311	2012	FORD	PICKUP TRUCK, SUPER DUTY, 4-DOOR, MEGA CAB, LONG BED, C/W FUEL TANK / TOOL BOX, W/ 12-	ED
			VOLT PUMP	
P287	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	GP
	2012	50000	DIGGUE TO LOW COLLUS ON CUMA MAN CONFEST, SHOWE AND CALLY ON STEED INDICATED	C.D.
P293	2012	DODGE	PICKUP TRUCK, CREW CAB, CUMMINS DIESEL ENGINE, 4WD, 641-KILOMETERS INDICATED	. GP
P296	2012	DODGE	PICKUP TRUCK, CREW CAB, CUMMINS DIESEL ENGINE, 4WD	GP
1230	2012	00000	THOUSE THOUSE AND ADMINISTRATE PROPERTY.	Ξ,
P315	2012	FORD	PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE	ZC
P302	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	CNRL
				HORIZON
B004	2042	DODOF	DICKUD YOUR CORP. OAR AND	PROJECT
P304	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	ED
P295	2012	DODGE	PICKUP TRUCK, CREW CAB, CUMMINS DIESEL ENGINE, 4WD, FUEL/GAS TANK W/ PUMP, 931-	MEDICINE
. 233			KILOMETERS INDICATED	HAT
NP013	2012	N/A	NITROGEN PUMPER [ASSUMED TRUCK MOUNTED]	ED
₽289	2012	DODGE	PICKUP TRUCK, CREW CAB, HEMI 5.7L HEMI ENGINE, 4WD, 2,508-KILOMETERS INDICATED	ED
	ļ			
P292	2012	DODGE	PICKUP TRUCK, CREW CAB, HEMI 5.7L HEMI ENGINE, 4WD, 2,159-KILOMETERS INDICATED	ED
P298	2012	DODGE	PICKUP TRUCK, CREW CAB, 4WD	ED
F430	2012	DODGE		
	1			

P324

P327 H759

H761

2014

2014

2014

2014

DODGE

DODGE

PICKUP TRUCK, CREW CAB, 4WD

PICKUP TRUCK, CREW CAB, 4WD

WESTERN STAR HYDRO-VACUUM TRUCK, TRI-DRIVE, DETAILS NOT PROVIDED

WESTERN STAR HYDRO-VACUUM TRUCK, TRI-DRIVE, DETAILS NOT PROVIDED

ED

ED

SWIFT CURRENT

SWIFT CURRENT

Equip. II Year Make Location P299 2012 DODGE PICKUP TRUCK, CREW CAB, 4WD P300 2012 DODGE PICKUP TRUCK, CREW CAB, 4WD CNRL P301 2012 DODGE PICKUP TRUCK, CREW CAB, 4WD ED P303 2012 DODGE PICKUP TRUCK, CREW CAB, 4WD CNRL 2012 FORD PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE FD P306 PICKUP TRUCK, CREW CAB, 4WD, TURBO DIESEL ENGINE P307 2012 FORD ED FORD PICKUP TRUCK, SUPER DUTY, 4-DOOR, MEGA CAB, LONG BED, C/W FUEL TANK / TOOL BOX, W/ 12-P310 2012 ΕD VOLT PUMP MICRON NTT001 2012 NITROGEN TUBE TRAILER FD P290 2012 DODGE PICKUP TRUCK, CREW CAB, SHORT BED, 5.7L HEMI ENGINE, 135,546-KILOMETERS INDICATED **SWIFT** CURRENT P305 2012 DODGE PICKUP TRUCK, SHORT BED, CREW CAB, 4WD, 114,507- KILOMETERS INDICATED **SWIFT** CURRENT KENWORTH TRUCK TRACTOR, TANDEM AXLE TR016 2012 SWIFT CURRENT 2012 DODGE PICKUP TRUCK, LONG BED EXTENSION, CREW CAB, 4WD P297 ΕD OVERLAND FRAC HEATER UNIT, 35-MM GP HU007 2012 P291 2012 DODGE PICKUP TRUCK, CREW CAB, 4WD P294 2012 DODGE PICKUP TRUCK, CREW CAB, CUMMINS DIESEL ENGINE, 4WD, 532-KILOMETERS INDICATED NTR011 2013 WESTERN STAR TRUCK TRACTOR, TRI-AXLE GP H758 2013 WESTERN STAR HYDRO-VAC, TRI-AXLE, W/ TORNADO ΕD 2013 INTERNATIONA HYDRO-VACUUM TRUCK, TRI-AXLE, W/ OPERATORS COMPARTMENT, EATON FULLER 18-SPEED SC H757 TRANSMISSION, FRONT TIRES 445/65R22.5, REAR 11R24.5, INTERNATIONAL ENGINE, HOTSY STEAMER, W/ WATER BLAST SYSTEM, HYDRAULIC BOOM 2013 INTERNATIONA HYDRO-VACUUM TRUCK, TRI-AXLE, EATON FULLER 18-SPEED TRANSMISSION, HOTSY STEAMER, W/ SWIFT H756 WATER BLAST SYSYTEM, HYDRAULIC BOOM, 65,695-KILOMETERS INDICATED, 3,288-HOURS CURRENT IND!CATED 2013 N2 PUMPER HT RECOVERY NP013 N/A FD H760 2014 WESTERN STAR HYDRO-VACUUM TRUCK, TRI-DRIVE, DETAILS NOT PROVIDED CNRL NP015 2014 N/A NITROGEN PUMPER TRAILER ED P326 2014 DODGE PICKUP TRUCK, CREW CAB, 4WD GP

Equip. #			Description	Location
P323	2014	DODGE	PICKUP TRUCK, CREW CAB, 4WD	GP
P325	2014	DODGE	PICKUP TRUCK, CREW CAB, 4WD	MH
PTI109	2014	KENWORTH	PRESSURE TRUCK, 600-HP, 6X4, WITH TRIPLEX PUMP, VESSEL, OPERATORS CAB [NEW IRON ADDED	GP
	ļ		IN 2015]	
H762	2014	WESTERN STAR	TRI-DRIVE HYDRO-VACUUM TRUCK	
H763	2014	WESTERN STAR	HYDRO-VACUUM TRUCK, TRI-DRIVE, DETAILS NOT PROVIDED	
RN089		· N/A	TRAILER, CREW SHACK	ED
RN093		N/A	TRAILER, CREW SHACK	ED
RN097		N/A	TRAILER, CREW SHACK	ED
1111057		17/5	TRAILIN GREW STACK	LD
RN100		N/A	TRAILER, CREW SHACK	ED
RN108		N/A	TRAILER, CREW SHACK	EĐ
7117200			The matery of the Total Control of the Control of t	CD
RN110		N/A	TRAILER, CREW SHACK	ED
RN112		N/A	TRAILER, CREW SHACK	ED
UT359		TRAILMOBILE	DRY VAN TRAILER	GP
		,	- 11 11 11 11 11 11 11 11 11 11 11 11 11	
PSF108		N/A	SKID UNIT, PRESSURE TEST UNIT, W/ PUMP AND STORAGE SYSTEMS	ED
MV046		N/A	TRIDEM WELLSITE TRAILER (LIVING QUARTERS W/ 2-BEDROOMS)	ED
RN081		N/A	TRAILER, CREW SHACK	ED
RN082		N/A	TRAILER, CREW SHACK	ED
XW073		MACK	TRUCK TRACTOR, TANDEM AXLE [INSUFFICIENT INFORMATION]	ED
UT360		MOND	STORAGE TRAILER, 53'	ED
UT361			STORAGE TRAILER, 53'	ED
CT2071-2			SPOOLER, W/ GEAR DRIVE	ED
CSC003		N/A	TUBING SPOOLER	ED
UT358		FRUEHAUF	DRY VAN TRAILER	ED
01336		TROCHAGE	PAT YOU THAILER	
RN035		N/A	PROPANE TANKS, ESTIMATED 50-GALLON	ED
RN051		N/A	TRAILER, CREW SHACK	ED
UT009		N/A	UTILITY TRAILER, FLATDECK, ESTIMATED 20'	ED
UTC318		N/A	TRAILER, FLATDECK, 18'	
UTH050		N/A	TRAILER, FLATDECK	
MV002		MOUNTAIN	TRI-AXLE CREW SHACK TRAILER, PLATE #S377-88	
		VIEW		<u></u>
UŢ108		N/A	BOOSTER FOR UT107	
		I		

Additional Assets

V065	N/A	N/A	VAC TRUCK	
V068	N/A	N/A	VAC TRUCK	
V063	N/A	N/A	VAC TRUCK	



Equip. H	Year	Make	Description	Location
SSK004	N/A	N/A	SKID STEER	
S005	N/A	N/A	STEAM TRUCK	
Q009	N/A	N/A	CUSTOM QUAS VAC	
CC2071	N/A	- N/A	CATCH TANK	
W077	N/A	N/A	WATER TRUCK	
TRP802	N/A	N/A	PICKER TRUCK	

SCHEDULE B

Process Letter

Deloitte

Deloitte Restructuring Inc. 700, 850 – 2 Street SW Calgary AB T2P 0R8 Canada

Tel: 403-503-1458 Fax: 403-718-3681 www.deloitte.ca

June 29, 2015

Attention: Potential Bidders for the Assets of Big Eagle

Dear Sirs:

Subject: Big Eagle Hydro-Vac Inc., Big Eagle Limited Partnership, Big Eagle Hydro-Vac (North)
Inc. and Big Eagle (North) Limited Partnership (the "Companies") – In Receivership

Deloitte Restructuring Inc. was appointed as receiver-manager (the "Receiver") of the property, assets and undertaking of the Companies pursuant to the terms of a June 8, 2015 Order of the Court of Queen's Bench of Alberta (the "Court").

Unless otherwise defined, all capitalized terms herein have the meanings ascribed to them in the AASP (as defined below).

Pursuant to an Order of the Court dated June 29, 2015 (the "Approval Order"), the Court authorized the Receiver to enter into a stalking horse Auction Services Agreement (the "ASA") with Great American Global Partners, LLC ("GA"). The Approval Order also authorized the Receiver to solicit competing bids to the ASA pursuant to the terms and conditions of Auction and Asset Purchase Solicitation Procedures (the "AASP").

Pursuant to section 2.3 of the AASP, the Receiver is hereby inviting you to make an Auction Bid or Asset Purchase Bid (or both) in relation to the Assets of the Companies pursuant to the AASP.

The ASA and the AASP can be found on the Receiver's website at http://www.insolvencies.deloitte.ca/en-ca/Pages/Big-Eagle-Group.aspx.

Each interested party is responsible for ensuring its own compliance in form and substance with the AASP. We strongly encourage you to review the ASA and the AASP in detail.

Pursuant to the AASP, a Participating Bidder must deliver a Qualified Auction Bid or Qualfied Asset Purchase Bid (or both) to the Receiver by no later than 5:00 p.m. Calgary time on July 15, 2015 (the "Bid Deadline"). In the event that the Receiver receives at least one Qualified Bid by the Bid Deadline, the Receiver shall invite GA and the Qualified Bidder(s) to the Auction prescribed by the AASP, which shall occur on July 17, 2015 at the offices of the Receiver. The Auction shall be conducted pursuant to Auction procedures developed and adopted by the Receiver.

The Successful Bid and the Backup Bid (if applicable) shall be approved by the Court at the Approval Hearing.

Potential Bidders for the Assets of Big Eagle
June 29, 2015
Page 2

If you have any questions or require further information, please contact the undersigned.

Yours truly,

DELOITTE RESTRUCTURING INC.

In its capacity as Receiver of Big Eagle Hydro-Vac Inc., Big Eagle Limited Partnership, Big Eagle Hydro-Vac (North) Inc. and Big Eagle (North) Limited Partnership and not in its personal capacity.

Jeff Keeble, CA•CIRP, CBV Senior Vice-President

Enclosure